Detecting and Analyzing Genomic Structural Variation using Distributed Computing

Christopher Whelan A.B. Computer Science, Harvard University, 1997

Presented to the Center for Spoken Language Understanding within the Oregon Health & Science University School of Medicine in partial fulfillment of the requirements for the degree Doctor of Philosophy in Computer Science & Engineering

February 2014

Center for Spoken Language Understanding School of Medicine Oregon Health & Science University

CERTIFICATE OF APPROVAL

This is to certify that the Ph.D. dissertation of Christopher Whelan has been approved.

> Dr. Kemal Sönmez, Thesis Advisor Associate Professor

> > Dr. Izhak Shafran Associate Professor

Dr. Brian Roark Research Scientist, Google, Inc.

Dr. Lucia Carbone Assistant Professor, Dept. of Behavioral Neuroscience

> Dr. Steven Bedrick Assistant Professor

Acknowledgments

This dissertation would not be possible without my co-advisors, Kemal Sönmez and Lucia Carbone. Kemal has been an excellent mentor and a great friend. He helped me to develop many ideas that went into this dissertation. Of all of his valuable contributions, I am most thankful for his ability to point me in new directions and re-energize me when I was discouraged or had painted myself into a corner.

Lucia has also been a terrific role model and an inspiring scientific leader. The day I was given the chance to collaborate with her and her lab was my luckiest moment at OHSU. Lucia has taught me most of the biology I know, given me the opportunity to work on tremendously exciting projects, and gone out of her way many times to introduce me to members of the genomics community; in particular, I owe a great debt of gratitude to her for bringing me annually to the Biology of Genomes meeting at Cold Spring Harbor. Computational work cannot have real value without actual biological data and real scientific questions to work on and learn from, both of which Lucia provided. Of course, Lucia also introduced me to the problem of structural variation detection; this dissertation could not exist without her. I would also like to thank the other members of the Carbone Lab, who have been a pleasure to work with and learn from: Larry Wilhelm, Josh Meyer, Nathan Lazar, Liz Terhune, and Kim Nevonen.

I would also like to thank Brian Roark and Zak Shafran for all of the great help they've given me. Brian is a fabulous teacher and always provided excellent feedback and made himself available for questions or just to talk. Cloudbreak started as a project in a course taught by Zak and Richard Sproat on problem solving with large clusters, which provided me with the core idea that eventually turned into this dissertation. Zak has done a great deal of other things for me as well: providing me with ideas to use in my research; inviting me to explore new topics in machine learning with him and his group; and last but not least acquiring and helping to configure and administer the Hadoop compute clusters at CSLU, without which this work would not be possible. On that last note, I would also like to thank cluster administrators who have put up with my sometimes taxing usage of our compute resources, including Rob Stites and Jason Brooks.

I would also like to acknowledge members of the broader genomics community who have provided feedback and encouragement for my projects, including Bob Handsaker, Steve McCarroll, and Ben Raphael.

Finally, my greatest thanks go to my wife Sarah and my son Colin, who displayed tremendous patience in putting up with their husband and father's graduate-school lifestyle. Sarah's support, encouragement, and love have been an integral component of this work, without which it could never have happened.

Contents

| A | Acknowledgments | | | |
|----------|-----------------------|---------|--|----|
| 1 | Intr | oducti | on | 1 |
| 2 | Bio | logical | Background | 6 |
| | 2.1 | Struct | ural Variations | 6 |
| | | 2.1.1 | SV Effects on Phenotype and Disease | 7 |
| | | 2.1.2 | Mechanisms and Signatures of SV Formation | 8 |
| | 2.2 | High-7 | Chroughput Short-Read Sequencing 1 | 10 |
| | | 2.2.1 | Sequencing Analysis Pipelines | 12 |
| | | 2.2.2 | Big Data from Sequencing | 16 |
| 3 | Alg | orithm | s for Structural Variation Detection | 8 |
| | 3.1 | The Fe | our Signals of SVs in Sequencing Data | 18 |
| | 3.2 | Read I | Pair Approaches | 19 |
| | | 3.2.1 | Ambiguously Mapped Read Pairs | 21 |
| | | 3.2.2 | Concordant Read Pairs | 22 |
| | 3.3 | Read I | Depth Approaches | 24 |
| | 3.4 | Split F | Read Approaches | 24 |
| | 3.5 | Assem | bly-Based Approaches | 25 |
| | 3.6 | Hybric | d Approaches | 26 |
| | | 3.6.1 | Support from Secondary Signals | 27 |
| | | 3.6.2 | Pipelines | 27 |
| | | 3.6.3 | Integrative Models | 28 |
| | 3.7 | An Ex | ample of an SV Detection Pipeline for a Cancer Dataset 2 | 28 |
| 4 | A F | ramew | ork for SV Detection in MapReduce | 32 |
| | 4.1 | MapR | educe and Hadoop | 32 |
| | 4.2 | Uses o | f Hadoop and MapReduce in Sequencing Tasks | 35 |
| | | 4.2.1 | DNA Resequencing | 36 |
| | | 4.2.2 | RNA-seq 3 | 38 |

| | | $4.2.3 de \text{ novo Assembly} \dots \dots \dots \dots \dots \dots \dots \dots \dots $ |
|----------|---|---|
| | | 4.2.4 Frameworks and Toolkits |
| | | 4.2.5 Other uses of Hadoop and MapReduce |
| | 4.3 | MapReduce Constraints on SV Algorithms |
| | 4.4 | A General MapReduce SV Detection Algorithm |
| | 4.5 | Discussion |
| 5 | Clo | udbreak |
| | 5.1 | Variant types detected |
| | 5.2 | Framework infrastructure |
| | 5.3 | Implementation of a MapReduce SV Algorithm |
| | 5.4 | Filtering Incorrect and Ambiguous Mappings |
| | 5.5 | Genotyping $\ldots \ldots 54$ |
| | 5.6 | Running in the Cloud |
| | | 5.6.1 Cloud-Enabled Genomics Tools |
| | | 5.6.2 Enabling Cloud Computing with Whirr |
| | 5.7 | Discussion |
| | | |
| 6 | Eva | luating Cloudbreak |
| 6 | Eva 6.1 | Iuating Cloudbreak 59 Evaluation Methods 59 |
| 6 | Eva 6.1 | Iuating Cloudbreak 59 Evaluation Methods 59 6.1.1 Choice of SV Detection Tools to Compare To 59 |
| 6 | Eva 6.1 | Iuating Cloudbreak 59 Evaluation Methods 59 6.1.1 Choice of SV Detection Tools to Compare To 59 6.1.2 Simulated and Biological Data Sets 60 |
| 6 | Eva 6.1 | Iuating Cloudbreak 59 Evaluation Methods 59 6.1.1 Choice of SV Detection Tools to Compare To 59 6.1.2 Simulated and Biological Data Sets 60 6.1.3 Parameters Used for Alignment and SV Detection 61 |
| 6 | Eva 6.1 | Iuating Cloudbreak 59 Evaluation Methods 59 6.1.1 Choice of SV Detection Tools to Compare To 59 6.1.2 Simulated and Biological Data Sets 60 6.1.3 Parameters Used for Alignment and SV Detection 61 6.1.4 SV Prediction Evaluation 61 |
| 6 | Eva 6.1 6.2 | Iuating Cloudbreak 59 Evaluation Methods 59 6.1.1 Choice of SV Detection Tools to Compare To 59 6.1.2 Simulated and Biological Data Sets 60 6.1.3 Parameters Used for Alignment and SV Detection 61 6.1.4 SV Prediction Evaluation 62 Results on Simulated Data 63 |
| 6 | Eva 6.1 6.2 | Iuating Cloudbreak 59 Evaluation Methods 59 6.1.1 Choice of SV Detection Tools to Compare To 59 6.1.2 Simulated and Biological Data Sets 60 6.1.3 Parameters Used for Alignment and SV Detection 61 6.1.4 SV Prediction Evaluation 62 Results on Simulated Data 63 6.2.1 Accuracy and Runtime 63 |
| 6 | Eva 6.1 6.2 | Iuating Cloudbreak 59 Evaluation Methods 59 6.1.1 Choice of SV Detection Tools to Compare To 59 6.1.2 Simulated and Biological Data Sets 60 6.1.3 Parameters Used for Alignment and SV Detection 61 6.1.4 SV Prediction Evaluation 62 6.2.1 Accuracy and Runtime 63 6.2.2 Choice of Window Size 64 |
| 6 | Eva 6.1 6.2 6.3 | Iuating Cloudbreak 56 Evaluation Methods 59 6.1.1 Choice of SV Detection Tools to Compare To 59 6.1.2 Simulated and Biological Data Sets 60 6.1.3 Parameters Used for Alignment and SV Detection 61 6.1.4 SV Prediction Evaluation 62 Results on Simulated Data 63 6.2.1 Accuracy and Runtime 63 6.2.2 Choice of Window Size 64 Results on Biological Data 64 6.2.2 Choice of Window Size 64 6.2.3 Results on Biological Data 64 |
| 6 | Eva6.16.26.3 | Iuating Cloudbreak59Evaluation Methods596.1.1 Choice of SV Detection Tools to Compare To596.1.2 Simulated and Biological Data Sets606.1.3 Parameters Used for Alignment and SV Detection616.1.4 SV Prediction Evaluation62Results on Simulated Data636.2.1 Accuracy and Runtime636.2.2 Choice of Window Size646.3.1 Accuracy and Runtime686.3.1 Accuracy and Runtime68 |
| 6 | Eva 6.1 6.2 6.3 | Iuating Cloudbreak59Evaluation Methods596.1.1 Choice of SV Detection Tools to Compare To596.1.2 Simulated and Biological Data Sets606.1.3 Parameters Used for Alignment and SV Detection616.1.4 SV Prediction Evaluation62Results on Simulated Data636.2.1 Accuracy and Runtime636.2.2 Choice of Window Size646.3.1 Accuracy and Runtime686.3.2 Breakpoint Resolution71 |
| 6 | Eva 6.1 6.2 6.3 6.4 | Iuating Cloudbreak56Evaluation Methods566.1.1 Choice of SV Detection Tools to Compare To566.1.2 Simulated and Biological Data Sets606.1.3 Parameters Used for Alignment and SV Detection616.1.4 SV Prediction Evaluation62Results on Simulated Data636.2.1 Accuracy and Runtime636.2.2 Choice of Window Size646.3.1 Accuracy and Runtime686.3.2 Breakpoint Resolution73Results on a Low-Coverage Cancer Data Set73 |
| 6 | Eva 6.1 6.2 6.3 6.4 6.5 | Iuating Cloudbreak56Evaluation Methods596.1.1 Choice of SV Detection Tools to Compare To596.1.2 Simulated and Biological Data Sets606.1.3 Parameters Used for Alignment and SV Detection616.1.4 SV Prediction Evaluation62Results on Simulated Data636.2.1 Accuracy and Runtime636.2.2 Choice of Window Size646.3.1 Accuracy and Runtime686.3.2 Breakpoint Resolution71Results on a Low-Coverage Cancer Data Set74 |
| 6 | Eva 6.1 6.2 6.3 6.4 6.5 6.6 | luating Cloudbreak 56 Evaluation Methods 59 6.1.1 Choice of SV Detection Tools to Compare To 59 6.1.2 Simulated and Biological Data Sets 60 6.1.3 Parameters Used for Alignment and SV Detection 61 6.1.4 SV Prediction Evaluation 62 Results on Simulated Data 63 6.2.1 Accuracy and Runtime 63 6.3.2 Choice of Window Size 64 6.3.1 Accuracy and Runtime 64 6.3.2 Breakpoint Resolution 71 Results on a Low-Coverage Cancer Data Set 73 Genotyping Variants 74 Notes on Evaluating Runtime 74 |
| 6 | Eva 6.1 6.2 6.3 6.4 6.5 6.6 6.7 | luating Cloudbreak 56 Evaluation Methods 59 6.1.1 Choice of SV Detection Tools to Compare To 59 6.1.2 Simulated and Biological Data Sets 60 6.1.3 Parameters Used for Alignment and SV Detection 61 6.1.4 SV Prediction Evaluation 62 Results on Simulated Data 63 6.2.1 Accuracy and Runtime 63 6.2.2 Choice of Window Size 64 6.3.1 Accuracy and Runtime 68 6.3.2 Breakpoint Resolution 71 Results on a Low-Coverage Cancer Data Set 73 Genotyping Variants 74 Notes on Evaluating Runtime 74 Choice of Aligner and Use of Multiple Mappings 74 |
| 6 | Eva 6.1 6.2 6.3 6.4 6.5 6.6 6.7 6.8 | luating Cloudbreak 56 Evaluation Methods 59 6.1.1 Choice of SV Detection Tools to Compare To 59 6.1.2 Simulated and Biological Data Sets 60 6.1.3 Parameters Used for Alignment and SV Detection 61 6.1.4 SV Prediction Evaluation 62 Results on Simulated Data 63 6.2.1 Accuracy and Runtime 63 6.3.2 Choice of Window Size 64 6.3.1 Accuracy and Runtime 65 6.3.2 Breakpoint Resolution 73 Results on a Low-Coverage Cancer Data Set 74 Notes on Evaluating Runtime 74 Choice of Aligner and Use of Multiple Mappings 74 Discussion 74 |

| 7 | \mathbf{Ext} | ending Local Feature Based Models of SV Detection in a Discrimi- |
|----|-----------------|---|
| | \mathbf{nati} | ive Machine Learning Framework |
| | 7.1 | Related Work |
| | 7.2 | SV Detection as a Sequence Labeling Problem |
| | 7.3 | Graphical Models for Sequence Labeling |
| | 7.4 | Integrating Features with Conditional Random Fields |
| | 7.5 | Features for SV Detection |
| | | 7.5.1 Read Pair Features |
| | | 7.5.2 Split-read features |
| | | 7.5.3 Read depth features |
| | | 7.5.4 Genome annotations |
| | | 7.5.5 Binarization of real-valued features |
| | | 7.5.6 A feature example |
| | | 7.5.7 Interaction and neighbor features |
| | 7.6 | Training the CRF |
| | 7.7 | Improving Cloudbreak Calls with CRF Predictions |
| | 7.8 | Results |
| | 7.9 | Features Selected by the CRF |
| | 7.10 | Discussion |
| 8 | Ana | alysis of Evolutionary Breakpoint Features |
| | 8.1 | Background |
| | 8.2 | Evolutionary Breakpoints in the Gibbon Genome Identified by BACs 104 |
| | | 8.2.1 Methods |
| | | 8.2.2 Results |
| | 8.3 | Analysis of Breakpoints from the Gibbon Genome Reference Sequence 106 |
| | | 8.3.1 Overlap with Genomic Features: Repeats and Genes |
| | | 8.3.2 Overlap with CTCF Binding Sites |
| | 8.4 | Discussion |
| 9 | Fut | ure Work |
| | 9.1 | Cloudbreak |
| | 9.2 | SV Detection with Discriminative Machine Learning |
| | 9.3 | Gibbon Genome Breakpoint Analysis |
| Bi | bliog | graphy $\ldots \ldots \ldots$ |

List of Tables

| 3.1 | A summary of published SV detection algorithms that combine more than one sequencing signal. | 27 |
|-----|---|-----|
| 6.1 | Deletions and insertions in the simulated data detected at maximum sensi- | |
| | tivity | 67 |
| 6.2 | Deletions and insertions in the simulated data detected at a low false dis- | |
| | covery rate | 67 |
| 6.3 | Cloudbreak accuracy with varying window sizes. | 69 |
| 6.4 | Precision, recall, and size of variants found in the NA18507 data set | 72 |
| 6.5 | Confusion matrices for the predicted genotype of deletions | 74 |
| 6.6 | Detailed runtimes for both data sets | 76 |
| 6.7 | Detected deletions and insertions on the simulated and NA18507 data sets $% \mathcal{A}$ | |
| | that overlap with repetitive elements | 78 |
| 7.1 | Feature definition for the CRF training and test data. | 90 |
| 7.2 | Most important features learned by the CRF model for deletion and inser- | |
| | tion breakpoints. | 98 |
| 8.1 | Enrichment counts and scores of features in gibbon breakpoint flanking re- | |
| | gions. | 110 |

List of Figures

| 1.1 | Similarity of genomic breakpoints that occur in cancer and evolution | 3 |
|-----|--|----|
| 2.1 | An example of a computational pipeline for high-throughput short-read DNA resequencing projects. | 13 |
| 3.1 | Detecting deletions and insertions from the distance between mappings of | |
| | paired reads. | 20 |
| 3.2 | The general algorithmic steps of a classic read-pair algorithm | 22 |
| 4.1 | The word count example MapReduce application. | 35 |
| 4.2 | Hadoop-enabled tools for tasks of three popular sequencing analysis pipelines. | 37 |
| 4.3 | An overview of the steps of the MapReduce SV detection workflow | 42 |
| 5.1 | An Example of the Cloudbreak deletion and insertion detection algorithm running in MapReduce. | 50 |
| 5.2 | Illustration of insert size mixtures at individual genomic locations | 53 |
| 6.1 | Cloudbreak accuracy on a simulated data set | 64 |
| 6.2 | Runtimes for each tool on the simulated data set $\ldots \ldots \ldots \ldots \ldots$ | 65 |
| 6.3 | Scalability of the Cloudbreak algorithm. | 66 |
| 6.4 | Cloudbreak runtimes with varying window sizes. | 68 |
| 6.5 | Cloudbreak breakpoint resolution with varying window sizes. \ldots . | 70 |
| 6.6 | Accuracy on the 37X NA18507 sample | 71 |
| 6.7 | Runtimes for each tool on the NA18507 data set | 71 |
| 6.8 | Breakpoint resolution on the NA18507 data set | 73 |
| 6.9 | Cloudbreak performance on the simulated data set using different alignment | |
| | strategies. | 77 |
| 7.1 | A Hidden Markov Model for the SV detection problem | 83 |
| 7.2 | A linear chain Conditional Random Field model of the SV detection problem. | 85 |
| 7.3 | An example deletion with features from the NA18507 data set. \ldots . | 92 |
| 7.4 | An example of the labeling scheme used for CRF model training and testing. | 93 |
| 7.5 | ROC curve for CRF-confirmed Cloudbreak deletion calls. | 96 |

| 7.6 | Breakpoint resolution for deletions on the NA18507 data set, including |
|-----|--|
| | Cloudbreak CRF calls |
| 8.1 | Enrichment of genomic features in regions of the human genome syntenic to |
| | gibbon breakpoint regions identified using BACs |
| 8.2 | Enrichment or depletion of gibbon genome breakpoints for several features. 109 |
| 8.3 | Enrichment or depletion of regions of the gibbon genome for various features, |
| | as a function of distance from the breakpoint |
| 8.4 | Shift analysis of breakpoint flanking region overlap with various features 112 |
| 8.5 | An example of CTCF binding sites in a breakpoint region |

Abstract

Genomic structural variations are an important class of genetic variants with a wide variety of functional impacts. The detection of structural variations using high-throughput short-read sequencing data is a difficult problem, and published algorithms do not provide the sensitivity and specificity required in research and clinical settings. Meanwhile, high-throughput sequencing is rapidly generating ever-larger data sets, necessitating the development of algorithms that can provide results rapidly and scale to use cloud and cluster infrastructures. MapReduce and Hadoop are becoming a standard for managing the distributed processing of large data sets, but existing structural variation detection approaches are difficult to translate into the MapReduce framework. We have formulated a general framework for structural variation detection in MapReduce, and implemented a software package called Cloudbreak, which detects genomic deletions and insertions with very high accuracy compared to existing popular tools. Through the use of MapReduce and Hadoop, Cloudbreak can scale to harness large compute clusters and big data sets, leading to much faster runtimes than existing methods. In addition, we show that Cloudbreak's formulation of the structural variation detection problem in terms of local feature generation allows it to simultaneously integrate many informative signals in statistical learning frameworks. We demonstrate this using conditional random fields, which enable learning conditional probability distributions over labels on sequences of observations, and show that it improves Cloudbreak's results, in particular increasing breakpoint resolution.

In addition to the development of Cloudbreak and its extensions, we describe a data analysis project in which we examined the genomic features that occur near evolutionary breakpoints in the genome of the gibbon, whose karyotype is heavily rearranged compared to other primate species. Using a distributed pipeline to conduct Monte Carlo permutation tests, we find a statistical enrichment of segmental duplications, certain families of transposable elements, and evolutionarily shared binding sites of the protein CTCF near the locations of gibbon rearrangements. These findings may help us understand the process by which structural variations formed and were preserved in the gibbon lineage.

Chapter 1

Introduction

The aim of the field of *genomics* is to characterize the structure and function of the DNA of an organism or population of organisms, with the ultimate goal of understanding how the sequences of nucleotides that make up the genome affect phenotypes and reveal evolutionary history. To do so, it is necessary to identify and understand the differences between the DNA from two samples, whether the samples come from two different individuals or two different tissues from the same individual. Given that each sample can contain DNA from multiple cells, and each cell in a human sample contains approximately 3 billion DNA bases packaged into two sets of 23 chromosomes, this is difficult and complex undertaking.

If, as is the case for humans, the species has been widely studied, a *reference genome* is often used in place of one of the samples. This allows variations between individuals to be described as variations between the sample and the reference. Experiments of this type are known as *resequencing* experiments. Assuming that a reference is used, or that the samples come from an individual or individuals from the same or closely related species, the majority of the DNA sequence will be the same between the two samples. In that case, variants can be categorized into one of several forms. The first is a difference of a single base of DNA at a particular location within the genome, or a *single nucleotide variant* (SNV). If the variation is shared between many individuals in a population, these are referred to as *single nucleotide polymorphisms* (SNPs). Another type of variation is the insertion or deletion of a small number of base pairs at a particular location, commonly referred to as *indels*. A final category of variants are genomic *structural variations* (SVs). This term describes variations that affect a large number of bases of DNA (in common usage at least 40 base pairs, ranging up to hundreds of megabases or entire chromosomes). SVs can take a variety of forms: strings of DNA can be deleted, inserted, duplicated, inverted, or translocated to a different chromosome. Because they can be large and frequent, SVs account for the majority of the bases that differ among normal human genomes [128, 42].

All of these types of variants can alter the function of a genome in different ways. SNVs can change the sequence of the proteins coded for by genes, sometimes altering their function and sometimes rendering them inactive, or they can alter regulatory elements and cause genes to be expressed at differing levels. Indels typically disrupt the function of a gene or regulatory element. SV's can cause a variety of functional changes, ranging from the deletion of exons to the formation of fusion genes such as the famous Brc-Abl Philadelphia chromosome in chronic myelogenous leukemia [87].

In fact, SVs are very common in some types of cancer, producing extremely rearranged genomes. Because of this, they have particular importance in cancer research, and it is therefore essential to be able to identify them in biological samples, as well as to characterize their functional impact and the mechanisms of their formation. The SVs that arise in cancer genomes have similarities to those that have arisen between different species in evolution (Figure 1.1). For example, the genomes of gibbon species, while still closely related to humans and other apes, have undergone a variety of chromosomal rearrangements and other structural variations when compared to these other species; many more, in fact, than are typical in species with similar levels of divergence. Understanding the evolutionary changes in gibbons, therefore, may shed light on the much faster processes that take place in progenitor cancer cells.

The current technology to detect genomic variations is massively parallel, high-throughput sequencing. This procedure involves amplifying the DNA from a sample and then shearing it into small fragments. The ends of those fragments are then sequenced, producing hundreds of millions or billions of *read pairs* for a sample using current, widely used instruments. The challenge of genomics is then to discover the variants present in the sample using only these short reads. By generating and sequencing enough fragments from a sample (quantified in terms of *coverage*, the average number of reads that cover any one locus in the reference genome), the hope is that it should be possible to capture almost all of the important variants in a given sample. Current practices suggest that 30X average coverage



Figure 1.1: Similarity of genomic breakpoints that occur in cancer and evolution. The regions around the circle represent the human chromosomes. Lines in the centers of the circles show rearrangements between and within chromosomes. A) Somatic rearrangements detected within a breast cancer cell line, adapted from [64]. B) Rearrangements with respect to the human genome sequence present in the gibbon genome, adapted from [29].

depth is required to achieve accuracy in detecting SNVs. Even with this level of coverage, however, the task is made difficult by errors in the sequencing process, the fact that mammalian genomes are filled with repetitive sequences that make it difficult to ascertain the location in the genome that generated a particular read pair, and the computational challenges of analyzing the volume of data generated.

While SNVs and indels can be characterized relatively well from sequencing data using current algorithmic approaches, the identification of SVs remains challenging. SVs often arise within repetitive regions of the genome, making it difficult to align the reads surrounding them unambiguously to the reference genome. In addition, when the boundaries of an SV (the *breakpoints*) fall within a read sequence, it can be difficult to map that read back to its point of origin in the reference sequence. Although some approaches are based on searching for reads that contain breakpoints, it is often necessary to fall back on other signals present in the read set: the distance between successfully mapped read pairs, which should match the size of the fragments generated in the sequencing protocol, and the depth of reads that cover individual loci along the genome. Detection of SVs with current high-throughput sequencing technology remains a difficult problem, with limited concordance between available algorithms and high false discovery rates [128].

While current approaches struggle with accuracy, they also often fail to consider speed and scalability. A 30X coverage data set for an individual sample, in compressed format with associated quality scores, contains over 100GB of data to analyze. A typical bioinformatic pipeline includes steps to run quality control checks on the raw data; align the reads to the reference genome; perform filtering and recalibration steps after alignment; call SNVs and indels; and finally search for SVs. While a great deal of effort has been put into developing, optimizing, and parallelizing fast methods for alignment and variant calling, SV detection algorithms have not received the same attention, primarily because research in SV detection algorithms has focused on improving accuracy. As large scale sequencing projects like the 1000 Genomes Project and The Cancer Genome Atlas grow, the need for fast and accurate algorithms is becoming more apparent. DNA sequencing is already moving into the clinic, which will only exacerbate this requirement by requiring rapid analysis turnarounds for patients.

One approach to scaling data analysis pipelines is to harness the power of distributed computing using frameworks that tie together clusters of servers. Google's MapReduce [45] framework was designed to manage the storage and efficient processing of very large scale data sets across clusters of commodity servers. Hadoop is an open source project of the Apache Foundation which provides an implementation of the MapReduce programming framework as well as a distributed file system (HDFS) for distributing the redundant storage of large data sets across a cluster. Hadoop and MapReduce are rapidly becoming a standard in industrial data mining applications. However, they require the use of a specific programming model, which can make it difficult to design general-purpose algorithms for arbitrary sequencing analysis problems like SV detection.

This dissertation presents several novel techniques for detecting SVs using distributed computing and machine learning, which derive from the development of an algorithmic framework for SV detection methods in MapReduce. In particular, the main contributions are:

• The description of an algorithmic framework for solving SV detection problems in

Hadoop and MapReduce based on the computation of local features along the genome from paired end mappings (Chapter 4).

- The development in this framework of a software package, Cloudbreak, for discovering genomic deletions up to 25,000bp long, and short insertions, which improves accuracy over existing approaches and uses distributed computing to achieve dramatically faster runtimes (Chapter 5 and Chapter 6).
- An evaluation of the strengths and weaknesses of Cloudbreak, as compared to several other popular SV detection tools, when tested on several real and simulated data sets (Chapter 6).
- An exploration of the use of local features as described in Chapter 4 to reformulate SV detection as a sequence labeling problem, and the corresponding implementation and evaluation of a conditional random field model to create a novel method for integrating different signals of structural variations (Chapter 7).

In separate work not related to the algorithmic developments listed above, we also present the results of data analysis projects which examined the SVs that have massively rearranged the genome of the gibbon in an evolutionary time frame. This has led to the additional contribution of:

• Identification of sets of genomic features that are enriched near the breakpoints of the structural variations are present between gibbons and humans. These include segmental duplications and some families of transposable elements, as well as evolutionarily shared transcription factor binding sites. This analysis enhances our understanding of gibbon genome rearrangements. (Chapter 8).

A preliminary version of parts of this work was peer-reviewed and accepted for oral presentation at the Third Annual RECOMB Satellite Workshop On Massively Parallel Sequencing (RECOMB-seq). A preliminary breakpoint analysis was published in [28]. Throughout this document, I have tried to explicitly identify any work that was carried out by my collaborators.

Chapter 2

Biological Background

In this chapter we will explore the nature of structural variations, discuss the mechanisms that may be behind their formation, and their effect on phenotypes and disease. We will then describe high-throughput sequencing and discuss the typical informatics pipelines that researchers create to analyze sequencing data, placing SV detection within the context of the other types of computation that must be done to support end-to-end analysis. Finally, we will discus the challenges that the increasing number and size of sequencing data sets represent.

2.1 Structural Variations

As we described in the introduction, structural variations (SVs) are rearrangements, relative to some reference genome sequence, of sequences of DNA with a length longer than 40 or 50bp. These can take the form of deletions of reference sequence, insertions of novel sequence, duplications of sequence, inversions of orientation of a stretch of sequence with respect to the rest of its chromosome, or translocations of sequence from one chromosome to another. Deletions, insertions, and duplications are also frequently referred to as *copy number variations* (CNVs). In addition, many complex SVs exist in which these simple SV types are combined.

SVs can occur and become part of a gene pool on several different biological and time scales. On the smallest level, SVs can rearrange the genomes of individual cells within an organism. If the SV contributes to the cell's ability to proliferate, or occurs in conjunction with another mutation that does so, it can give rise to a tumor, or a new subpopulation or clone within an existing tumor. On a population scale, the genomes of individuals within a species harbor SVs with respect to one another, either because they were transmitted through the germline or because they arose *de novo* in certain individuals. Finally, on a larger scale, SVs within a species may become fixed, contributing to differences between the genomes of varying species on an evolutionary scale.

2.1.1 SV Effects on Phenotype and Disease

On all of these levels, SVs are associated with differences in phenotypes and have been shown to associate with many types of disease, making them an important area of study. Of course, our knowledge of these effects has been shaped by the types of SVs that researchers have been able to detect in the past several decades given the assays at their disposal and the scale at which samples could be analyzed. These limitations have biased our understanding towards the large variants with large effect sizes that were discoverable before the advent of modern array and sequencing technologies.

Cancer genomes contain many SVs that have been linked to tumor progression. The famous Brc-Abl mutation [87] mentioned previously is the most famous example of a fusion gene, in which portions of the genome coding for two different genes are made adjacent due to an SV. This results in the creation of new gene which can take on novel functions, which in turn drive cancer growth. Fusion genes can be caused through any type of SV that causes a novel adjacency between regions of the genome, including deletions, duplications, translocations, and inversions, and there have been dozens of recurring fusion genes identified across varying cancers to this point [10]. In another example, the genomic locus that contains the MYC gene, which among other functions regulates cell proliferation [50], is frequently duplicated multiple times in breast cancer cells [55]. These duplications change the amount of protein made by the cell, causing a dosage-related defect in cell regulation. In other cancer cells, a dramatic process called *chromothripsis* has been observed, in which entire chromosomes appear to "shatter" and then be pieced back together. This gives rise to hundreds of SVs, some of which can promote cancer development [179]. On the other end of the spectrum in terms of variant size, small deletions of less than 150bp in untranslated coding regions of genes have been observed to contribute to aberrant gene expression in leukemia [73].

On the scale of the human population, SVs and CNVs have been implicated in a variety of diseases (for a comprehensive recent review see Weischenfeldt et al. [191]). The best studied of these are large CNVs, including deletions or duplications of genes. These can can include variants that arise *de novo* in individuals, either giving rise to rare disorders [115] or contributing to the risk of more common conditions such as Crohn's disease [122], autism [169], and schizophrenia [188]. However, most CNVs in the human genome are polymorphic within the population and contribute to human genetic diversity [123]. The rise of microarrays and high-throughput sequencing has sped the discovery of smaller variants that affect coding sequence by deleting exons. To give one recent example, an 8kb deletion of an exon in the BAG3 gene has been tied to risk for cardiomyopathy [141].

There are also numerous examples of deletions that affect human phenotypes even though they do not directly overlap with the coding regions of genes, instead interacting with regulatory elements that control gene expression levels. For example, deletions ranging from 36kb to 319kb in size that are over one megabase away from the SOX9 gene locus can alter its expression by removing an enhancer element, giving rise to a rare cleft palate syndrome called Pierre Robin sequence [21]. Expanding on this idea, a survey of CNVs between inbred mouse strains showed that 28% of gene expression differences between strains could be statistically explained by CNVs, but that only 7% of the identified CNVs directly overlapped one of the differentially expressed genes, suggesting that SVs may shape many aspects of organism phenotypes in complex ways [26]. In another dramatic example from other species, the well-studied stickleback fish, which has repeatedly evolved adaptations to fresh water in several locations around the world as it moved from the sea into lakes, owes one of those adaptations, the loss of spines on its pelvis, to recurrent sub-2kb deletions of an enhancer that regulates the Pitx1 gene [31].

2.1.2 Mechanisms and Signatures of SV Formation

Given that many of the phenotypic effects of SVs described above are deleterious to the organism, it is natural to ask how they arise in genomes. Biologists have identified several different mechanisms that can cause SVs. By studying many of the genomic disorders

mentioned in the previous section, geneticists found that often CNVs recur in different individuals at the same locations in the genome. Examining these locations revealed that they are homologous to sequences repeated elsewhere in the genome in the form of segmental duplications, stretches of DNA that are over 1kb in length with greater than 90%sequence identity [170]. This helped to identify the process of non-allelic homologous recombination (NAHR) [112]. In the best studied form of NAHR, the cell attempts to repair DNA in the cell that has suffered double-stranded breaks (DSBs). To do so, it attempts to find homologous DNA in the nucleus that can be used as a template to repair the broken strand. However, repeats in the genome provide the opportunity for DNA from the wrong genomic location to be used in this repair process, creating genomic breakpoints in the repaired DNA. In mammals, as little as 295bp of non-allelic homologous sequence can be responsible for NAHR events [111]. The human genome contains many copies of transposable elements, including 500,000 of the LINE family and over one million of the Alufamily, with average lengths of 6kb and 300bp, respectively. Both Alu [104] and LINE [158] are frequently involved in NAHR events. As these and other transposons copy themselves into new locations in the genome, they also give rise to their own class of SV, mobile element insertions (MEIs), which leave their own signatures of small tandem site duplications (TSDs) at either end of the insertions.

In contrast to NAHR, other mechanisms can create SVs without extensive sequence similarity at the breakpoint. Non-homologous end joining (NHEJ) and microhomology mediated end joining (MMEJ) are two such processes [67]. Both of these mechanisms involve the repair of DSBs in the cell and typically result in SV breakpoints that have very small (less than 20bp) homologies on either side. Other processes of SV formation that leave a signature of microhomologies at the breakpoint are Fork Stalling and Template Switching (FoSTeS) [101] and microhomology-mediated break-induced replication (MMBIR) [67]. In these models, as the cell replicates all of its DNA two replication forks can switch DNA templates, causing complex structural variations to form. Finally, the chromothripsis process mentioned above also gives rise to microhomologies [113].

Given that different SV formation mechanisms have different signatures that can be determined by the presence of varying amounts of duplication or homology at or near the breakpoints, it is possible to analyze SVs to gain insight into their formation in different contexts. For example, Conrad et al. [41] surveyed CNVs from 40 unrelated individuals using a targeted sequencing method that enabled them to determine exact sequences around the breakpoints of some events. They found that although the best studied human genomic disorders are due to recurrent SVs caused by NAHR, NHEJ and MMEJ were responsible for the great majority of CNVs they were able to detect, a result later confirmed by the 1000 Genomes Project [128], although studies in mice have found that MEI may be more frequent albeit harder to detect [197]. In Chapter 8, we will discuss efforts to analyze the sequence context of genomic rearrangements between species.

2.2 High-Throughput Short-Read Sequencing

The pace of analysis of genome rearrangements and SVs has risen dramatically with the widespread adoption of high-throughput short-read sequencing (for a review see Shendure and Ji [172]). In early projects to interrogate SVs in DNA samples researchers had to painstakingly map individual variants using time-consuming microscopy (for very large rearrangements), fluorescence in situ hybridization (FISH), southern blotting, or polymerase chain reaction (PCR) assays [14]. The development of microarray technology allowed simultaneous testing of many genomic probes on a sample, although the probes had to be developed using prior knowledge of what variants to expect. Short-read sequencing, however, allows direct DNA sequence data to be collected in a relatively unbiased way from the entire genome, overcoming these limitations.

High-throughput short-read sequencing technology is currently dominated by Illumina, Inc. (San Diego, CA), and unless otherwise noted we will be referring to data generated by Illumina instruments when we refer to short-read or next-generation sequencing data in this document. The Illumina sequencing process begins with creation of genomic DNA library, which has billions of small fragments of DNA, sheared randomly across the genome, with adapter sequences attached to the ends. These adapters attach randomly to locations on the surface of the sequencing instrument's *flow cell*, and then are amplified in place to form clusters on the flow cell, each consisting of thousands of copies of a DNA fragment from the library. The sequencer then initiates a process in which the complements of each strand of DNA in the clusters are synthesized one base at a time. Each complementary base added to the new molecule contains a fluorescent tag, and the instrument takes a picture of the flow cell at each step. Pooling the color signal from each strand in a cluster, the sequencer optically analyzes the image from each time step to call the base at that position in each cluster. Because all clusters can be analyzed simultaneously in a single flow cell image, Illumina's sequencing instruments can now process billions of fragments at the same time. Illumina's HiSeq 2500 instrument, the current throughput leader, can produce 1.2 billion paired-end reads, or 120Gb of data, in 27 hours.

While this technique produces unprecedented amounts of sequencing data, the downside is that read lengths are limited by the extent to which the synthesis process can be regulated and kept in sync across all of the clusters on the flow cell. In Illumina's current highthroughput instruments, this limits read lengths to 100bp or 150bp. The downsides of this can be mitigated somewhat through techniques that sequence both ends of a DNA fragments, producing the paired-end reads referred to earlier. If the size of the fragment is approximately known, the distance between the two sequenced ends can be estimated, as we will discuss in detail in Section 3.2. These paired-end techniques typically use fragments of between 250 and 500bp, although in some cases *mate pair* fragment sequencing can produce reads from the ends of much larger fragments (6-8kb).

Because nearly half of the human genome is made up of repetitive elements, the analysis of the short reads produced by these sequencing technologies present a unique set of challenges [182]. As we mentioned earlier, repetitive elements such as *Alu* and LINE appear many times in the genome. If a read, or pair of reads, falls entirely within a single repetitive element, it will be impossible to determine the location in the genome it came from if there has not been enough sequence divergence of the repeat over evolutionary time to distinguish it from the others. Because many SV formation mechanisms depend on sequence homologies at the breakpoints, they are particularly likely to fall into repeats in the genome, complicating the task of SV detection from short-read data. In addition to making SV detection difficult, short-read sequencing – by the unprecedented volume of data it produces – also presents challenges to developing scalable infrastructure and algorithms for processing sequencing data in general. We will return to the problem of SV detection in the next chapter, and in the remainder of this chapter will give an overview of the applications of high-throughput sequencing data, and will discuss the computational challenges they represent.

2.2.1 Sequencing Analysis Pipelines

To transform raw sequencing reads into usable information typically requires a series of discrete steps. As genomic bioinformatics has matured, these have been increasingly modularized and made into components that can be connected together to create analytics pipelines designed for particular applications. In this section, we will describe pipelines for several of the most popular applications of high-throughput sequencing.

DNA Resequencing

As we mentioned in Chapter 1, resequencing projects attempt to identify variants present in the genomic DNA of samples from a population for which there exists a reference sequence. Variants can then be specified in standard formats (typically VCF, the Variant Call Format [44]) that describes them in terms of the sequence and coordinates of the reference. These can then be used in study designs such as case/control or genome wide association studies (GWAS), or, in clinical contexts, can be searched directly using prior knowledge for variants likely to be associated with diseases or conditions.

Figure 2.1 shows a high-level representation of the steps involved in a large-scale resequencing project using the Genome Analysis Toolkit [46]. The main components are:

1. Quality Control and Data Preparation. Although not shown in Figure 2.1, most pipelines typically begin by running a set of reports to assess the quality of the raw data, often with a standardized QC suite such as FastQC [8] and data preparation steps that include trimming the portions of the input sequences that have base calls with low confidence scores or that contain sequence that matches known artifacts of



Figure 2.1: An example of a full computational pipeline for high-throughput short-read DNA resequencing projects. This figure describes the Genome Analysis Toolkit (GATK) best practices pipeline, and is representative of the computational steps involved in a full resequencing pipeline designed to process many samples simultaneously. Reprinted by permission from Macmillan Publishers Ltd: Nature Genetics 43:5pp491-498 [46], copyright 2011

the sequencing process (adapter trimming). These components operate on raw data in the FASTQ format [40].

2. Read Mapping. In this phase, reads are mapped to the reference genome to find their most likely coordinates of origin. There exist a wide variety of short-read mappers that are heavily optimized for the task of aligning short DNA sequences to reference genomes (for a survey from several years ago see Li and Homer [108]). These typically use some form of a seed-and-extend strategy, in which exact matches of portions of the input read to the reference genome are found quickly using precomputed indices. These indices can take the form of hash tables or some form of enhanced suffix array, or more popularly, the suffix trie equivalent FM-index [58], which uses the Burrows-Wheeler Transform (BWT) [25] to convert a suffix trie into a highly compressed structure that is particularly useful for indexing highly repetitive genome sequences while still giving fast (under most conditions linear in the length of the query pattern) lookup times. These exact matches are then extended using direct comparisons or through dynamic programming techniques such as the Smith-Waterman alignment algorithm [176]. Examples of hash-based aligners include Novoalign [142] and MOSAIK [103], while BWA [106] and Bowtie 2 [94] are popular BWT-based aligners. The output of this step is typically in the Sequence Alignment/Map (SAM) format [107], which in addition to the reads stores their alignment coordinates on the reference genome, mapping quality scores, and information about mismatches to the reference.

- 3. Local Realignment, Duplicate Marking, and Recalibration. In some pipelines there is often an additional quality control step after read mapping. PCR amplification in the preparation of sequencing libraries from DNA samples can give rise to duplicate fragments, which are typically identified and removed at this point. In addition, this step can include a more expensive realignment phase for groups of reads that appear to span short insertions and deletions (indels, typically under 40bp in size), and a recalibration of the base call quality scores reported by the sequencer based on the empirically observed distribution of mismatches between the reads and the reference sequence [46].
- 4. Variant Calling. The next phase includes calling of single nucleotide variants (SNVs), indels, and SVs. To call SNVs and indels, algorithms examine each possible variant indicated by mismatches of reads to the reference. They typically evaluate the number of reads supporting the variant against the total number of reads at that location, as well as prior beliefs, in a Bayesian framework. Popular SNP and INDEL callers include the GATK [124], SAMtools [107], and SOAPsnp [109]. In some cases, multiple samples from a population are examined simultaneously, giving additional power to detect variants in that population. We will discuss SV callers in more detail

in Chapter 3.

5. Integrative Analysis/Filtering. After raw variants have been called, additional sources of outside information can be used to adjust their confidence scores or filter out false positives. This can include data from existing catalogs of genomic variants or knowledge about the population structure of the samples. If multiple samples from the same pedigree are being sequenced, data can be reconciled based on the principles of inheritance. DePristo et al. [46] describe one such set of filters appropriate for a large scale study involving many individuals from a population. In Section 3.7 we will describe a simple filtering scheme we implemented for a cancer sequencing project.

Other Sequencing Applications

In addition to DNA resequencing to detect genomic variants, there are many other applications of short-read sequencing. We will not go into detail on how each works, but will briefly list the goals and major informatic components of several of the most common so that we can address a full picture of the computational scaling challenges that short-read sequencing represents.

- **RNA-seq.** In RNA-seq (see Oshlack et al. [146] for a review), the goal is to interrogate the expression of genes across samples, for example in different disease conditions. Rather than sequencing DNA, libraries are constructed from RNA taken from the sample. Similarly to DNA resequencing, the reads are QC'd and aligned to the reference genome, although there are read mappers especially designed for RNA-seq that include functionality to map reads over exon junctions. The reads that map to annotated genes on the reference are then analyzed and quantified to determine the genes that are being expressed in the sample, along with their relative expression levels and isoforms.
- ChIP-seq The intent of ChIP-seq (reviewed by Park [147]) is to determine the locations in which DNA, or the chromatin that contains it, is being bound by protein transcription factors or modified, with the ultimate goal of determining the modifications that are driving gene expression. In this case fragments of DNA that are

bound to the protein of interest are preferentially separated and sequenced. After read trimming and mapping, the locations of binding events can be identified by searching for "peaks" of coverage on the reference genome.

• de novo Assembly When there is no reference genome present for the organism being sequenced, the goal is to try to reconstruct the full sequence of the input genome based only on the short reads. A full discussion of de novo assembly algorithms is beyond the scope of this section; for a review see Nagarajan and Pop [135]. The general approach is to build a graph representing overlaps between the reads in the form of either a string graph [133] or a de Bruijn graph, and then walk the graph to determine the underlying sequence. This is complicated by the fact that mammalian genomes are highly repetitive. The need to construct a graph data structure linking all of the input reads means that assembly is typically an extremely memory-intensive process, making it difficult to distribute computation.

There are of course additional sequencing applications, including workflows specifically designed for cancer-related projects and metagenomics, which aims to discover all of the microorganisms present in a diverse sample.

2.2.2 Big Data from Sequencing

All of these sequencing applications are creating an enormous amount of data that is pushing the limits of the computational infrastructure and algorithms used by biologists. For example, the European Bioinformatics Institute currently stores multiple petabytes of genomic data, and that amount has been taking less than one year to double since the advent of high-throughput sequencing in 2008 [120]. This has led to increasing calls for new bioinformatics strategies for storing, managing, sharing, and processing that data. An early recognition of the fact that computational requirements would soon outstrip the capacity of the prevailing model of processing data on local servers in individual laboratories created a movement towards centralized databases and processing on remote servers enabled by standardized data formats and annotations [177]. After the creation of large scale cloud compute services, bioinformatics leaders quickly realized their potential to ease the burden of processing sequencing data [178, 164, 162]. In particular, they noted that cloud computing would allow the scaling of compute infrastructure to meet demand, rather than having to plan for, acquire, and maintain compute clusters capable of handling the predicted peak load. In addition, cloud computing offers a potential solution to the problem of sharing large data sets in the highly collaborative field of genomics; by storing large data sets in the cloud that are accessible to multiple groups, collaborators could become less dependent on the prevalent strategy of sharing data by shipping hard drives to one another. Several of these arguments also noted that cloud computing is a natural fit for creating distributed algorithms that can be scaled to larger data sets by adding more compute resources, in particular through the use of the MapReduce programming model [164, 162]. We will return to the topic of cloud computing in Chapter 4, where we will develop a framework for detecting SVs in MapReduce. Before we do so, however, we will review existing algorithms for SV detection in Chapter 3.

Chapter 3

Algorithms for Structural Variation Detection

In Chapter 2, we described the biological impact of SVs, and the development of short-read sequencing technology, along with the computational challenges they represent. In this chapter we review existing algorithms for SV detection based on short-read sequencing data. We will discuss the four major computational strategies for SV detection from sequencing data, and review specific approaches, with an emphasis on algorithms that use paired sequencing reads as a primary source of information. Finally, as a demonstration of how SV detection can fit into the pipeline models we described in Chapter 2, we will describe a small pipeline for SV detection that we developed that used an existing tool, BreakDancer, to characterize SVs in a low-coverage cancer sample.

3.1 The Four Signals of SVs in Sequencing Data

Most popular SV detection algorithms can be placed into one of four categories based on the type of signal they extract and analyze from sequencing data sets [4, 83]. The first three categories depend upon first aligning short reads to the reference genome. In Section 2.2, we described paired-end sequencing, in which both ends of a single DNA fragments are sequenced. Read pair (RP) based methods use the distance between and orientation of the mappings of these sequenced ends to identify the signatures of SVs. Out of all SV detection techniques, RP approaches are the most commonly used based on their ability to detect many types of SVs and their computational tractability [4]. Read depth (RD) approaches identify regions of the genome with anomalous coverage by read mappings, which may indicate the presence of deletions or duplications, a subcategory of SV known as copy number variations (CNVs). Split Read (SR) approaches attempt to find local mappings of portions of individual reads that span SV breakpoints. Finally, assembly-based methods attempt to construct as much of the genome sequence as possible directly from the reads, without first mapping them to the reference genome. The constructed sequence is then compared to the reference to identify SVs. Beyond these four categories, several recent approaches have attempted to integrate more than one type of signal to increase accuracy. In this review of published algorithms, we will focus on those that are built to handle whole-genome DNA resequencing from a single sample at a time; there are of course other approaches for different applications, including SV detection from targeted resequencing data such as exome sequencing, and approaches which are designed only to operate on many samples at once.

3.2 Read Pair Approaches

Given correct mappings of paired reads to the reference genome and a reliable estimate of the length of the fragments from which the reads came, it is relatively straightforward to detect SVs. For example, Figure 3.1 shows the signatures of insertions and deletions based on the insert size of paired-end mappings: deletions lead to a longer distance between read mappings than expected, insertions to a shorter distance. Similarly, because protocols determine the orientation in which the read will be sequenced, inversions can be detected if they diverge from expectations. Finally, inversions can be detected if the reads map to different chromosomes in the reference genome. These ideas were first applied to highthroughput short-read sequencing data by Korbel et al. [85] and Campbell et al. [27], who built upon strategies developed for analyzing SVs through the mapping of sequenced ends of larger DNA fragments (bacterial artificial chromosomes (BACs) and fosmids) from non-high-throughput sequencing data [187, 156, 184].

Most read pair approaches use a basic strategy outlined in Figure 3.2. They begin by separating paired end mappings onto the reference genome into those that are *concordant* and those that are *discordant*, i.e. those which deviate from the expected insert size or



Figure 3.1: Detecting deletions and insertions from the distance between mappings of paired reads. For the case of a deletion, a fragment with a 310bp internal insert size is created from the sample DNA, in a region in which 250bp has been deleted relative to the reference. The two reads when mapped to the reference will be 460bp apart, providing evidence for a deletion if we expect the internal insert size to be approximately 300bp. A 310bp internal insert size fragment that overlaps a 150bp insertion in the sample, conversely, will give a distance between reads of 180bp when mapped to the reference.

orientation of the library. With only a few exceptions, discussed below, RP approaches discard concordant pairs and consider only discordant pairs for the remainder of their analysis. The next step is then to cluster the discordant mappings such that each cluster coherently supports a single candidate SV call. Finally, they filter or rank the candidate SV calls, for example only keeping those with support from multiple discordantly mapped read pairs.

The BreakDancerMax component of BreakDancer [37] is probably the most widely used of these algorithms¹. As described above, BreakDancer first searches for discordant read pairs based on a hard threshold on the distance between mapped paired reads. It then looks for regions of the genome that anchor more discordant read pairs than expected according to its model; if two of these regions are connected by a minimum number of discordant read pairs, it calls an SV that links them. It calls breakpoints based on the inner boundaries of the two connected regions, as illustrated in Figure 3.2 (d), and then

¹Based on a sampling of Google Scholar citation counts as of 12/28/2013: BreakDancer, 270; VariationHunter, 153; PEMer, 111; GASV, 65; SVDetect, 46.

assigns a confidence score by computing the likelihood of the SV based on a null model in which discordant read pairs are distributed across the genome according to a mixture Poisson model. Other seminal RP-based tools include GASV [173], PEMer [84], Variation-Hunter [72], HYDRA [154], and SVDetect [202]. These all operate on similar principles, differing primarily in the method used to cluster discordant read pairs that support the same potential SV call, and in the filtering steps used to select candidate SV calls. For example, GASV formulates the clustering step in terms of a geometric model in which the goal is to find the intersection of the coordinates of breakpoints that each discordant read pair could possibly support. Marschall et al. [118] insightfully observed that most of the popular clustering procedures are computationally equivalent to solving some variant of the problem of finding maximum cliques in a graph structure in which nodes are read pairs, and edges exist if the pairs could possibly support the same variant. VariationHunter and HYDRA were also the first approaches to also integrate ambiguously mapped discordant read pairs, a topic we will consider in the next section.

Read pair approaches have the advantage of being theoretically able to detect any type of SV except for multiple copy number duplications. Their disadvantages stem from the fact that they depend on comparing mapping distances between reads to the unknown size of the fragments from which they came. This means that they cannot capture the breakpoints of SVs with single nucleotide resolution, and that they depend on having a sequencing library with a tight distribution of fragment sizes in order to have power. In addition, the fact that many SV breakpoints occur in regions with high sequence similarity to other parts of the genome, as discussed in Section 2.1.2, means that it may be difficult to unambiguously map reads to those locations.

3.2.1 Ambiguously Mapped Read Pairs

Many of these approaches use only reads that are unambiguously mapped to the reference genome; this has the advantage of using the same set of alignments that are used for calling SNVs and indels in most sequencing pipelines. A second group of RP methods attempt to include discordant read pairs which cannot be unambiguously mapped to the reference genome in their analysis, in an effort to improve sensitivity in repetitive regions



Figure 3.2: The general algorithmic steps of a classic read-pair algorithm. a) The algorithm accepts a set of alignments of paired reads to the reference as input. b) The algorithm identifies discordant read pairs. c) Discordant pairs are clustered to find groups that could support the same algorithm. d) Clusters of discordant read pairs are filtered (in this case, by the number of supporting read pairs), and bounds on the potential breakpoints are identified.

of the genome. One approach to incorporating this type of information can be found in *soft clustering* algorithms, which assign each ambiguously mapped read pair to one of its mappings such that it clusters with other discordant read pairs. These approaches include VariationHunter [72], which allocates ambiguously mapped reads by optimizing a maximum parsimony explanation of all discordant reads; HYDRA [154], which takes a similar approach based on heuristics; and GASVPro [174], which uses a Markov-Chain Monte Carlo sampling strategy to assign a read to its correct mapping. Even though they do consider more information than methods solely based on concordant reads, most methods in this category (with the exception of VariationHunter) use only a limited number of ambiguous discordant mappings per read pair, in part because of the storage and computational requirements necessary to process all or most ambiguous mappings of each read pair in a high-coverage data set. In Chapters 5 and 6, we will show that the MapReduce distributed computing framework has the potential to help address these challenges.

3.2.2 Concordant Read Pairs

The RP approaches listed above only consider the mapped insert sizes of discordant read pairs. (GASVPro does consider some concordant mappings in its final output, which we will mention in Section 3.6, but only in the sense of computing an RD signal rather than actually considering the insert sizes of the concordant pairs themselves.) Since discordant read pairs make up only a very small fraction of the mapped reads in a typical sequencing data set, these approaches are not considering much of the data available. Several algorithms have tried to use the information present in concordant read pairs to generate SV predictions. One simple example can be found in ChopSticks [199], which refines the breakpoint locations of deletion calls based on discordant pairs by looking for concordant pairs that overlap the ends of the predicted variant. This approach only works for homozygous deletions, however, and is therefore limited in scope.

More comprehensive attempts to include concordant RP information were described in MoDIL [102] and the BreakDancerMini component of BreakDancer [37]. MoDIL models the distribution of insert sizes at candidate locations in the genome using a Gaussian mixture model (GMM) describing the insert sizes observed; deletions and insertions are represented as additional components in the mixture. This has two advantages: because reads are not categorized as concordant or discordant based on a hard threshold, it is possible to detect smaller insertions and deletions; and these approaches can explicitly model the zygosity (presence of the variant on one or both of the pairs of chromosomes in the cell) of the variant in the sample, and potentially classify the variant as homozygous or heterozygous. The disadvantage of this approach in these implementations has been the computational requirements. BreakDancerMini considered only concordant read pairs, after processing discordant pairs in the BreakDancerMax. For each location in the genome they executed a sliding window Kolmogorov-Smirnov test comparing the distribution of concordant insert sizes to that of the entire library. This limited the approach to detecting only very small variants; the authors of BreakDancer no longer recommend running BreakDancerMini, presumably because of its computational requirements, instead suggesting SR-based approaches such as Pindel [200] to find smaller insertion and deletion variants. In Chapters 5 and 6, we will show that a strategy that considers the RP signal present in concordant read pairs can be made to give highly accurate result with very fast runtimes through the use of an algorithm developed in the MapReduce distributed computing framework.

Finally, CLEVER [118] took an alternative approach based on constructing a graph

linking all paired reads, both concordant and discordant, that could support the same allele at a position, and then clustering reads based on finding maximum cliques in the graph. As mentioned above, they showed that this is conceptually similar to the clustering approaches used by other algorithms for discordant reads. They demonstrated that their approach was superior at detecting smaller variants because of its lack of hard thresholds for discordancy, and provided a relatively efficient implementation that can process a full 30X genome in approximately 8 hours.

3.3 Read Depth Approaches

Read-depth (RD) approaches consider the changing depth of coverage of concordantly mapped reads along the genome to infer the presence of SVs. For example, a homozygously deleted region will have zero coverage in the reference genome, while a region that has been duplicated many times, as can happen in some regions of the genome and in some cancers, will have a much higher coverage than average. These approaches differ mainly in the statistical and signal processing techniques used to identify anomalous regions. For example, CNVnator [2] uses a mean-shift approach to segment the genome into CNV regions. Other approaches in this category include MrFAST [5], Event-Wise Testing [201], and SegSeq [38].

RD approaches are good at finding large deletions and duplications. As previously noted, they are the only approach that can identify segments of the genome that have been duplicated multiple times. Their disadvantages are their lack of ability to reliably detect smaller events, and their breakpoint resolution, which is even lower than than of RP approaches.

3.4 Split Read Approaches

Split-read (SR) methods look for breakpoints within individual reads by mapping portions of the read to different genomic locations. Due to the computational challenge involved in aligning reads to the reference genome while allowing for very large gaps between portions of the read, they use different strategies to guide the search. Pindel [200] looks for paired reads in which one read in the pair aligned to the reference genome but the other did not. Supposing that the other read may contain a breakpoint, it searches the reference nearby for split read mappings. CREST [189] takes advantage of aligners that insert gaps at the ends of read alignments when there are many mismatches between the read and the reference, known as *soft clipping*. By looking for multiple alignments with soft clips at the same reference coordinate, it can identify breakpoints. SplazerS [51] did not use heuristics to guide the SR search, instead designing a unique mapping strategy based on mapping the prefixes and suffixes of reads independently. They showed that their unbiased search is more sensitive than other approaches, at the cost of greatly increased runtimes.

Split read approaches can identify SVs with high specificity and single base breakpoint accuracy. They are particularly good at detecting smaller variants. However, their sensitivity is limited by coverage and the length of the reads. As read lengths increase with advances in sequencing technology, they will play a larger role in SV detection.

3.5 Assembly-Based Approaches

An alternative approach to mapping reads to the species reference to discover variants is to first attempt to directly assemble the genomic sequence from which the reads were generated (AS approaches). This typically involves the construction of a *de Bruijn* graph to represent the overlapping k-mers in the entire read set, and then walking the graph to construct the longest possible unambiguous sequence of k-mers. Although most work in assembly is focused on *de novo* assembly, where there is no reference for the organism being sequenced, there have been attempts to find variants in a resequencing context by comparing the contigs that result from assembling reads to a reference genome. This strategy for detecting SVs was first demonstrated by Li et al. [110] using assemblies created at the Beijing Genomics Institute. One published assembly approach that is targeted at detecting variants including SVs is Cortex [75]. Cortex uses the reference to guide assembly with a colored de Bruijn graph structure, and can therefore identify SVs by walking colored paths in its graph. In general the task of creating *de novo* assemblies for human-sized genomes, however, is still an extremely involved and difficult process that requires a great
deal of technical knowledge, as well as extremely high-memory compute nodes.

In practice, AS approaches are more likely to be used to verify candidate SV calls generated by other approaches. In these workflows, the reads mapped to the reference genome near the site of the possible variant are put into a *de novo* assembly process. The resulting sequencing generated by the assembly process is then aligned to the reference to see if it supports the existing call. TIGRA [36] is a recently published assembly tool explicitly for this process; other groups have reported building in-house custom pipelines to accomplish this [152, 116].

While AS approaches can theoretically identify any type of SV, in practice assembly requires extremely high coverage (typically 100X). In addition, the computational requirements necessitate high-memory servers, making the task difficult to run on widely available, non-specialized hardware. Finally, genome assembly using short reads tends to collapse identical repeats, leading to a loss of visibility in repetitive regions of the genome and in segmental duplications [6]. Because SVs are enriched in these repetitive regions, this could potentially lead to many false negative calls, and potentially even false positives if repetitive regions are assembled incorrectly, as a loss of a set of repeats could look like a deletion.

3.6 Hybrid Approaches

Recently, many approaches have been published with the goal of combining more than one of the signals of SVs (RP, RD, SR, and AS) in order to improve accuracy. Table 3.1 lists those that we are aware of. In general these can be divided into three types: those that generate candidate SV calls using one primary signal, and then use a second signal to refine or filter those predictions; those that incorporate existing tools as modular components into a pipeline that then merges the candidate calls from each component; and those that have an integrative model of all three signals, typically using a feature-based statistical approach. We will review several examples of each category in the next sections.

| Algorithm | Primary Signal | Secondary Signals | Comments |
|-------------------|----------------|-------------------|---|
| GASVPro [174] | RP | RD | support RP predictions with RD signals |
| DELLY [157] | RP | SR | refine RP predictions with SR mappings |
| PRISM [77] | RP | SR | refine RP predictions with SR mappings |
| Meerkat [198] | RP | SR | refine RP predictions with SR mappings |
| SoftSearch [66] | RP | SR | support RP predictions with softclips |
| Bellerophon [68] | RP | SR | support RP predictions with softclips (translocations only) |
| SVSeq [203] | \mathbf{SR} | RP | support SR predictions by at least one discordant RP |
| CNVer [125] | RD | RP | support RD predictions with discordant RP mappings |
| inGAP-sv [153] | RD | RP | refine RD predictions with discordant RP mappings |
| Nord et al. [139] | RD | SR | support RD predictions with softclips |
| PeSV-Fisher [54] | RP | RD | support RP predictions with RD signals |
| SVMerge [194] | Pipeline | SR,RP,RD,AS | combines BreakDancer, Pindel, RDXexplorer, others |
| HugeSeq [90] | Pipeline | SR,RP,AS | combines BreakDancer, Pindel, BreakSeq, others |
| LUMPY [98] | Pipeline | SR,RP | modular approach, merges breakpoint intervals |
| iSVP [129] | Pipeline | SR,AS,RP | deletions only; merges GATK, Pindel, BreakDancer |
| Zinfandel [171] | Integrative | RP,RD | HMM based on RD, RP distances |
| forestSV [126] | Integrative | RP,RD | Random forest classifiers on RD and RP features |
| SVM^{2} [39] | Integrative | RP,RD | SVM classifiers based on RD and RP features |
| SVMiner [69] | Integrative | RP,RD | GMM clustering based on RD, RP features |

Table 3.1: A summary of published SV detection algorithms that combine more than one sequencing signal. Most use one primary signal and then either discard those predictions not supported by the secondary signal or refine the breakpoints of the primary prediction using secondary data. Pipelines independently run several modules based on different signals and then merge the results. Integrative approaches generate primary predictions based on a statistical model that includes features from multiple signals.

3.6.1 Support from Secondary Signals

One example of a tools that uses one of the four basic signals outlined above but then incorporates other signals into its algorithms to refine the call set is GASVPro [174]. GASVPro is primarily an RP based method, but it uses RD signals to validate its predicted breakpoints, assuming that coverage directly around the breakpoint, and in predicted deleted regions, should be reduced. DELLY [157] and PRISM [77], meanwhile, use RP based approaches to identify candidate SV regions, and then guide an SR search for the exact breakpoints of those SVs. Typically, these approaches that incorporate secondary signals improve the specificity of the primary approach, at the cost of a decrease in sensitivity.

3.6.2 Pipelines

SVMerge [194] and HugeSeq [90] are examples of pipelines that independently execute multiple algorithms of different types and then attempt to merge the results together. The latter ranked calls by the number of components that predicted them, while the former filtered and validated the union of the independent call sets using local assembly. While the integration of these approaches could detect any type of variant detectable by any individual algorithm, it is difficult to combine results from different approaches in a principled manner, and the large number of dependencies and complex parameterization and configuration required has prevented adoption of these pipelines outside of the laboratories in which they were created.

3.6.3 Integrative Models

Finally, several algorithms have been developed that integrate more than one signal at once in a statistical model to generate SV calls. All of these have created features based on components of RD and RP signals. For example, Zinfandel [171] created an HMM where the observations included RD and RP features, and the hidden states included differently sized deletions and duplications with hand-tuned transition probabilities. forestSV [126] and SVM² [39] created feature vectors and then trained machine learning classifiers on example data sets; the former demonstrated that it can be useful to include additional features related to genomic context in feature vectors for predicting SVs. In Chapter 7, we will demonstrate a novel integrative approach that incorporates many more features using conditional random fields.

3.7 An Example of an SV Detection Pipeline for a Cancer Dataset

To illustrate how these tools are used in practice, in this section we describe our experiences in developing a pipeline for SV detection using existing tools to try to detect genomic rearrangements in a low-coverage cancer data set. Our pipeline roughly follows the steps we outlined in Section 2.2.1. We will return briefly to this data set in Section 6.4.

The data came from samples taken from a patient with acute myelomonocytic leukemia (AML). Our collaborator Dr. Jeffrey Tyner collected peripheral blood under a written and oral informed consent process reviewed and approved by the Institutional Review Board of Oregon Health & Science University. Known cytogenetic abnormalities associated with this specimen included trisomy 8 and internal tandem duplications within the FLT3 gene. He then isolated cells for sequencing by separating mononuclear cells on a Ficoll gradient, followed by red cell lysis. Mononuclear cells were immunostained using antibodies specific for CD3, CD14, CD34, and CD117 (all from BD Biosciences) and cell fractions were sorted using a BD FACSAria flow cytometer. This enabled us to isolate cell fractions including non-cancerous T-cells (CD3+), malignant monocytes (CD14+), and malignant blasts (CD34, CD117+), which represent an intermediate state between the normal CD3+ T-cells and the CD14+ tumor cells.

We sequenced all three cell isolates on an Illumina Genome Analyzer II, producing 128,819,200 76bp paired-end reads for the CD14+ cells, 194,945,868 reads for the CD3+ cells, and 172,182,321 reads for the CD117+ cells. We then developed a pipeline to detect somatic SVs in the CD14+ tumor cells. This pipeline roughly follows the form described in Section 2.2.1 for variant calling from DNA resequencing data, although we were not interested in SNV calls and therefore did not include them in our pipeline. Our pipeline began with a QC step, in which we gathered statistics on read quality and possible duplication rates using FastQC [8], and then trimmed adapter sequence (artifacts of the Illumina sequencing process) using cutadapt [119]. We then aligned reads to the hGRC37 human reference genome using Novoalign [142]. We chose Novoalign because it has higher accuracy than many other aligners, at a cost of significantly longer runtimes [161]; we were able to make Novoalign runtimes managable by splitting input reads into chunks and simultaneously aligning them across a compute cluster using custom scripts to coordinate the HTCondor grid engine [181]. We then used Picard Tools [24] to identify and remove duplicate reads. To call SVs, we used BreakDancer [37], as a preliminary version of the simulations we will describe in Section 6.1.2 showed that it had slightly better accuracy than other the other tools we evaluated. By subtracting SV calls from the CD3+ normal cells from those called for the CD14+ tumor cells, we were able to identify a set of candidate somatic SV calls which might be associated with the patient's AML.

The QC and deduplication steps of our pipeline revealed that due to the small amount of input DNA in our samples, we had very high duplication rates in our sequencing data, with 15.69% and 85.54% of the reads being duplicates in the CD14+ and CD3+ data sets, respectively. This led to very low physical depths of coverage of the genome: 8.3X for the CD14+ cells and 1.3X for the C3+ cells. Nevertheless, BreakDancer still made over six thousand SV calls, far more than we could verify. Examination of selected calls, together with the knowledge that the cells' karyotype did not include gross rearrangements, showed that many were false positives due to incorrectly aligned reads in repetitive regions of the genome, particularly in areas with many simple repeats such as the regions near chromosome centromeres and telomeres. Therefore, we developed a filtering pipeline that used BreakDancer SV scores and genome annotations, including interval files containing the genomic coordinates of transposable elements, peri-centromeric and telomeric regions, and segmental duplications. This annotation pipeline is open-source and available at https://github.com/cwhelan/sv-annotate.

Using this filtering pipeline we were able to reduce the set of candidate SVs to a small number of high-quality calls that we could verify. This list included 38 deletions, 13 inversions, and 14 translocations. With the help of our collaborators Alberto L'Abbate and Tiziana Storlazzi at the University of Bari, Italy, we were able to test a subset of predictions by PCR. They designed PCR primers using Primer3 [185], considering the chromosome localization and orientation of the interval involved in the candidate rearrangement, and checked them for specificity using the BLAT tool of the UCSC Human Genome Browser [80]. All the primer pairs were preliminarily tested on the patient genomic DNA and a normal genomic DNA as control. The PCR conditions were as follows: 2 min at 95°C followed by 35 cycles of 30 sec at 95°C, 20 sec at 60°C, and 2 min at 72°C. All the obtained PCR products were sequenced and analyzed by BLAT for sequence specificity.

Overall, PCR confirmed nine out of eleven tested deletions, two out of seven tested inversions, and only one out of seven tested translocations. Nevertheless, the confirmed SVs included several that might be relevant to the patient's AML, including a deletion in the gene CTDSPL/RBPS3, an AML tumor suppressor [206]; and another deletion in NBEAL1, a gene up-regulated in some cancers [35]. We are currently investigating these deletions to determine their functional impact on this patient.

Our experiences in developing this pipeline illustrate several of the themes of this dissertation. Even for a low-coverage data set, to compute high-quality alignments using Novoalign we resorted to data parallelization over a compute cluster coordinated by custom scripting, but were unable to speed up the runtime of BreakDancer. In the next chapter, we will discuss methods to easy distribution of sequencing analysis pipelines over compute clusters and to the compute cloud, and develop a truly distributable algorithmic framework for SV detection. In addition, the high false positive rate of our candidate SV calls illustrates the need for more accurate algorithms; we will develop and evaluate a novel algorithm in Chapters 5 and 6, and discuss a possible method for increasing its accuracy in Chapter 7.

Chapter 4

A Framework for SV Detection in MapReduce

In the previous two chapters, we discussed the need for scalable approaches to processing high throughput sequencing data, and explored algorithms for detecting genomic structural variations while noting that efficient processing has not been a primary concern in their design. In this chapter we will explore the MapReduce computing paradigm and its open source Hadoop implementation. We will then discuss ways in which Hadoop has been applied to sequencing tasks. Finally, we will describe a general approach for implementing SV detection algorithms in MapReduce and Hadoop, which we will explore in more detail in later chapters.

4.1 MapReduce and Hadoop

MapReduce [45] is a parallel computing framework designed at Google to handle computation over very large data sets - in particular, the vast amounts of data produced by Google's web crawlers. These volumes of data are too large to store and access efficiently on single instances or storage arrays, and must therefore be stored in distributed file systems (DFS), in which portions of the data are stored on individual nodes distributed throughout the cluster, rather than on a central file server; MapReduce runs in conjunction with the Google File System (GFS) [60]. MapReduce was designed to simplify the development of applications that need to process large volumes of data that are distributed across clusters of many servers, while hiding the complexity of data and processing distribution and load balancing from the application developer. An overriding goal in its design was to develop a robust framework in scenarios where clusters are composed of hundreds or thousands of commodity machines, which may have high failure rates and heterogeneous performance characteristics. In particular, it is optimized to provide the following benefits automatically to applications built according to its programming model:

- Fault tolerance. Both data and processing tasks have redundancy built into the application framework due to the expectation of a certain failure rate among the nodes in the cluster. As mentioned earlier, data is distributed over the file system such that individual blocks of data reside on individual nodes in the cluster. Rather than storing single copies of each block, however, the DFS ensures that multiple copies of each block are stored in the file system on different nodes, and if a node fails, will re-replicate blocks to other nodes so that no data is lost. The same philosophy is also applied to the task scheduler/tracker, which breaks up jobs into many small tasks. If the tracker notices an unresponsive task, for example caused by a hardware error on an individual node, it will restart additional tasks to operate on the same input block of data. In this way jobs can continue processing and complete successfully even when worker nodes fail in the middle of their processing.
- Data locality. Taking note of the fact that the most expensive part of distributed computing is often transferring data over slower network connections, MapReduce makes every effort to schedule tasks on nodes that physically hold a copy of their input data. This philosophy of "bringing the code to the data" is key to processing large data sets quickly without saturating cluster networks.
- Scalability. The framework's goal is allow the size of both the data sets and the cluster to grow seamlessly. As more data is added to a data set, the DFS divides it into blocks and distributes it across available space on the cluster, avoiding bot-tlenecks in storage capacity, and increasing the number of nodes that have copies of portions of the data set locally. As more nodes are added to the cluster to increase its capacity, they automatically expand the capacity and have data replicated to them. Since tasks are also broken up into small independent pieces, new nodes can also be seamlessly incorporated into the cluster's processing jobs.

Clearly, the desirable properties of the DFS are dependent on being able to divide the data into small blocks that can be distributed easily around the cluster. Similarly, the scalability goals for processing depend on dividing up compute jobs into small tasks that can run independently on different cluster nodes. To achieve this, MapReduce requires that developers structure their application into functional components known, unsurprisingly, as Map and Reduce tasks. This structure is inspired by a functional technique from the Lisp programming language. In functional programming, map is a function that takes a single-argument function m and a list l as arguments and returns a new list which is the result of applying m to every element of l. Meanwhile, reduce is a function that takes a binary operator r and a list l, and returns the result $r(r(r(l_1, l_2), l_3), ...)$, applying r to every element of l and a running result. Many operations can be expressed as an application of map to a list followed by an application of reduce. For example, to find the sum of squares of a list of integers l, one would write reduce(sum, map(square, l)). Although the semantics of MapReduce's functions are slightly different than those of functional programming [91], its creators hoped to enable the same expressive power while ensuring that applications are decomposed into parallelizable pieces.

In MapReduce, Map tasks are responsible for examining every record in a block of the input set, and emitting information in the form of $\langle key, value \rangle$ pairs. Reduce tasks then take as input all of the values that were emitted by a mapper under a particular key and and produce one or more outputs that summarize or aggregate those values. In order to accomplish the necessary data handling to ensure that reducers receive all of the keys for a particular value, in between the map and reduce phase MapReduce executes a "shuffle and sort" procedure, in which each node sorts the output of all of its mappers by key, sends the results for each key to the machine on which the reducer for that key will run, and then on the reduce machine merges the incoming data from map nodes. This distributed sorting phase is often the key to the efficiency and scalability of MapReduce algorithms.

Figure 4.1 shows the canonical example MapReduce application. The task is to count the number of occurrences of every word that appears in a large text corpus. In this simple implementation, mappers take as input blocks of the text input. For every word w that they encounter, they emit the key/value pair $\langle w, 1 \rangle$. Reducers then sum all of the values



Figure 4.1: The word count example MapReduce application. Mappers examine blocks of text data and emit the value one under a key for each word that they encounter. Reducers sum the values they receive for each word, resulting in a count of the number of times each word occurs in the data set.

for each word key w, which gives the count of occurrences for that word in the input.

Hadoop is the Apache Software Foundation's open source implementation of the framework described by Google in Dean and Gehemewat's MapReduce paper. It includes an implementation of the MapReduce job scheduler, called the *TaskTracker*, as well as a distributed file system that mimics GFS called HDFS. The open source community has developed Hadoop extensively, with support from many corporations including Yahoo! and Facebook, making it a widely adopted and extended platform for distributed computing.

4.2 Uses of Hadoop and MapReduce in Sequencing Tasks

Now that we understand the components of Hadoop and MapReduce, we can explore its use in sequencing-related bioinformatic tasks. Figure 4.2 gives an overview of the published applications of Hadoop to three popular sequencing pipelines: DNA resequencing, RNAseq, and *de novo* assembly. In some cases, researchers have developed native MapReduce algorithms to solve particular problems. In other instances, Hadoop has been used to run existing tools in parallel; of course, this is only possible when the tasks are embarrassingly parallel at some level. Finally, there are several lower-level APIs, toolkits, and frameworks that have been created in an attempt to ease the development of Hadoop pipelines. These toolkits provide libraries for reading and writing popular sequencing related file formats, wrappers for commonly used tools, and functions to manipulate short read sequences and alignment records. We will discuss each of these pipelines in more detail in the following sections.

4.2.1 DNA Resequencing

As we discussed in Section 2.2.1, there are multiple steps that are typically part of DNA resequencing pipelines. These steps included quality control analysis, alignment to the genome reference, a series of processes including realignment in problematic regions and removal of duplicate reads, steps to call SNVs and SVs, and finally annotation and analysis of the variants that were discovered.

A native algorithm for mapping short reads to a reference genome was first demonstrated in Cloudburst [163]. This algorithm creates keys for each k-mer that appears in both the reads and the reference. The reducers then bring together the reads and portions of the reference that match on the same k-mer, and determine whether the match on the seed can be extended to an alignment of the entire read to the reference. Because the read and reference portions are each transferred to the reducers for every single k-mer which appears in them, this algorithm, while accurate, generates large amounts of data that must be shuffled, sorted, and potentially cross the network. Therefore, it is difficult to scale it to very large read sets. This shows the difficulty of designing MapReduce algorithms that can scale and take advantages of Hadoop's infrastructure. CloudAligner [137] uses a similar seed-and-extend algorithm, although the exact details of their map and reduce steps are not specified.

Since short read mapping is an embarrassingly parallel task and highly optimized non-MapReduce alignment tools with full feature sets are readily available, it is more efficient in practice to use the map phase of a Hadoop job to align splits of input reads with existing tools, and then merge results in the reduce phase. SEAL [151] and the CRS4 Sequencing and Genotyping Platform's CSGP workflow [150] are two such pipelines which use the

| | | | | anaina | | | | | |
|--|--------------------|-------------------|---|--|-------------------------------------|--|--|--|--|
| | | | DNA Resequ | encing | | | | | |
| | Quality Control | Alignment | Recalibration Deduplication Realignment | SNV Calling | SV Calling | Variant Annotation and Analysis | | | |
| SeqPig Crossbow SeqInCloud CSGP Workflow SEAL CloudBurst CloudAligner SeqWare BlueSNP | | | | | | | | | |
| DNA coo | | | | | | | | | |
| TIVA-SEQ | | | | | | | | | |
| | Quality Control | Alignment | Transcript Identification | Differential Expression Analysis | Pathway and Gene Set Analysis | SNV Calling | | | |
| SeqPig FX Myrna Eoulsan YunBe Kim et al. 2011 | | | | | | | | | |
| de novo Assembly | | | | | | | | | |
| | Quality Control | k-mer Counting | Error Correction | Assembly | | | | | |
| BioPig Quake Contrail CloudBrush | | | | | | | | | |
| Native Ma | pReduce Alg | orithm ——— | — Use o pre-ex | f Hadoop to parall kisting tools | elize | | | | |
| | A | APIs, Toolki | ts, and Script | ing Framewo | orks | | | | |
| BioPig FASTA I/O, BAM I/O, k-mer API, Wrappers for Assemblers SeqPig FASTA/Q I/O, BAM I/O, Alignment API Hadoop-BAM BAM I/O, Alignment API | | | | | | | | | |

Figure 4.2: Hadoop-enabled tools for tasks of three popular sequencing analysis pipelines. Solid lines indicate native MapReduce applications developed specifically for Hadoop. Dotted lines indicate pipelines that parallelize the execution of existing tools on splits of the input data using Hadoop. In addition, several toolkits and frameworks are listed that are broadly applicable to many portions of these pipelines. BWA [106] aligner in each map task. By emitting mappings under a key for the alignment location, they are then able to accomplish the duplicate removal step in the reduce function in a manner similar to that of the popular non-MapReduce tool Picard [24]. Similarly, Crossbow [95] uses Hadoop to parallelize Bowtie [96]. Crossbow then calls single nucleotide variants in the reduce phase using SOAPsnp [109], allowing it to function as a minimum viable pipeline for sequencing. Crossbow also includes infrastructure for running in the Amazon Elastic Compute Cloud, which has contributed to its popularity. SeqInCloud [131] similarly combines BWA with the GATK's [124] realignment and SNV calling steps, along with infrastructure to run in the Microsoft Azure cloud computing platform.

After variant discovery, the usual next step in a resequencing project is to analyze the variants found to detect functional impact (i.e. what genes might be disrupted by the variants found) or to find associations between variants and phenotypes, as in case/control sequencing projects. SeqWare [143] uses HBase [11], a Hadoop database based upon Google's BigTable [32], to provide functionality to annotate and query the variants. SeqWare uses the genomic coordinates of variants and coverage information as schema keys within HBase's MapReduce queries. BlueSNP [74], meanwhile, implements Hadoop algorithms for finding significant loci and estimating p-values through permutation analysis in genome wide association studies.

To our knowledge, there are no publicly available Hadoop/MapReduce software tools for SV detection. Several non-Hadoop pipelines such as HugeSeq [90] and SVMerge [194] use grid scheduling engines like SGE to distribute SV detection tasks across compute clusters. For example, the HugeSeq pipeline is an end-to-end pipeline for DNA resequencing that integrates BreakDancer [37], Pindel [200], CNVnator [2], and BreakSeq [89] for SV calling. However, these pipelines can only achieve parallelization by sample or at most by chromosome of the reference, limiting their scalability.

4.2.2 RNA-seq

The goals of RNA-seq are to determine the transcripts and isoforms being expressed, quantify their differential expression, and potentially call DNA variants based on the sequences of the RNA transcripts. Myrna [93] and FX [71] are Hadoop pipelines that execute alignment, transcript identification, and differential expression calculations in multiple Hadoop jobs. Myrna uses Bowtie in each mapper to distribute alignment work across the cluster, and then parallelizes each of the differential expression calculations by first assigning alignments to transcripts, gathering the data by sample to compute normalization factors, and then re-gathering the normalized counts under keys representing transcripts to compute statistics. FX has a less complex workflow, in which alignments are executed in a map-parallel fashion using the GSNAP alignment tool [195], and then has simple steps to count reads by gene for differential expression statistics, and to call SNPs in genes. Both Myrna and FX have infrastructure to automatically create Hadoop clusters in Amazon EC2. Eoulsan [78] is a very similar workflow, which allows a number of aligners to be used in the mapping step, and then computes read counts per gene in parallel, before computing differential expression statistics outside of the Hadoop environment using existing tools. For use after expression levels have been computed, YunBe [204] computes statistics to determine whether particular gene pathways have been perturbed in two sets of samples, using a MapReduce strategy where mappers produce expression values from the sample data under keys that represent specific pathways, and reducers aggregate all data for each pathway to determine a "perturbation statistic" between sample conditions. Finally, Kim et al. [81] described a pipeline for SNV detection from RNA-seq data based on a Hadoop job in which the mappers emit each base and quality score from the aligned reads under a key representing the genomic coordinate, and reducers call the absence or presence of a SNV at each location.

4.2.3 *de novo* Assembly

The third major pipeline for which there has been significant interest in applying Hadoop and MapReduce is *de novo* assembly. Most state-of-the-art algorithms for assembly depend on building large data structures that model the overlaps between reads (in the case of string graph based approaches) or between k-mers found in the reads (in the case of de Bruijn-graph algorithms) as edges in a graph. Given the high depth of coverage that is needed to produce a quality assembly, these graphs become very large, and the need to traverse them requires random rather than sequential access to the data. This would seem to make the problem a poor fit for Hadoop, which is optimized for sequential access of large data sets on commodity hardware. Nevertheless, several groups have attempted to create assembly algorithms using MapReduce. Contrail [165] builds a de Bruijn graph by emitting each pair of consecutive k-mers from the reads as a key-value pair to form a k-mer adjacency matrix. CloudBrush [33] uses a "prefix-and-extend" strategy [34], in which all k-mers that appear in the reads are used as keys, with the reads themselves as values. Extension procedures then test whether the k-mer is the prefix of one of the reads and the suffix of another. Both of these tools have then developed MapReduce procedures for graph pruning and traversal that are necessary to produce contigs from their respective graph data structures. Finally, some helpful read pre-processing steps for assembly have been implemented in Hadoop: Quake [79] is a widely used MapReduce k-mer counter, and BioPig [140] also contains facilities for k-mer analysis of large read sets.

4.2.4 Frameworks and Toolkits

In addition to the pipelines listed above, recently several sequencing-related frameworks and toolkits have been created to ease the development of Hadoop applications. These tools include APIs and library code to handle reading from and writing to files in popular sequencing data formats. For example, Hadoop-BAM [138] allows manipulation of data in the SAMtools [107] binary alignment format BAM. Two libraries, SeqPig [167] and BioPig [140], were also recently published that provide file formats and commonly used functions in the Apache Pig [12] environment, which provides a high-level programming language called Pig Latin that allows easy expression of simple data analysis tasks. These libraries allow for rapid development of scripts that gather statistics on read and alignment data sets, and Pig automatically optimizes their execution by translating them into one or more MapReduce jobs.

4.2.5 Other uses of Hadoop and MapReduce

Although not implemented in Hadoop, the very widely used Genome Analysis Toolkit [124] (GATK) implements many sequencing and variant calling functions using a MapReduce programming model. Although it currently can only distribute tasks across clusters using traditional grid engines, future versions of this package may be re-implemented to use Hadoop. Other notable sequencing applications that have been implemented in Hadoop include ChIP-seq peak calling [57], and computing genome mappability scores [100].

4.3 MapReduce Constraints on SV Algorithms

To this point, we know of no Hadoop-based SV detection algorithms. This may be because the need to separate logic into mappers and reducers makes it difficult to implement traditional RP-based SV detection approaches in MapReduce, particularly given the global clustering of paired end mappings at the heart of many RP approaches. MapReduce algorithms, by contrast, excel at conducting many independent calculations in parallel. The sequencing applications that have been implemented in MapReduce succeed by dividing processing into a series of local computations, for example calling a SNV at a particular location in the genome given the reads that cover it, to use the example of Crossbow [95], the most widely accepted sequencing MapReduce algorithm. As we noted in Chapter 3, SV approaches that are similarly based on local computations have been described: the RP-based SV callers MoDIL [102] and forestSV [126] try to solve the SV detection problem by computing scores or features along the genome and then producing SV predictions from those features in a post-processing step. In the remainder of this chapter, we formalize this idea and develop it as a potential framework for implementing SV detection algorithms in MapReduce.

4.4 A General MapReduce SV Detection Algorithm

Using the strategy of computing local features along the genome, we have developed a conceptual algorithmic framework for SV detection in MapReduce, which is outlined in Algorithm 1. This framework divides processing into three separate MapReduce jobs: an alignment job, a feature computation job, and an SV calling job. The overall workflow of the algorithm and its implementation on a compute cluster or cloud is summarized in Figure 4.3.



Figure 4.3: An overview of the steps of the MapReduce SV detection workflow. Reads are first uploaded to a Hadoop cluster from local storage. Processing is then then divided into three MapReduce jobs: 1) Mapping with sensitive settings. 2) Computation of features across the genome. 3) Calling structural variations based on the features computed in the previous step. Finally, SV predictions can be downloaded from the Hadoop cluster and examined and annotated. Cloudbreak can also use the Apache Whirr library to automatically provision Hadoop clusters on and deploy data to cloud providers such as Amazon Elastic Compute Cloud.

The ALIGN READS job uses existing alignment tools to discover mapping locations for each read pair. Aligners can be executed to report multiple possible mappings for each read, or only the best possible mapping. Given a set of read pairs, each of which consists of a read pair identifier rpid and two sets of sequence and quality scores $\langle s, q \rangle$, each mapper aligns each pair end set $\langle s, q \rangle$ in either single- or paired end mode and emits possible mapping locations under the rpid key. Reducers then collect the alignments for each paired end, making them available under one key for the next job.

In the COMPUTE FEATURES job, we compute a set of features for each location in

the genome. To begin, we tile the genome with small fixed-width, non-overlapping intervals. For the experiments reported in Chapter 6 we use an interval size of 25bp (see Section 6.2.2 for an exploration of the effects of different window sizes on accuracy and runtime). Let $L = \{l_1, l_2, \ldots, l_N\}$ be the set of intervals covering the entire genome. Let $R^1 = \{r_1^1, r_2^1, \ldots, r_M^1\}$ and $R^2 = \{r_1^2, r_2^2, \ldots, r_M^2\}$ be the input set of paired reads. Let $A^1 = \{a_{m,1}^1, a_{m,2}^1, \ldots, a_{m,K}^1\}$ and $A^2 = \{a_{m,1}^2, a_{m,2}^2, \ldots, a_{m,L}^2\}$ be the set of alignments for the left and right reads from read pair m. For any given pair of alignments of the two reads in a read pair, $a_{m,i}^1$ and $a_{m,j}^2$, let the ReadPairInfo $rpi_{m,i,j}$ be information about the pair relevant to detecting SVs, e.g. the fragment size implied by the alignments and the likelihood that the alignments are correct. We then leave two functions to be implemented depending on the application:

LOCI :
$$\langle a_{m,i}^1, a_{m,j}^2 \rangle \to L_m \subseteq L$$

 $\Phi : \{ \text{ReadPairInfo } rpi_{m,i,j} \} \to \mathbb{R}^N$

The first function, LOCI, maps an alignment pair to a set of genomic locations to which it is relevant for SV detection; for example, the set of locations overlapped by the internal insert implied by the read alignments. We optimize this step by assuming that if there exist concordant mappings for a read pair, defined as those where the two alignments are in the proper orientation and with an insert size within three standard deviations of the expected library insert size, one of them is likely to be correct and therefore we do not consider any discordant alignments of the pair. The second function, Φ , maps a set of ReadPairInfos relevant to a given location to a set of real-valued vectors of features useful for SV detection.

Finally, the third MapReduce job, CALL VARIANTS, is responsible for making SV calls based on the features computed at each genomic location. It calls another applicationspecific function:

POSTPROCESS : $\{\phi_1, \phi_2, \dots, \phi_N\} \rightarrow \{\langle \text{SVType } s, l_{start}, l_{end} \rangle\}$

This function maps the sets of features for related loci into a set of SV calls characterized by their type s (i.e Deletion, Insertion, etc.) and their breakpoint locations l_{start} and l_{end} .

```
Algorithm 1 The algorithmic framework for SV calling in MapReduce.
 1: job Alignment
        function MAP(ReadPairId rpid, ReadId r, ReadSequence s, ReadQuality q)
 2:
           for all Alignments a \in ALIGN(\langle s, q \rangle) do
 3:
 4:
                EMIT(ReadPairId rpid, Alignment a)
        function REDUCE(ReadPairId rpid, Alignments a_{1,2,...})
 5:
            AlignmentPairList ap \leftarrow VALIDALIGNMENTPAIRS(a_{1,2,...})
 6:
 7:
            EMIT(ReadPairId rp, AlignmentPairList ap)
   job Compute SV Features
 8:
 9:
        function MAP(ReadPairId rp, AlignmentPairList ap)
           for all AlignmentPairs \langle a_1, a_2 \rangle \in ap do
10:
               for all GenomicLocations l \in \text{LOCI}(a_1, a_2) do
11:
                   ReadPairInfo rpi \leftarrow < InsertSize(a_1, a_2), AlignmentScore(a_1, a_2) >
12:
                   EMIT(GenomicLocation l, ReadPairInfo rpi)
13:
        function REDUCE(GenomicLocation l, ReadPairInfos rpi_{1,2,...})
14:
           SVFeatures \phi_l \leftarrow \Phi(\text{InsertSizes } i_{1,2,\dots}, \text{AlignmentScores } q_{1,2,\dots})
15:
16:
            EMIT(GenomicLocation l, SVFeatures \phi_l)
17: job CALL SVS
        function MAP(GenomicLocation l, SVFeatures \phi_l)
18:
            EMIT(Chromosome(l), \langle l, \phi_l \rangle)
19:
        function REDUCE(Chromosome c, GenomicLocation l_{1,2,\ldots}, \phi_{1,2,\ldots})
20:
           StructuralVariationCalls svs_c \leftarrow \text{POSTPROCESS}(\phi_{1,2,\dots})
21:
```

We parallelize this job in MapReduce by making calls for each chromosome in parallel, which we achieve by associating a location and its set of features to its chromosome in the map phase, and then making SV calls for one chromosome in each reduce task.

4.5 Discussion

The framework that we have just described is agnostic to the type of structural variations that the user wishes to detect. In the next chapter, we describe Cloudbreak, our implementation of this framework that identifies small deletions and insertions. To do so, it defines the relevant information from each pair (the ReadPairInfo) as information about the insert size implied by the mapping of the paired reads, sends them to every location which is spanned by that read pair in the reference genome, and then computes features from them by modeling the expected distribution of insert sizes at each location with a Gaussian Mixture Model. We demonstrate some of the strengths of this particular implementation in Chapter 6. However, we believe that this general framework could be applied to many other SV detection problems. For example, to detect inversions, one might define a different ReadPairInfo for each pair that includes information about the orientation of the mappings. Translocation detection programs might emit ReadPairInfos that contain information about the chromosomes linked to by interchromosomal mappings, which the Φ function would then cluster to see if a preponderance of those mappings point to the same breakpoint location. In addition to paired mapping locations, it would also be possible to design ReadPairInfo and feature function definitions that allowed the computation of read depth or split read related features, enabling the integration of more of the signals available in the data set. We will explore one possible way to integrate disparate features such as these in Chapter 7. Use of the Hadoop/MapReduce computing framework would ensure that any of these applications, if carefully designed, could enjoy the benefits of redundancy, data locality, and scalability we described in Section 4.1, and therefore will be able to grow to meet the rising demands of genomics applications in the near future.

Chapter 5

Cloudbreak

In the previous chapter, we described and formalized a general strategy for building SV detection algorithms in MapReduce and Hadoop. In this chapter, we describe an soft-ware package, Cloudbreak, that we have developed using this algorithmic framework. To build Cloudbreak, we implemented the infrastructure necessary to support the algorithmic framework we described in Section 4.4, and also provided implementations of the three application-specific functions we described there. Here we will describe this implementation, as well as our choices for the three user-defined functions we specified in the framework definition, and additional functionality we developed to genotype calls and to facilitate the deployment of Cloudbreak on public cloud platforms including Amazon's EC2. In Chapter 6, we will provide an evaluation of the algorithm's accuracy and performance characteristics.

5.1 Variant types detected

Cloudbreak is our implementation of a detection algorithm for genomic deletions (40bp-25,000bp) and small insertions based on examining the insert sizes of paired end mappings. We chose this application because small deletions in the range of 50bp to 150bp are particularly difficult to detect using many existing SV algorithms [4, 127]. This is because most read-pair based algorithms use a hard cutoff based on the variance of the fragment size distribution to select discordant read pairs, as described in Section 3.2. By taking advantage of many compute cores using MapReduce, we can design an algorithm that considers all of available data (both concordant and discordant read pairs) in a generative statistical framework, as we will describe in Section 5.3.

5.2 Framework infrastructure

In addition to the specific algorithm for detecting deletions and insertions (which take the form of implementations of the user-defined functions we described in the previous chapter), the Cloudbreak package also contains the infrastructure necessary to implement the three MapReduce jobs defined in our MapReduce algorithmic framework. Providing a fully featured, multi-job Hadoop application requires several implementation decisions:

- Programming language and method of interacting with Hadoop. Hadoop applications can be developed in several ways. A native application is written in Java and directly uses the Hadoop application programming interface (API) to start jobs, implement Map and Reduce functions, and set advanced Hadoop configurations. Alternatively, applications can be developed in any language using the Hadoop streaming interface, as long as the input to and output from all map and reduce tasks is textual and certain conventions are followed with respect to data format. Finally, for C/C++ applications, the Hadoop *pipes* interface can be used to marshal input and output data from tasks. Each method has its own advantages and disadvantages. Native applications constrain the developer to Java but enjoy the best performance when the jobs are data intensive rather than CPU bound [47]. Streaming applications allow greater programming language flexibility but make it somewhat more difficult to organize complex applications and take advantage of advanced Hadoop features. Pipes, meanwhile, allow for maximum performance for CPU intensive applications. We opted to develop Cloudbreak as a native application to take advantage of the tight integration with the Hadoop API improved I/O performance on data intensive portions of the workflow. For the ALIGN READS job, which executes existing shortread alignment tools, a mapper class written in Java invokes the external tools using the system runtime environment.
- File formats and compression. Hadoop applications usually store their data in HDFS in text format or in sequence files, a binary format that allows numeric

or complex data types to be stored in a key-value pair structure easily accessed by Hadoop. In addition, varying levels of compression can be used, although only certain compression types allow Hadoop to automatically split large files by HDFS blocks for processing by different map tasks, which is a key consideration for building fully parallelized applications. Cloudbreak uses sequence files for input and intermediate files, although for alignments the values are stored as text strings containing full records in the Sequence Alignment Map (SAM) format [107], to allow for easy exports of data. For compression we use the Snappy [63] compressor/decompressor (codec), a compression scheme developed at Google which aims for reasonable file size reduction with very fast compression and decompression speeds. This makes it ideal for data-intensive Hadoop applications. Given that the output data from the ALIGN READS job is alignment records, a future implementation goal is to switch to using Hadoop-BAM [138], a library for storing SAM/BAM records efficiently in HDFS; Hadoop-BAM did not posses necessary functionality at the time of Cloudbreak's initial implementation and so we proceeded with a text-based representation of alignment records.

• Distribution of auxiliary files. In some cases all tasks require access to large input files, such as genome reference indices for alignment, or genome annotation files. Hadoop offers a *distributed cache* service, which places copies of the files on each node that will host tasks for the job so that they will not all need to copy the files over the network. Cloudbreak makes use of the distributed cache to distribute index and annotation files, as well as the alignment executables for the ALIGN READS job.

Our implementation of the ALIGN READS job contains wrappers to execute the aligners BWA [106], GEM [117], Novoalign [142], RazerS 3 [190], mrFAST [5], and Bowtie 2 [94]. This job can also be skipped in favor of importing a pre-aligned BAM file directly into HDFS. The code is structured in such a way that to add a new aligner, developers would simply create a class that finds the necessary index files in the distributed cache and determines the proper command line parameters for aligner execution, and parses the aligner output if it is in a non-standard format.

Cloudbreak can be executed on any Hadoop cluster; Hadoop abstracts away the details of cluster configuration, making distributed applications portable. We deployed Cloudbreak on an internal 56-node cluster running the Cloudera CDH3 Hadoop distribution, version 0.20.2-cdh3u4. In addition, Cloudbreak can create and operate on Hadoop clusters located in commercial and private compute clouds (see Section 5.6).

The source code and user manual for Cloudbreak are publicly available at https: //github.com/cwhelan/cloudbreak. We hope that by publishing under an open-source license, we will facilitate the adoption of our Cloudbreak implementation, as well as provide a base from which other computational researchers can develop their own SV detection algorithms for Hadoop.

5.3 Implementation of a MapReduce SV Algorithm

In Section 4.4, we described three user-defined functions that can be implemented to create an SV detection application in our MapReduce framework. These functions were named LOCI, Φ , and POSTPROCESS. These functions map aligned read pairs to locations on the genome to which they are relevant, compute a set of local features for each genomic location based on the relevant read pairs, and call variants based on the features computed for neighboring genomic locations. Cloudbreak contains implementations of these functions that combine to allow it to detect deletions (of size 40bp-25,000bp) and insertions. A detailed description of each of these implementations appears below, and an illustration of each phase of the Cloudbreak algorithm working on a simple example in MapReduce is shown in Figure 5.1.

LOCI Because we are detecting deletions and short insertions, we map ReadPairInfos from each possible alignment to the genomic locations overlapped by the implied internal insert between the reads. For efficiency, we define a maximum detectable deletion size of 25,000bp, and therefore alignment pairs in which the ends are more than 25kb apart, or in the incorrect orientation, map to no genomic locations. In addition, if there are multiple possible mappings for each read in the input set, we optimize this



Figure 5.1: An Example of the Cloudbreak deletion and insertion detection algorithm running in MapReduce. A) In the first MapReduce job, mappers scan input reads in FASTQ format and execute an alignment program in either paired-end or single-ended mode to generate read mappings. Reducers gather all alignments for both reads in each pair. B) In the second MapReduce job, mappers first emit information about each read pair (in this case the insert size and quality) under keys indicating the genomic location spanned by that pair. Only one genomic location is diagrammed here for simplicity. Reducers then compute features for each location on the genome by fitting a GMM to the distribution of spanning insert sizes. C) Mappers group all emitted features by their chromosome, and reducers find contiguous blocks of features that indicate the presence of a variant.

step by assuming that if there exists a concordant mapping for a read pair, defined as a mapping pair in which the two alignments are in the proper orientation and with an insert size within three standard deviations of the expected library insert size, it is likely to be correct and therefore we do not consider any discordant alignments of the pair.

 Φ To compute features for each genomic location, we follow the mixture of distributions

approach (see Section 3.2.2) first described by Lee et al. [102], who observed that if all mappings are correct, the insert sizes implied by mappings which span a given genomic location should follow a Gaussian mixture model (GMM) whose parameters depend on whether a deletion or insertion is present at that locus. Figure 5.2 shows several examples of the mixtures observed for various types of variants. If there is no indel, the insert sizes implied by spanning alignment pairs should follow the distribution of actual fragment sizes in the sample, which is typically modeled as normally distributed with mean μ and standard deviation σ . If there is a homozygous deletion or insertion of length l at the location, μ should be shifted to $\mu + l$, while σ will remain constant. Finally, in the case of a heterozygous event, the distribution of insert sizes will follow a mixture of two normal distributions, one with mean μ , and the other with mean $\mu + l$, both with an unchanged standard deviation of σ , and mixing parameter α that describes the relative weights of the two components. The features generated for each location l include the log-likelihood ratio of the filtered observed data points under the fit GMM to their likelihood under the distribution $N(\mu, \sigma)$, the final value of the mixing parameter α , and μ' , the estimated mean of the second GMM component.

The choice of a mixture of distributions model has several benefits. Firstly, it is a generative model for the entire data set, including concordant and discordant read pairs. This removes the need to set hard thresholds that define discordant read pairs, and allows the detection of smaller variants given a tight enough insert size distribution. Second, the parameters that are estimated can be used to refine and classify predictions. For example, the mixing parameter α can be used to genotype variants, as we will describe in Section 5.5. In addition, the estimated μ' parameter gives a prediction for how many genomic locations the variant might cover, which we leverage in the POSTPROCESS function described below to integrate local features into variant calls.

To implement our model, at each genomic location we fit the parameters of the GMM using the Expectation-Maximization algorithm. Let $Y = y_{1,2,\dots m}$ be the observed

insert sizes at each location after filtering, and say the library has mean fragment size μ with standard deviation σ . Because the mean and standard deviation of the fragment sizes are selected by the experimenter and therefore known a priori (or at least easily estimated based on a sample of alignments), we only need to estimate the mean of the second component at each locus, and the mixing parameter α . Therefore, we initialize the two components to have means μ and \bar{Y} , set the standard deviation of both components to σ , and set $\alpha = .5$. In the E step, we compute for each y_i and GMM component j the value $\gamma_{i,j}$, which is the normalized likelihood that y_i was drawn from component j. We also compute $n_j = \sum_i \gamma_{i,j}$, the relative contributions of the data points to each of the two distributions. In the M step, we update α to be $n_2 - |Y|$, and set the mean of the second component to be $\frac{\sum_m \gamma_{m,2} y_m}{n_2}$. We treat the variance as fixed and do not update it, since under our assumptions the standard deviation of each component should always be σ . We repeat the E and M steps until convergence, or until a maximum number of steps has been taken. Prior to fitting the GMM at each location, we attempt to filter out incorrect mappings for that location using an outlier-detection based clustering scheme and an adaptive mapping quality cutoff; see Section 5.4 for details.

POSTPROCESS The third MapReduce job is responsible for making SV calls based on the features defined above. We convert our features along each chromosome to insertion and deletion calls by first extracting contiguous genomic loci where the log-likelihood ratio of the two models is greater than a given threshold. To eliminate noise we apply a median filter with window size 5. We end regions when μ' changes by more than 60bp (2σ) , and discard regions where the average value of μ' is less than μ or where the length of the region differs from μ' by more than μ .

5.4 Filtering Incorrect and Ambiguous Mappings

One of our goals in developing Cloudbreak was to see if the use of distributed computing in the MapReduce framework would allow us to develop an algorithm that could process multiple mappings for ambiguously mapped reads in a reasonable amount of time, especially



Figure 5.2: Illustration of insert size mixtures at individual genomic locations. A) there is no variant present at the location indicated by the vertical line (left), so the mix of insert sizes (right) follows the expected distribution of the library centered at 200bp, with a small amount of noise coming from low-quality mappings. B) a homozygous deletion of 50bp at the location has shifted the distribution of observed insert sizes. C) A heterozygous deletion at the location causes a mixture of normal and long insert sizes to be detected. D) A heterozygous small insertion shifts a portion of the mixture to have lower insert sizes.

because the size of data sets can grow very large if many possible mappings are kept for all ambiguous reads. Most SV detection tools that use multiple mappings attempt to identify the correct mapping for each ambiguously mapped pair; for example, GASVPro [174] uses an MCMC approach to sample from the distribution of possible mapping locations for each read, and VariationHunter [72] attempts to assign mappings to reads through a combinatorial optimization approach. Cloudbreak, in contrast, attempts to solve this problem at the genomic location level by filtering mappings during the feature computation step.

To handle incorrect and ambiguous mappings, we assume that in general they will not form normally distributed clusters in the same way that correct mappings will, and therefore use an outlier detection technique to filter the observed insert sizes for each location. We sort the observed insert sizes and define as an outlier an observation whose kth nearest neighbor is more than $n\sigma$ distant, where k = 3 and n = 5. In addition, we rank all observations by the estimated probability that the mapping is correct and use an *adaptive quality cutoff* to filter observations: we discard all observations where the estimated probability the mapping is correct is less than the score of the maximum quality observation minus a constant c. This allows the use of more uncertain mappings in repetitive regions of the genome while restricting the use of low-quality mappings in unique regions. Defining MISMATCHES(a) to be the number of mismatches between a read and the reference genome in the alignment a, we approximate the probability p_c^k of each end alignment being correct by:

$$p_c^k(a_{m,i}^k) = \frac{\exp(-\text{MISMATCHES}(a_{m,i}^k)/2)}{\sum_j \exp(-\text{MISMATCHES}(a_{m,j}^k)/2)}$$

By multiplying $p_c(a_{m,i}^1)$ and $p_c(a_{m,i}^2)$, we can approximate the likelihood that the pair is mapped correctly.

5.5 Genotyping

In theory, it should be possible to use the parameters of the fit GMM to infer the genotype of each predicted variant. Assuming that our pipeline is capturing all relevant read mappings near the locus of the variant, the genotype should be indicated by the estimated parameter α , the mixing parameter that controls the weight of the two components in the GMM. We set a simple cutoff on the average value of α for each prediction to call the predicted variant homozygous or heterozygous, and use the same cutoff for deletion and insertion predictions. Somewhat surprisingly, we observed on the cutoff point that distinguishes homozygous from heterozygous variants is significantly less than the expected .5; based on empirical observations on a simulated data set, we set the threshold to .35 (see Section 6.5 for more details).

5.6 Running in the Cloud

In Section 2.2.2, we discussed the increasing recognition of the genomics community of the need for tools that ease the use of cloud computing resources [164, 178]. This has spurred the development of a variety of applications and toolkits that use the APIs of Infrastructure as a Service (IaaS) providers, such as Amazon EC2, to allocate compute resources on public compute clouds. In this section, we will briefly review some of the existing cloud-enabled tools, including those that are based on traditional grid scheduling engines and those that enable Hadoop and MapReduce pipelines. Many of the latter take advantage of Amazon's Elastic MapReduce (EMR) service, an offering from Amazon specifically for creating and running Hadoop jobs that simplifies the creation and monitoring of clusters on EC2. We will then discuss how Cloudbreak enables the use of cloud computing in a provider-neutral way using the Apache Whirr library.

5.6.1 Cloud-Enabled Genomics Tools

Several groups have created general-purpose toolkits for creating, allocating, and managing cloud resources for biological data processing. For example, the CloudBioLinux project [86] provides virtual machine images for use in Amazon EC2 that are pre-configured with a wide range of open-source bioinformatics applications. Galaxy CloudMan [3] allows for the automatic creation of clusters in the Amazon cloud configured to run workflows in the popular Galaxy environment [61], backed by the SGE grid scheduling engine and with interfaces to Amazon's S3 storage service. CloVR [9] is a cloud cluster manager that includes several pipelines for metagenomics managed by SGE. Finally, Elastream [76] is a newer offering that can provision and manage cloud clusters using either grid schedulers or EMR.

Several commercial providers are also offering services that allow elastic cloud computing for sequencing pipelines, including DNAnexus (Mountain View, CA), Illumina's (San Diego, CA) BaseSpace cloud, Seven Bridges Genomics' (Cambridge, MA) IGOR platform, and an offering from Biodatomics (Bethesda, MD). These commercial cloud services are in many cases backed by Amazon's EC2, but are wrapped in additional sequencing-specific APIs and user interfaces by the vendor. To our knowledge Biodatomics is the only commercial bioinformatics vendor that offers automatic Hadoop cluster provisioning, however.

There are also a number of bioinformatics suites designed for specific applications that are able to allocate resources on EC2 automatically. Apart from those that use Hadoop, which we will discuss in the next paragraph, these include: Atlas2 Cloud [56], which allows users to run the Genboree Workbench workflow for variant calling and annotation in personal genomics DNA resequencing projects; SIMPLEX [59], which enables cloud processing for an exome alignment and variant calling pipeline based on BWA and the GATK; a set of ChIP-seq tools of the modENCODE and ENCODE projects [183] that can create clusters to run analysis pipelines on EC2 or the Bionumbus private cloud [145].

Finally, several of the Hadoop applications we discussed in Section 4.2 come with command-line or graphical interfaces that automate their deployment on commercial cloud services. Most of these leverage EMR. The most widely used EMR-enabled tool is Crossbow [95], which has a UI that will create Hadoop workflows using EMR. Crossbow's ability to use EMR was recently made more robust by Rainbow [205], which wraps Crossbow with additional facilities for transferring large files in parallel, monitoring clusters for failing nodes, and aggregating results from multiple samples run simultaneously. As mentioned in Section 4.2, FX [71] and Euolsan [78] are both Hadoop workflows for RNA-seq processing; both have the ability to automatically create EMR jobs on Amazon EC2.

5.6.2 Enabling Cloud Computing with Whirr

In contrast to the tools mentioned above, Cloudbreak leverages the Apache Whirr [13] library to automatically create Hadoop clusters in the cloud. Whirr differs from the strategies referred to in the last section in that it provides a unified application programming interface (API) for provisioning and running cloud services that is agnostic to the cloud IaaS provider. Although the largest and most popular cloud IaaS is provided by Amazon Web Services though their Elastic Compute Cloud (EC2), Whirr is a facade API that allows the transparent substitution of other cloud services such as Rackspace, Microsoft Azure, or private clouds such as those built with Eucalyptus. Most of the Hadoop-enabled applications and toolkits mentioned in the previous section depend on Amazon Elastic MapReduce to provision Hadoop clusters. Cloudbreak's use of Whirr breaks this dependency on a single vendor.

Using Whirr's API, Cloudbreak is able to provision Hadoop clusters which can then be terminated when processing is complete. This eliminates the need to invest in a standing cluster and allows a model in which users can scale their computational infrastructure as their need for it varies over time. Through a command line interface, Cloudbreak uses Whirr functionality to offer commands that:

- Automatically provision Hadoop clusters in the cloud. After specifying the parameters of the Hadoop cluster desired in a property file, a single Cloudbreak command will request the creation of the necessary compute instances in the cloud, will configure them with the proper Hadoop services to provide a fully functioning cluster, and will start proxies that make the Hadoop cluster's management and reporting UIs available to the user. Cloudbreak users can configure clusters with their credentials for the cloud service provider, the number of worker nodes to include in the cluster as well the hardware requirements for each node, and, if desired, the pricing model. This last point is particularly useful on EC2, where spot pricing allows users to bid for unclaimed compute capacity on Amazon's cloud. Spot pricing can dramatically reduce costs compared to full price on-demand instances. The disadvantage is that instances can be reallocated if a higher bid comes in, resulting in termination of the processes they are running. Hadoop's facility for automatic redundancy of data and tasks can mitigate this risk.
- Transfer data into cloud compute clusters. Typically when using cloud compute services it is fastest to store large data sets in a cloud storage service such as Amazon's Simple Storage Service (S3). From there it is fast to transfer them into and out of cloud compute services like EC2. However, moving data into S3 can be time-consuming depending on the network connections and number and size of the files in the data set. Cloudbreak includes code which communicates with S3 to do a multi-threaded upload of large data files, enabling much faster transfer times.
- Destroy clusters when processing is complete. Cloudbreak, by using the Whirr

API, can destroy allocated cloud clusters when processing is complete, ensuring that compute costs can be managed efficiently.

Cloudbreak's user manual contains detailed instructions and examples describing how to leverage cloud computing. We hope that by making cloud computing readily accessible through Cloudbreak's command line interface, more researchers will have the opportunity to leverage Hadoop's distributed computing model, even if they do not have local Hadoop clusters available at their institutions.

5.7 Discussion

The strategy of fitting a GMM to the distribution of insert sizes at a given genomic location has been used by the SV detection tools MoDIL [102] and SVM² [39]. SVM² fits a mixture of distributions only to candidate regions of the genome identified through a preliminary analysis for the purpose of genotyping variants as homozygous or heterozygous. This leaves MoDIL as the only tool that attempts to model the distribution of insert sizes across the entire genome. As we will see in the next chapter, MoDIL is prone to excessively long run times that make it impractical to run for large-scale genomics data sets. Cloudbreak, on the other hand, through its use of MapReduce and Hadoop, is able to efficiently distribute computation so that given a sufficiently large cluster it can deliver the benefits of this strategy with very fast runtimes.

For those researchers that wish to take advantage of cloud computing in order to avoid the expense and maintenance costs of running their own large Hadoop clusters, Cloudbreak is able to automatically provision, transfer data to, and destroy Hadoop clusters using IaaS providers. Although there are other cloud-enabled sequencing analysis tools, Cloudbreak is unique in using the Apache Whirr library to provide an IaaS provider-agnostic solution. In addition to the obvious benefit of avoiding vendor lock-in, we believe that this will become increasingly important in the future as research agencies and clinical providers begin to create private or semi-private clouds to manage the analysis of sensitive personal genomic data in a controlled setting.

Chapter 6

Evaluating Cloudbreak

In this chapter we evaluate Cloudbreak and compare its accuracy, runtime, and additional features to a variety of other popular tools. We begin by defining some of the methods and parameters of our tests, and then describe experiments we have carried out using simulated data and two real data sets. Finally, we explore the runtime characteristics of the various tools and the extent to which they can be parallelized.

6.1 Evaluation Methods

6.1.1 Choice of SV Detection Tools to Compare To

We compared the performance of Cloudbreak for detecting deletions and insertions to a selection of popular tools: BreakDancer [37], GASVPro [174], Pindel [200], and DELLY [157]. BreakDancer and Pindel are two of the most highly cited SV detection tools, representing classic RP-based methods and SR-based methods, respectively. GASVPro is a hybrid RP method that integrates RD signals and ambiguous mappings into an RP framework based on discordant pairs. DELLY is a recent hybrid RP-SR method that uses split-read mapping to refine candidate calls made with RP information. DELLY produces two sets of calls, one based solely on RP signals, and the other based on RP calls that could be supported by SR evidence; we refer to these sets of calls as DELLY-RP and DELLY-SR. We also attempted to evaluate MoDIL on the same data given that it is the most algorithmically similar method to Cloudbreak. All of these methods detect deletions. Insertions can be detected by BreakDancer, Pindel, and MoDIL.

6.1.2 Simulated and Biological Data Sets

As has been observed elsewhere, there is no available test set of real Illumina sequencing data from a sample that has a complete annotation of SVs from the reference. Therefore, testing with simulated data is important to fully characterize an algorithm's performance characteristics. On the other hand, it is important that the simulated data contain realistic SVs that follow patterns of SVs observed in real data. To address this, we took one of the most complete lists of SVs from a single sample available, the list of homozygous insertions and deletions from the genome of J. Craig Venter [105]. Using these variants, we simulated a 30X read coverage data set for a diploid human Chromosome 2 with a mix of homozygous and heterozygous variants. Since there are relatively few heterozygous insertions and deletions annotated in the Venter genome, we used the set of homozygous indels contained in the HuRef data (HuRef.homozygous_indels.061109.gff) and randomly assigned each variant to be either homozygous or heterozygous. Based on this genotype, we applied each variant to one or both of two copies of the human GRCh36 chromosome 2 reference sequence. We then simulated paired Illumina reads from these modified references using dwqsim from the DNAA software package [70]. We simulated 100bp reads with a mean fragment size of 300bp and a standard deviation of 30bp, and generated 15X coverage for each modified sequence. Pooling the reads from both simulations gives 30X coverage for a diploid sample with a mix of homozygous and heterozygous insertions and deletions.

We downloaded a data set of reads taken from a DNA sample of Yoruban individual NA18507, experiment ERX009609 from the Sequence Read Archive. This sample was sequenced on the Illumina Genome Analyzer II platform with 100bp paired end reads and a mean fragment size (minus adapters) of 300bp, with a standard deviation of 15bp, to a depth of approximately 37X coverage.

To create a gold standard set of insertions and deletions to test against, we pooled annotated variants discovered by three previous studies on the same sample. These included data from the Human Genome Structural Variation Project reported by Kidd et al. [82], a survey of small indels conducted by Mills et al. [127], and insertions and deletions from the merged call set of the Phase 1 release of the 1000 Genomes Project [1] which were genotyped as present in NA18507. We merged any overlapping calls of the same type into the region spanned by their unions. It should be noted that the 1000 Genomes call set was partially produced using DELLY and BreakDancer, and therefore those calls are ones that those tools are sensitive to, biasing this test in their favor.

6.1.3 Parameters Used for Alignment and SV Detection

We aligned simulated reads to hg18 chromosome 2, NA18507 reads to the hg19 assembly. Alignments for all programs, unless otherwise noted, were found using BWA aln version 0.6.2-r126, with parameter -e 5 to allow for longer gaps in alignments due to the number of small indels near the ends of larger indels in the Venter data set. We also tested the effect of including multiple possible mapping locations for ambiguously mapped reads in results reported in Section 6.7. For those tests, we used two different sets of reads with multiple mapping locations reported. The first used alignments generated with BWA in paired-end mode, reporting up to 25 additional hits for each mapping using the -n and -N parameters for bwa sampe and the script xa2multi.pl. For the second, we attempted to generate an exhaustive list of possible mapping locations by running the GEM aligner in single-ended mode on each read in each pair individually, reporting up to 1000 additional hits per alignment. GEM was executed in parallel using Hadoop tasks which wrap GEM version 1.362 (beta), with parameters -e 6 -m 6 -s 2 -q ignore -d 1000 -max-big-indel-length 0. These parameters request all hits for a read that are within an edit distance of 6 of the reference, within 2 strata of the best hit, with a maximum of 1000 possible alignments reported for each read. GASVPro also accepts ambiguous mappings but expects them to be realigned with a more sensitive alignment tool; following the details given in their manuscript we extracted read pairs that did not align concordantly with BWA and re-aligned them with Novoalign V2.08.01, with parameters -a -r -Ex 1100 -t 250.

We ran BreakDancer version 1.1_2011_02_21 in single threaded mode by first executing bam2cfg.pl and then running breakdancer_max with the default parameter values. To run BreakDancer in parallel mode we first ran bam2cfg.pl and then launched parallel instances of breakdancer_max for each chromosome using the -o parameter. We ran DELLY version 0.0.9 with the -p parameter and default values for other parameters. For
the parallel run of DELLY we first split the original BAM file with BamTools [18], and then ran instances of DELLY in parallel for each BAM file. We ran GASVPro version 1.2 using the GASVPro.sh script and default parameters. Pindel 0.2.4t was executed with default parameters in single CPU mode, and executed in parallel mode for each chromosome using the -c option. We executed MoDIL with default parameters except for a MAX_DEL_SIZE of 25000, and processed it in parallel on our cluster with a step size of 121475. To execute other SV detection tools in parallel we wrote simple scripts to submit jobs to the cluster using the HTCondor scheduling engine [181] with directed acyclic graphs to describe dependencies between jobs.

6.1.4 SV Prediction Evaluation

Due to the inability of most tools to determine the exact breakpoint coordinates of an SV given read pair data, as well as the potential for uncertaintly due to error in real gold standard data sets, it is necessary to define a rule for determining whether or not a predicted SV call is correct or not. There are many ways of doing so, each with their own characteristics. For example, the authors of BreakDancer [37] and VariationHunter [72] considered a predicted deletion to be correct if had a 50% reciprocal overlap with a deletion in the test set. For GASVPro [174], Sindi et al. defined a "double uncertainty" metric, in which tolerance parameters ϵ and δ could be defined for the predictions and for the test set respectively, and a deletion prediction of the interval [x, y] is deemed to be correct if there exists an interval in the test set [a, b] for which there is overlap between both pairs of intervals $\langle [x - \epsilon, x + \epsilon], [a - \delta, a + \delta] \rangle$ and $\langle [y - \epsilon, y + \epsilon], [b - \delta, b + \delta] \rangle$. The forthcoming SMASH [180] benchmark of variant callers, including SV detection tools, counted a deletion call as a match if the left breakpoint of the call was within 100bp of the left breakpoint of the deletion interval in the test set and the difference between lengths of the two intervals was less than or equal to 100bp. The authors of the iSVP pipeline [129], meanwhile, assigned a quality score between 0 and 1 based on the ratio of the length of the intersection of the predicted and test variants, with 50bp margins added to each, to the length of the union of the two intervals.

To evaluate Cloudbreak we decided to use looser standards for calling a prediction

true, because we would like to test the maximum potential sensitivity of the algorithmic approach even if the resolution of the breakpoints is limited. We feel that this is appropriate given that most calls from SV detection tools will likely be validated either through an additional computational step such as local assembly, or through wet lab techniques such as PCR. Therefore, we use the following criteria to define a true prediction given a gold standard set of deletion and insertion variants to test against: A predicted deletion is counted as a true positive if a) it overlaps with a deletion from the gold standard set, b) the length of the predicted call is within 300bp (the library fragment size in both our real and simulated libraries) of the length of the true deletion, and c) the true deletion has not been already been discovered by another prediction from the same method. For evaluating insertions, each algorithm produces insertion predictions that define an interval in which the insertion is predicted to have occurred with start and end coordinates s and e as well as the predicted length of the insertion, l. The true insertions are defined in terms of their actual insertion coordinate i and their actual length l_a . Given this information, we modify the overlap criteria in a) to include overlaps of the intervals $\langle s, \max(e, s+l) \rangle$ and $\langle i, i+l_a \rangle$. In this study we are interested in detecting events larger than 40bp, because with longer reads, smaller events can be more easily discovered by examining gaps in individual reads. Both Pindel and MoDIL make many calls with a predicted event size of under 40bp, so we remove those calls from the output sets of those programs. Finally, we exclude from consideration calls from all approaches that match a true deletion of less than 40bp where the predicted variant length is less than or equal to 75bp in length.

6.2 Results on Simulated Data

6.2.1 Accuracy and Runtime

Figure 6.1 shows Receiver Operating Characteristics (ROC) curves of the performance of each algorithm for detecting deletions and insertions on the simulated data set. All approaches show excellent specificity at high thresholds in this simulation. Cloudbreak provides the greatest specificity for deletions at higher levels of sensitivity, followed by DELLY. For insertions, Cloudbreak clearly provides the best combinations of sensitivity



Figure 6.1: Cloudbreak accuracy on a simulated data set. Receiver Operating Characteristic (ROC) curves showing the specificity and sensitivity of each tool to deletions and insertions larger than 40bp on a simulated set of reads giving diploid coverage of 30X on human chromosome 2. Deletions and insertions from the Venter genome were randomly added to one or both haplotypes. Each point on a curve represents a different threshold on the confidence of predictions made by that tool. Thresholds vary by: Cloudbreak - likelihood ratio; BreakDancer, DELLY, GASVPro - number of supporting read pairs; Pindel - simple score.

and specificity. Figure 6.2 shows the runtimes for each tool on the simulated data set, parallelized when possible. Runtimes exclude alignment, which should be similar for all tools. Cloudbreak's runtime is half that of BreakDancer, the next fastest tool, processing the simulated data in under six minutes. Of course, Cloudbreak uses many more CPUs as a distributed algorithm. See Section 6.6 for a more detailed discussion of runtimes and the amount of parallelization that was done for each tool. The output which we obtained from MoDIL did not have a threshold that could be varied to correlate with the trade-off between precision and recall and therefore it is not included in ROC curves; in addition, MoDIL ran for 52,547 seconds using 250 CPUs in our cluster, so results are not included in the runtime figure. Apart from the alignment phase, which is embarrassingly parallel, the feature generation job is the most computationally intensive part of the Cloudbreak workflow. Therefore, to test the algorithm's scalability we measured the runtime of that job on Hadoop clusters made up of varying numbers of nodes and observed that linear speedups can be achieved in this portion of the algorithm by adding additional nodes to the cluster until a point of diminishing returns is reached (Figure 6.3).



Figure 6.2: Runtimes for each tool on the simulated data set, not including alignment time, parallelized when possible. See Section 6.6 for details on measuring and parallelizing the runtime of each tool.

Choosing the correct operating point or threshold to set on the output of an SV calling algorithm can be difficult when operating on a new data set. The use of simulated data and ROC curves allows for some investigation of the performance characteristics of algorithms at varying levels. First, we characterized the predictions made by each algorithm at the operating point which gives them maximum sensitivity. For Cloudbreak we chose an operating point at which marginal improvements in sensitivity became very low. The results for both deletion and insertion predictions are summarized in Table 6.1. MoDIL and Cloudbreak exhibited the greatest recall for deletions. Cloudbreak has high precision and recall for deletions at this threshold, and discovers many more small deletions. For insertions, Cloudbreak has the highest recall, although recall is low for all four approaches. Cloudbreak again identifies the most small variants. Pindel is the only tool which can consistently identify large insertions, as insertions larger than the library insert size do not produce mapping signatures detectable by read-pair mapping.

We also used the ROC curves to attempt to characterize the predictions made by each algorithm when a low false discovery rate is required. Table 6.2 shows the total number of simulated deletions found by each tool when choosing a threshold that gives an FDR closest to 10% based on the ROC curve. At this more stringent threshold, Cloudbreak identifies



Figure 6.3: Scalability of the Cloudbreak algorithm. Runtime of the Cloudbreak feature generation job for the simulated Chromosome 2 data is shown on Hadoop clusters consisting of varying numbers of compute nodes. Clusters were created in the Amazon Elastic Compute Cloud.

more deletions in every size category than any other tool. Performance on insertions never reached an FDR of 10% for any threshold, so insertion predictions are not included in this table.

6.2.2 Choice of Window Size

A key parameter choice for Cloudbreak is the size of the fixed-width, non-overlapping windows used for local feature computation. The experiments reported thus far in this chapter all used a window size of 25bp. Using the simulated data set, we evaluated the effect of choosing differing window sizes on runtime, breakpoint resolution, and accuracy. Figure 6.4

| | | Prec. | Recall | F1 | 40-100bp | $101-250 {\rm bp}$ | 251-500 bp | 501-1000 bp | $> 1000 \mathrm{bp}$ |
|------|--------------|-------|--------|--------|-------------------------|--------------------|-----------------|------------------------|-----------------------|
| | Total Number | | | | 224 | 84 | 82 | 31 | 26 |
| 0 | Cloudbreak | 0.638 | 0.678 | 0.657 | 153 (9) | 61(0) | 62(0) | 12(0) | 15(0) |
| on | BreakDancer | 0.356 | 0.49 | 0.412 | 89 (0) | 54(0) | 53(0) | 8(0) | 15(0) |
| leti | GASVPro | 0.146 | 0.432 | 0.218 | 83 (2) | 32(0) | 55(0) | 8(0) | 15(0) |
| De | DELLY-RP | 0.457 | 0.613 | 0.613 | 114(3) | 68 (0) | 66 (0) | 9(1) | 17(0) |
| | DELLY-SR | 0.679 | 0.166 | 0.266 | 0 (0) | 3(0) | 49(0) | 6(0) | 16(0) |
| | Pindel | 0.462 | 0.421 | 0.44 | 96 (11) | 24(0) | 48(0) | 5(0) | 15(0) |
| | MoDIL | 0.132 | 0.66 | 0.22 | 123(6) | 66 (3) | 66 (11) | 17 (7) | 23 (8) |
| s | Total Number | | | | 199 | 83 | 79 | 21 | 21 |
| on | Cloudbreak | 0.451 | 0.305 | 0.364 | 79 (32) | 32 (18) | 11 (8) | 1(0) | 0 (0) |
| erti | BreakDancer | 0.262 | 0.0968 | 0.141 | 23(5) | 14(5) | 2(1) | 0 (0) | 0 (0) |
| lnse | Pindel | 0.572 | 0.196 | 0.292 | 52(25) | 5(1) | 10 (9) | 3 (2) | 9 (9) |
| | MoDIL | 0.186 | 0.0521 | 0.0814 | 14 (1) | 4 (0) | 1(0) | 2 (2) | 0(0) |

Table 6.1: The number of simulated deletions and insertions in the 30X diploid chromosome 2 with Venter indels found by each tool at maximum sensitivity, as well as the number of those variants that were discovered exclusively by each tool (in parentheses). The total number of variants in each size class in the true set of deletions and insertions is shown in the first row of each section.

| | 40-100bp | 101-250bp | 251-500bp | 501 -1000bp | $> 1000 \mathrm{bp}$ |
|--------------|------------------|----------------|------------------------|------------------------|----------------------|
| Total Number | 224 | 84 | 82 | 31 | 26 |
| Cloudbreak | 68 (17) | 67 (10) | 56 (5) | 11 (3) | 15 (0) |
| BreakDancer | 52(8) | 49(2) | 49(0) | 7(0) | 14 (0) |
| GASVPro | 35(2) | 26~(0) | 26~(0) | 2(0) | 6 (0) |
| DELLY-RP | 22(1) | 56(1) | 40(0) | 8(0) | 12 (0) |
| DELLY-SR | 0 (0) | 2(0) | 28(0) | 2(0) | 10 (0) |
| Pindel | 60 (32) | 16(0) | 41(2) | 1(0) | 12 (0) |

Table 6.2: The number of simulated deletions in the 30X diploid chromosome 2 with Venter indels found by each tool at a 10% FDR, as well as the number of those deletions that were discovered exclusively by each tool (in parentheses). The total number of deletions in each size class in the true set of deletions is shown in the second row of the header.

shows the runtime of the feature computation and variant calling steps of Cloudbreak on the Chromosome 2 data set using differing window sizes. Runtime decreases dramatically with larger window sizes. For very small window sizes, the runtime is dominated by the variant extraction job, which is only parallelized by chromosome and therefore only runs on one core when running on the simulated data set. Table 6.3 shows several accuracy measures for Cloudbreak running with varying window sizes, using the same threshold for each run. The greatest recall is achieved at very low window sizes, probably because



Figure 6.4: Runtimes for Cloudbreak on the Chromosome 2 simulated data set using differing choices of window size. Runtimes include the feature computation and variant calling Cloudbreak Hadoop jobs.

evidence for smaller variants is less likely to be mixed in with non-variant supporting read pairs from the flanking regions. However, overall accuracy is high until window sizes reach 50bp, at which point recall decreases dramatically. Finally, we examined the effect of breakpoint accuracy on window size and found that it had little effect (Figure 6.5).

6.3 Results on Biological Data

6.3.1 Accuracy and Runtime

Figure 6.6 shows the performance of each algorithm on the NA18507 data set when compared against the gold standard set for both deletions and insertions. We were unable

| Window Size | Calls | True Positives | Precision | Recall | F1 |
|-------------|-------|----------------|-----------|--------|-------|
| 1 | 274 | 228 | 0.832 | 0.57 | 0.677 |
| 5 | 268 | 228 | 0.851 | 0.57 | 0.683 |
| 10 | 258 | 223 | 0.864 | 0.557 | 0.678 |
| 25 | 240 | 217 | 0.904 | 0.542 | 0.678 |
| 50 | 215 | 194 | 0.902 | 0.485 | 0.631 |
| 100 | 162 | 141 | 0.87 | 0.352 | 0.502 |

Table 6.3: Accuracy measures for Cloudbreak on the Chromosome 2 simulated data set using different choices of window size. For each window size the same Cloudbreak score threshold was used (1.98).

to run MoDIL on the whole-genome data set due to the estimated runtime and storage requirements. All other algorithms show far less specificity for the gold standard set than they did for the true variants in the single chromosome simulation, although it is difficult to tell how much of the difference is due to the added complexity of real data and a whole genome, and how much is due to missing variants in the gold standard set that are actually present in the sample. For deletions, Cloudbreak is the best performer at the most stringent thresholds, and has the highest or second highest precision at higher sensitivity levels. Cloudbreak has slightly lower accuracy for insertions than the other tools at more stringent thresholds, although it can identify the most variants at higher levels of sensitivity. Figure 6.7 and Table 6.6 show the runtime of each of the tools on the NA18507 dataset. Cloudbreak processes the sample in under 15 minutes on our cluster, more than six times as fast as the next fastest program, BreakDancer, even when BreakDancer is run in parallel for each chromosome on different nodes in the cluster.

Given the high number of false positives produced by all tools at maximum sensitivity indicated by the ROC curves, we decided to characterize the predictions made by each tool at more stringent thresholds. We examined the deletion predictions made by each algorithm using the same cutoffs that yielded a 10% FDR on the simulated chromosome 2 data set, adjusted proportionally for the difference in coverage from 30X to 37X. For insertions, again, we were forced to use the thresholds that gave maximum sensitivity for each tool due to the high observed FDR rates in the simulated data. The precision and recall at these thresholds with respect to the gold standard set, as well as the performance



Figure 6.5: Breakpoint resolution for Cloudbreak on the Chromosome 2 simulated data set using differing choices of window size.

of each algorithm at predicting variants of each size class at those thresholds, is shown in Table 6.4. For deletions, Cloudbreak has the greatest sensitivity of any tool at these thresholds, identifying the most variants in each size class. Pindel exhibits the highest precision with respect to the gold standard set. For insertions, Pindel again has the highest precision at maximum sensitivity, although according the ROC curve it is possible to choose a threshold for Cloudbreak with higher precision and recall. At maximum sensitivity, Cloudbreak identifies 120 more insertions from the gold standard set than Pindel, giving it by far the highest recall.



Figure 6.6: Accuracy on the 37X NA18507 sample. ROC curves for deletion and insertion prediction performance, tested against the combined gold standard sets of deletions taken from [82], [127], and [1].



Figure 6.7: Runtimes for each tool on the NA18507 data set, not including alignment time, parallelized when possible. See Section 6.6 for details on measuring and parallelizing the runtime of each tool.

6.3.2 Breakpoint Resolution

We expected that the methods tested here would vary in their breakpoint resolution: SR methods can achieve single nucleotide breakpoint resolution, while RP methods depend on insert size evidence from overlapping fragments that will likely not indicate the exact location of the breakpoint. This distinction can be seen clearly in Figure 6.8. Cloudbreak

| | | Prec. | Recall | F1 | 40-100bp | $101-250 {\rm bp}$ | 251-500 bp | 501-1000 bp | $> 1000 \mathrm{bp}$ |
|-----|--------------|--------|--------|--------|------------------|-------------------------|-----------------|----------------|----------------------|
| | Total Number | | | | 7,462 | 240 | 232 | 147 | 540 |
| ns | Cloudbreak | 0.0943 | 0.17 | 0.121 | 573 (277) | 176 (30) | 197 (18) | 121 (6) | 399(24) |
| tio | BreakDancer | 0.137 | 0.123 | 0.13 | 261(29) | 136(3) | 178(0) | 114(0) | 371(0) |
| ele | GASVPro | 0.147 | 0.0474 | 0.0717 | 120(21) | 40(2) | 85(0) | 36(0) | 128(0) |
| Д | DELLY-RP | 0.0931 | 0.1 | 0.0965 | 143(6) | 128(3) | 167(1) | 103(0) | 323(1) |
| | DELLY-SR | 0.153 | 0.0485 | 0.0736 | 0(0) | 26(0) | 123(0) | 66(0) | 203(0) |
| | Pindel | 0.179 | 0.0748 | 0.106 | 149(8) | 61(0) | 149(0) | 69(1) | 217(0) |
| ns | Total Number | | | | 536 | 114 | 45 | 1 | 0 |
| tio | Cloudbreak | 0.0323 | 0.455 | 0.0604 | 265 (104) | 49 (24) | 3(1) | 0 (0) | 0 (0) |
| ser | BreakDancer | 0.0281 | 0.181 | 0.0487 | 97(10) | 27(5) | 2(1) | 0 (0) | 0 (0) |
| In | Pindel | 0.0387 | 0.239 | 0.0666 | 144 (45) | 14(7) | 7 (6) | 1 (1) | 0 (0) |

Table 6.4: The precision and recall with respect to the gold standard set of deletions and insertions for each tool on the NA18507 data, as well as the number of variants found in each size class found. Exclusive predictions are in parentheses. For deletions, the same cutoffs were used as for the simulated data as in Table 6.2, adjusted for the difference in coverage from 30X to 37X. For insertions, the maximum sensitivity cutoff was used.

has the lowest resolution of any of the RP tools tested, with GASVPro also having relatively poor resolution. Pindel and DELLY-SR, meanwhile, use split read mappings to pinpoint the exact breakpoint correctly in many cases. Cloudbreak's resolution suffers because the independent calculation of features from the distribution of overlapping insert sizes does not allow the algorithm to keep track of the actual coordinates of the mapped reads from each pair, which other tools use to set bounds on the true breakpoint locations. Cloudbreak's goal, however, is to increase sensitivity and specificity to actual variants in the hopes that such calls could still be useful even if their resolution is limited. For example, it is possible to build SV detection pipelines in which a set of RP-based candidate calls is further validated *in silico* in a post-processing step, and such approaches typically also refine breakpoint resolution. In silico validation can be accomplished through a variety of methods; in one example, a recently published approach validated RP-based calls by performing a *de novo* assembly of the reads surrounding candidate calls that had been created by an RP-based approach [116]. Another approach to call refinement can be seen in studies that improve the breakpoint resolution of RP calls to single-nucleotide resolution using SR-based validation [92]. In Chapter 7, we will explore an alternative approach to in silico validation of SV calls based on a discriminative machine learning technique, and show that we can use it to greatly improve the breakpoint resolution of Cloudbreak's calls.



Figure 6.8: Breakpoint resolution for each tool for deletions from the gold standard set on the NA18507 data. For each correctly predicted deletion, we calculated the difference in length between the true deletion and the prediction.

6.4 Results on a Low-Coverage Cancer Data Set

We also tested Cloudbreak on the sequencing data set obtained from a patient with acute myeloid leukemia (AML) described in Section 3.7. This data set consisted of 76bp paired end reads with a mean insert size of 285bp and standard deviation of 50bp, yielding sequence coverage of 5X and physical coverage of 8X. Using a pipeline consisting of Novoalign, BreakDancer, and a set of custom scripts for filtering and annotating candidate SVs, we had previously identified a set of variants present in this sample and validated several using PCR, including 8 deletions. Cloudbreak was able to identify all 8 of the validated deletions, showing that it is still sensitive to variants even when using lower coverage data sets with a greater variance of insert sizes.

| | | Actual Genotypes | | | | | | |
|-----------|--------------|------------------|--------------|------------|--------------|--|--|--|
| | | Simulat | ed Data | NA18507 | | | | |
| | | Homozygous | Heterozygous | Homozygous | Heterozygous | | | |
| Predicted | Homozygous | 35 | 2 | 96 | 21 | | | |
| Genotypes | Heterozygous | 0 | 39 | 2 | 448 | | | |

Table 6.5: Confusion matrices for the predicted genotype of deletions found by Cloudbreak on both the simulated and NA18507 data sets.

6.5 Genotyping Variants

Because Cloudbreak explicitly models zygosity in its feature generation algorithm, it can predict the genotypes of identified variants. We tested this on both the simulated and NA18507 data sets. For the NA18507 data set, we considered the deletions from the 1000 Genomes Project, which had been genotyped using the population-scale SV detection algorithm Genome STRiP [65]. Cloudbreak was able to achieve 92.7% and 95.9% accuracy in predicting the genotype of the deletions it detected at our 10% FDR threshold in the simulated and real data sets, respectively. Table 6.5 shows confusion matrices for the two samples using this classifier. None of the three input sets that made up the gold standard for NA18507 contained a sufficient number of insertions that met our size threshold and also had genotyping information. Of the 123 insertions detected by Cloudbreak on the simulated data set, 43 were heterozygous. Cloudbreak correctly classified 78 of the 80 homozygous insertions and 31 of the 43 heterozygous insertions, for an overall accuracy of 88.6%.

6.6 Notes on Evaluating Runtime

We implemented and executed Cloudbreak on a 56-node Hadoop cluster, with 636 map slots and 477 reduce slots. Not including alignment time, we were able to process the Chromosome 2 simulated data in under six minutes, and the the NA18507 data set in under 15 minutes. For the simulated data set we used 100 reducers for the compute SV features job; for the real data set we used 300. The bulk of Cloudbreak's execution is spent in the feature generation step. Extracting deletion and insertion calls take under two minutes each for both the real and simulated data sets; the times are equal because each reducer is responsible for processing a single chromosome, and so the runtime is bounded by the length of time it takes to process the largest chromosome.

Cloudbreak's elapsed times are faster than all of the other tools tested; however, there are several ways in which to compare runtime performance between tools that support different levels of parallelization. In Table 6.6 we display a comparison of runtimes on the real and simulated data sets for all of the tools evaluated in this work. We report runtimes for each tools run in its default single-threaded mode, as well as for levels of parallelization achievable with basic scripting, noting that one of the key advantages of Hadoop/MapReduce is the ability to scale parallel execution to the size of the available compute cluster without any custom programming. Pindel allows multi-threaded operation on multicore servers. Pindel and BreakDancer allow processing of a single chromosome in one process, so it is possible to execute all chromosomes in parallel on a cluster that has a job scheduler and shared filesystem. BreakDancer has an additional preprocessing step (bam2cfg.pl) which runs in a single thread. DELLY suggests splitting the input BAM file by chromosome, after which a separate DELLY process can be executed on the data for each chromosome; splitting a large BAM file is a time consuming process and consumes most of the time in this parallel workflow, in fact making it faster to run in single-threaded mode. GASVPro allows parallelization of the MCMC component for resolving ambiguously mapped read pairs; however, this requires a significant amount of custom scripting, and we did not find that the MCMC module consumed most of the runtime in our experiments, so we do not attempt to parallelize this component. The MoDIL distribution contains a set of scripts that can be used to submit parallel jobs to the SGE scheduling engine or modified for other schedulers; we adapted these for use in our cluster.

In parallel execution, the total time to execute is bounded by the runtime of the longestrunning process. In the case of chromosome-parallelizable tools including BreakDancer, Pindel, and DELLY, this is typically the process working on the largest chromosome.¹ In the case of MoDIL's run on the simulated data, we found that the different processes

¹We note that one BreakDancer process, handling an unplaced contig in the hg19 reference genome, never completed in our runs and had to be killed manually; we exclude that process from our results.

| | | Simul | ated Data | | NA | A18507 | |
|-------------|----------|------------|------------|-------|-------------|------------|-------|
| | SV Types | Single CPU | Parallel | Proc. | Single CPU | Parallel | Proc. |
| Cloudbreak | D,I | NA | 290 | 312 | NA | 824 | 636 |
| BreakDancer | D,I,V,T | 653 | NA | NA | $134,\!170$ | $5,\!586$ | 84 |
| GASVPro | D,V | 3,339 | NA | NA | $52,\!385$ | NA | NA |
| DELLY | D | 1,964 | NA | NA | 30,311 | 20,224 | 84 |
| Pindel | D,I,V,P | 37,006 | 4,885 | 8 | 284,932 | $28,\!587$ | 84 |
| MoDIL | D,I | NA | $52,\!547$ | 250 | NA | NA | NA |

Table 6.6: Runtimes (elapsed) on both data sets of each tool tested, in single-processor and parallel mode. For parallel runs, Proc. is the maximum number of simultaneously running processes or threads. All times are in seconds. The types of variants detected by each program are listed with the abbreviations: D - deletion; I - insertion; V - Inversion; P - duplication; T - translocation. Interchromosomal translocations are only detected by BreakDancer in single CPU mode.

varied widely in their execution times, likely caused by regions of high coverage or with many ambiguously mapped reads. Cloudbreak mitigates this problem during the timeconsuming feature generation process by using Hadoop partitioners to randomly assign each genomic location to one of the set of reducers, ensuring that the work is evenly distributed across all processes. This distribution of processing across the entire cluster also serves to protect against server slowdowns and hardware failures - for example, we were still able to complete processing of the NA18507 data set during a run where one of the compute nodes was rebooted midway through the feature generation job.

6.7 Choice of Aligner and Use of Multiple Mappings

As mentioned in Section 5.4, one goal for developing Cloudbreak was to see if the use of Hadoop would allow productive use of large sets of possible mappings for ambiguously mapped read pairs. Therefore, we also tested the effect of using different aligners and including multiple mappings for ambiguously mapped reads. In addition to the BWA paired end best alignments used in the results reported above, we also used BWA in paired end mode set to report up to 25 possible mapping locations for each ambiguously mapped read pair, as well as the GEM aligner run in single ended mode on each read in each pair and set to report up to 1000 additional mappings per read. Interestingly, we



Figure 6.9: Cloudbreak performance on the chromosome 2 simulation using different alignment strategies. ROC curves show the number of true positives and false positives for each operating point for deletions and insertions. The Cloudbreak alignment strategies are: 1) "Cloudbreak": Alignments generated with BWA in paired-end mode, reporting the best hit for each pair. 2) "Cloudbreak-BWA-PE-MM": Alignments generated with BWA in paired-end mode, reporting up to 25 additional hits for each mapping. 3) "Cloudbreak-GEM-SE-MM": Alignments generated by running the GEM aligner in single-ended mode, reporting up to 1000 additional hits per alignment. Considering multiple mappings improves Cloudbreak's specificity for insertions but decreases sensitivity to deletions.

found that including multiple mappings increases accuracy on insertions, but decreases performance for deletions (Figure 6.9). This seems to indicate that while BWA's paired end mode is very good at picking correct alignments for pairs that map with long implied insert sizes, it seems to not be as good at finding correct pairings for the shorter distances indicative of insertions. We also examined the ability of all of the approaches, including Cloudbreak using both the BWA unambiguous best alignments and the GEM alignments, to detect events in repetitive regions of the genome, as shown in Table 6.7. We found that all of the tested methods detected a similar proportion of such variants, although we did find that when used with the GEM multiple mappings, Cloudbreak does exclusively find a set of both deletion and insertion variants. In terms of runtime, use of the large set of multiple mappings did not change elapsed time to run the simulated data sample greatly, taking 308 seconds versus the 290 sections reported in the previous section. For the NA18507 dataset, the difference in runtime was more substantial, with Cloudbreak moving from taking 824 seconds to process the best alignments to 2,310 seconds to process

| | | Simulated Data | | NA1 | 18507 |
|------|----------------|------------------|-------------------------|-----------------|--------------------------|
| | | Non-repeat | Repeat | Non-repeat | Repeat |
| | Total Number | 120 | 327 | 562 | 8059 |
| | Cloudbreak | 62(0) | 241 (3) | 300 (44) | $1166 \ (150)$ |
| TO. | Cloudbreak-GEM | 65(2) | 230(4) | 226(8) | 1001 (17) |
| ons | BreakDancer | 42(0) | 177~(0) | 192(11) | 868(20) |
| leti | GASVPro | 37(1) | 156(1) | 79~(6) | 330(16) |
| De | DELLY-RP | 57(1) | 217(2) | 152 (4) | 712(3) |
| | DELLY-SR | 3(0) | 71(0) | 27~(0) | 391~(0) |
| | Pindel | 26(4) | 162(7) | 109(2) | 536~(6) |
| | MoDIL | 72(12) | 223 (16) | NA | NA |
| | Total Number | 133 | 270 | 341 | 355 |
| ns | Cloudbreak | 32 (7) | 91 (30) | 169 (41) | 148 (43) |
| tio | Cloudbreak-GEM | 21 (0) | 58(7) | 137(30) | 135 (32) |
| sei | Breakdancer | 17(5) | 22(3) | 82(22) | 44(8) |
| In | Pindel | 25 (16) | 54(26) | 84(13) | 82(15) |
| | MoDIL | 5(0) | 16(2) | NA | NA |

Table 6.7: Detected deletions and insertions on the simulated and NA18507 data sets identified by each tool, broken down by whether the deletion overlaps with a RepeatMasker-annotated element. For all calls the maximum sensitivity cutoff was used.

the multiple mappings data set, a runtime which is still almost 50% less than the next fastest approach, BreakDancer parallelized by chromosome. Although the runtimes were very manageable, we see no substantial benefit in terms of accuracy in using large sets of ambiguous mappings, at least in our current implementation.

6.8 Discussion

We have demonstrated the Cloudbreak can deliver excellent accuracy at identifying regions that contain deletion and insertion SVs while at the same time achieving dramatically faster runtimes than other approaches through the use of the Hadoop framework and its implementation of parallel computing with data locality. In particular, the algorithm we developed for Cloudbreak is better able to identify small variants (50bp-150bp) than other tools. The downside to Cloudbreak's performance on our evaluations is its poor breakpoint resolution. As we mentioned previously, we believe that in most cases, predictions from SV detection tools require further validation, either through wet lab techniques or by further computational validation through techniques such as the split-read refinement offered by tools like DELLY, or through breakpoint assembly techniques such as TIGRA [36]. In the next chapter, we will demonstrate an additional technique that improves Cloudbreak's resolution: a discriminative machine learning framework that can identify breakpoint locations by integrating read pairing, depth of coverage, and split read signals from sequencing data sets.

Chapter 7

Extending Local Feature Based Models of SV Detection in a Discriminative Machine Learning Framework

In the formulation of a general algorithmic framework for SV prediction in MapReduce that we described in Chapter 4, a crucial step is the computation of a set of features for each of the small, non-overlapping windows with which we tiled the genome reference. In the implementation of Cloudbreak described in Chapter 5, these features were the parameters which are estimated by fitting a GMM to the distribution of insert sizes that span that location. However, in general the nature of these features are purposefully not specified in the algorithmic framework to allow flexibility in the nature of the applications that it could potentially be used to build.

Given a set of arbitrary features that encode information about a set of loci that are connected in a sequence, a natural approach is to apply techniques from machine learning to identify regions of interest in that sequence, rather than the heuristics and noise reduction techniques from signal processing that are used in the Cloudbreak implementation described in Chapter 5. In this chapter we will reformulate the problem as a sequence labeling task, discuss possible machine learning frameworks that could be used to solve it, explore the ways in which features can be engineered to integrate multiple signals of SVs, and show that it is possible to achieve modest but positive improvements using conditional random fields, a discriminative machine learning technique.

7.1 Related Work

Discriminative machine learning techniques have been applied to SV detection in the tools forestSV [126] and SVM² [39]. SVM² computes a set of statistics based on differences between the coverage and insert size distributions observed at nearby locations on the genome. If the statistics pass a set of coarse filtering rules, they are used as features for a support vector machine (SVM) classifier that the authors trained on a set of simulated short insertion and deletions. forestSV trained a Random Forest classifier [23] to detect deletions and duplications based on validated and false positive predictions from many samples in the 1000 Genomes Project data set. Their tool is based upon computing features for a sliding window moving across the genome, as well as for the flanking regions of that window. The authors demonstrated that it was possible to combine disparate signals by including features that were based on read depth as well as discordant read pair mappings in their feature set. The forestSV algorithm was similar to Cloudbreak in that the classifier made calls at each window, and a postprocessing function then merged windows classified with the same label into variant calls. Both of these methods demonstrate innovative ways of learning from existing data, and integrating read pair and split read based signals. However, neither of these tools use machine learning techniques that take into account the sequential nature of the data; the classifiers are run independently on each candidate variant location (in the case of SVM²) or window (in the case of forestSV), and the labels that are assigned in each invocation of the classifier do not affect the neighboring regions. We will show that by treating the problem as a sequence labeling task, it is possible to take advantage of learning techniques that operate on the entire sequence of observations at once.

7.2 SV Detection as a Sequence Labeling Problem

In our MapReduce algorithmic framework, the features computed at each window are transformed into variant calls by a function that examines the features along each chromosome of the reference sequentially and identifies contiguous blocks of regions with features values that combine to coherently indicate the presence of a variant. This part of the process, which we named the POSTPROCESS function in Algorithm 1, can be thought of as a sequence labeling problem based on a series of observation features. To take the example of deletion detection, we could define two labels: "Deletion" if the window participates in a deletion variant, and "No Deletion" if it does not. More formally, we can let Y_i be the label at genomic window i, where $1 \le i \le n$ and n is the number of windows in the chromosome. If we similarly name the features for genomic window $i X_i$, we can then refer to the entire sequence of labels and features as Y and X, respectively.

The goal of the POSTPROCESS function can be broken down into two parts: first, to assign a label to each window in the reference sequence, and second, to consolidate neighboring windows with the same label into variant calls which affect larger regions. Analyzing the implementation of Cloudbreak described in Chapter 5 in this model, we used a simple linear threshold on the likelihood ratio of the insert size data to label each window, and then consolidated neighboring windows with hand-tuned rules such as the median filter and restriction on the estimated mean of the second GMM component. These latter rules were made necessary by the noisy nature of the data and the simplicity of the window-labeling procedure. However, if we had a completely accurate window-labeling procedure, we could remove much of the complexity of the consolidation step. Since real world data is noisy the best that we can do is to try to find the most likely sequence of labels given the observed data:

$\underset{Y}{\arg\max} P(Y|X)$

7.3 Graphical Models for Sequence Labeling

To achieve the goal of a more principled and more accurate window-labeling procedure we would need to be able to take into account information from multiple features, as well as the labels of nearby windows. In other words, the label of each window should be dependent not only on the observed features at that window but also on the labels of other windows nearby. Probabilistic graphical models provide a useful framework for defining and learning models that describe these types of dependencies. One class of graphical



Figure 7.1: A Hidden Markov Model for the SV detection problem.

model that has been used extensively for sequence labeling tasks in bioinformatics are Hidden Markov Models (HMMs). HMMs model sequence labeling problems by assuming that the observations at each point are dependent on a hidden state, and that the state at a given time is dependent on the state at the previous time step. If one creates an HMM such that the hidden states correspond to the labels of interest, the task of learning an HMM is equivalent to learning a probability distribution $P(Y_i|Y_{i-1})$ that describes the likelihood of moving from one state to another, and a distribution $P(X_i|Y_i)$ that gives the likelihood of seeing a particular observation given the true label at that time point, which can be modeled by a directed graphical network (Figure 7.1). Therefore, HMMs are generative models that model the entire distribution over labels and observations, or the joint distribution P(Y, X). One published CNV detection algorithm, Zinfandel [171], used an HMM to predict the presence of deletions and duplications. That algorithm used a feature set consisting of the read depth at each location and likelihood of the distribution of insert sizes given varying fixed potential sizes of deletions.

One downside to modeling the joint distribution over features and labels is that what we are really interested is the conditional distribution P(Y|X): given the observations, what is the most likely sequence of labels? Since P(Y,X) = P(Y|X) * P(X), by training an HMM we are forced to learn P(X), or the distribution over sequences of observations, even though it does not help us achieve our goal. A second drawback to HMMs, particularly in the proposed application of SV detection, is the fact that the model assumes that $P(X_i)$ is independent of $P(X_{i-1})$ given Y_i . In other words, observations drawn from the same state should all be independent. For the SV detection problem described here, this seems highly restrictive: for many features we might imagine, observations from two different genomic windows that are participating in the same deletion variant are likely to be much more similar than two windows that are participating in different deletion variants, even though they all share the label "Deletion". As the developers of Zinfandel noted, this required them to create a grid of "Deletion" and "Deletion Flank" states, with each row corresponding to a different size class of deletion variant.

Conditional random fields [88] (CRFs) offer solutions to each of these drawbacks. In contrast to HMMs, CRFs are Markov Random Fields which can be represented by undirected graphical models. Because the model is undirected, it is possible to directly learn the conditional distribution P(Y|X) from a set of training data. Training and inference of these models is tractable for simplified graphical structures; for our application we will use a *linear chain CRF*, which mimics the structure of a sequence of observations and its labels (Figure 7.2). Furthermore, CRFs represent the factors that relate the observation sequence to the label node at position i in the sequence in the graph through a log-linear model over a set of k feature functions:

$$\exp\left(\sum_k \theta_k f_k(y_i, y_{i-1}, X)\right)$$

This means that the feature functions for each position i in the sequence can incorporate information from observations any point in the observation sequence. This removes the requirement that observations be independent from one another, and allows the consideration of features taken from neighboring observations, something that could be useful in our context of SV detection, where multiple small genomic windows are spanned by single reads and fragments.

7.4 Integrating Features with Conditional Random Fields

We implemented a linear-chain conditional random field model to label genomic insertions and deletions. The model was developed using Factorie [121]. Factorie is a toolkit written in Scala that supports a wide variety of factor-graph based models, trainers,



Figure 7.2: A linear chain Conditional Random Field model of the SV detection problem. Label nodes can be connected by feature functions to any of the observations in the sequence.

and tools, particularly for NLP tasks such as named entity recognition. Our Factoriebased implementation of a structural variation detection program consists of two parts: a modular and configurable set of code for data management of features defined along the genome, along with conversion of real-valued features into into binary binned feature values, and code to construct, train, and run inference on linear-chain CRF models. The code developed for this project was written in Scala and is available on GitHub at http://github.com/cwhelan/svfactorie.

7.5 Features for SV Detection

As described in Chapter 2, there are three main signals available for SV detection in short read sequencing data sets: those that come from read pairing information, those that come from read depth, and those that come from split reads. The creation of a discriminative machine learning framework as described above can be used with any arbitrary set of features. Therefore it is possible to create feature sets that combine information from all three SV detection signals and integrate them into this framework. In addition, we can model interactions between features and use those in our predictions. Finally, the arbitrary nature of the feature function allows us to incorporate prior knowledge about given genomic regions, including sequence annotations. We have constructed a feature set that includes all of these types of features, as described below:

7.5.1 Read Pair Features

Read pair features are those that are based upon the inferred insert sizes and orientations linking paired reads, as described previously. We use the following RP features:

- The three features generated by Cloudbreak and described in Chapter 5: the log likelihood ratio of the insert sizes observed at each window in the two-component GMM fit vs. the likelihood under the expected normal distribution for the sample; the estimated mean μ₂ of the second component of the two-component GMM, and α, the estimated weight of the second component in the two-component GMM.
- Insert size change point features: in addition to the insert size calculations described in the Cloudbreak implementation, we also wanted to consider alternative features based on insert sizes. In particular, we wanted to test whether the addition of features that indicate whether a window represents a point at which the insert size distribution in a local surrounding region is changing would improve identification of variants. We compute such a score as follows: let *i* be the index of a window in the genome, and $S_{i..j}$ be the list of observed insert sizes spanning the windows labeled *i* through *j*. Now consider a local neighborhood of size *n* which comprises the windows $i - \frac{n}{2}, ..., i, ..., i + \frac{n}{2}$. Let $\Theta_{i..j}$ be model parameters that can be estimated from the distribution of insert sizes $S_{i..j}$, and $P(S_{i..j}|\Theta_{i..j})$ be the likelihood of observing $S_{i..j}$ under the model with parameters Θ . We can calculate a change score for each window *i* by examining the likelihood of the data in the halves of the neighborhood to the right and left of that window, by estimating $P(S_{i-\frac{n}{2}..i})$ and $P(S_{i+1..i+\frac{n}{2}})$. Simultaneously, we can perform the same estimates for all of the data in the neighborhood and compute $P(S_{i-\frac{n}{2}..i+\frac{n}{2}})$. The likelihood ratio between these two models:

$$\frac{P(S_{i-\frac{n}{2}..i}) * P(S_{i+1..i+\frac{n}{2}})}{P(S_{i-\frac{n}{2}..i+\frac{n}{2}})}$$

then serves as a score indicating the presence of a change between the segments to the left and right of point i. This has the advantage of being a parameter-free calculation, since parameters are re-estimated from each segment for each calculation. According

to the Cloudbreak's model, it would be natural to estimate two component GMMs for each segment of insert size data. However, in practice we found that changes were well captured by the likelihood ratio score when only a single-component Gaussian is used for each segment, and so we have used that model for the sake of efficiency. We use a neighborhood size of 200 bp.

7.5.2 Split-read features

Split read signals are caused when a read overlaps a genome breakpoint, disrupting the full alignment of that read. If only simple variants are considered, such a scenario should reveal the exact location of the breakpoint. However, these are difficult to resolve in practice, because the algorithm must first identify potentially split reads from those that fail to align or partially align, and then align the two ends of the split read to the genome, potentially at large distances from one another. True split-read based SV detection methods use dynamic programming and heuristics to try to accomplish this; however, the difficulty of doing so with current read lengths means that they have low sensitivity (although very high specificity). Rather than conduct an exhaustive split read alignment search, we instead use indirect evidence of split reads that can be extracted from standard first-pass alignments:

- Soft clipping: When reads are aligned to the genome by short read mappers like BWA, they can in some cases be only partially aligned. In BWA, this occurs when a parameterized clipping penalty is less than the additional cost of aligning the rest of the read according to a Smith-Waterman dynamic programming alignment. Aligners refer to this "soft clipping", and flag the read with specific indicators when it occurs. Although there are other possible reasons for soft clipping (for example, the quality could fall at the end of the read, producing many erroneous mismatches with the reference), this could potentially be indicative of a structural variation breakpoint disrupting the read's alignment to the reference. For integration into the CRF feature set, we created a feature track that counts the number of times a soft clip occurred in each 25bp genomic window.
- Singleton alignments: For paired end reads, sometimes only one read in the pair

can be successfully aligned to the genome. These are referred to as "singleton" alignments. Potentially a singleton alignment could indicate that the other read in the pair contains a variant that prevented it from being aligned, and therefore that a breakpoint could lie somewhere within the distance of the insert size, or could have disrupted the alignment of the other read in the pair. We count the number of singleton alignments in each window.

7.5.3 Read depth features

The third major signals of structural variations from short-read sequencing data are related to read depth. In the context of deletions, one would expect to see fewer reads aligning to sections of the genome which have been deleted in the sample; if the deletion is 100% represented in the sample and is homozygous, any reads aligning to that location must be incorrect alignments. In practice, coverage depth is affected by noise resulting from incorrect mappings and biases of the sequencing process (for example, the amount of GC content in a given fragment of DNA will affect its representation in the sequencing data set).

- Coverage depth: As one feature, we calculate the average coverage depth of each base in each 25bp window.
- Coverage change points: We applied the change point detection algorithm outlined in the previous section to identify windows at which the coverage changes. Although we experimented with different neighborhood sizes, we found that this was not a very valuable feature for breakpoint identification. For the results reported below, we used a neighborhood size of 600bp.
- Coverage Drops: We speculated that for the specific case of deletion detection, the most informative coverage statistic might be one that indicates whether or not the coverage at a given window drops relative to its local neighborhood. Therefore, we compute for each window the drop in coverage depth from the mean coverage depth of the windows that lie in the surrounding 500bp.

7.5.4 Genome annotations

Finally, the ability to include arbitrary features in the model allows us to incorporate prior knowledge about the genome into our feature set. In particular, some types of genomic features frequently cause incorrect alignments; training the model based on those feature annotations could help the variant caller calibrate its confidence in each prediction.

- Repetitive regions: We took the UCSC RepeatMasker track for the reference genomes used in the testing and training sets and created a binary feature for each window in the genome, which was true if the window overlaps with a repetitive element designated by RepeatMasker [175].
- Simple repeats: these are a subset of the repetitive regions track that are made up of repeated k-mers of lengths from one to six. Simple repeats are very difficult to align to, and are therefore the source of many alignment errors. We set a binary feature for each window if it overlapped with an element from the UCSC Simple Repeats track [22] for the reference genome.
- Segmental duplications: These are larger regions (10kb+) that have at least one other copy in the genome with high sequence similarity between the two regions, and are again the source of many alignment errors. We used the UCSC Segmental Duplications track to set binary features on each window in the genome.

7.5.5 Binarization of real-valued features

Although conditional random fields can support real-valued features, the fact that they are linear models means that it is often helpful to convert real-valued features into related set of binary-valued features for labeling problems. For example, this can help prevent the domination of the output by single real-valued features with very high values. Therefore, we built two binarization schemes into our code base for real valued features. The first allows the user who is training the system to specify a set of cut points $c_{1..k}$ which divide the range of the variable into k+1 bins. If the value of the feature lies in bin *i*, the system sets binary feature b_i to true and all others to false. The second binarization procedure is

| Feature Name | Type | Column | Description | Cutpoints |
|---------------------------|---------------------------------------|--------|--------------------------|------------------------------|
| mu1 | binnedReal | 1 | μ_2 from Cloudbreak | 260.0,340.0 |
| | | | GMM | |
| lr | $\operatorname{cumulativeBinnedReal}$ | 2 | Likelihood Ratio from | 0.75, 2.5, 10.0, 75.0, 500.0 |
| | | | Cloudbreak GMM | |
| singletons | $\operatorname{cumulativeBinnedReal}$ | 6 | Singleton alignments | 1.0,2.0 |
| changePoint | $\operatorname{cumulativeBinnedReal}$ | 7 | Insert size change point | 5.0, 15.0, 50.0 |
| $\operatorname{softClip}$ | $\operatorname{cumulativeBinnedReal}$ | 8 | Soft clipped alignments | 1.0, 2.0, 3.0 |
| rdepth | $\operatorname{cumulativeBinnedReal}$ | 9 | Average read depth | 0.1, 0.25, 0.5, 0.75, 1.0 |
| covChangePoint | $\operatorname{cumulativeBinnedReal}$ | 10 | Depth of coverage | 10.0,20.0,30.0 |
| | | | change point scores | |
| covDrop20 | $\operatorname{cumulativeBinnedReal}$ | 11 | Coverage drop | 5.0, 10.0, 20.0 |
| simpleRepeat | boolean | 13 | Simple Repeat | |
| repeat | boolean | 14 | Repeat | |
| segdup | boolean | 15 | Segmental Duplication | |

Table 7.1: Feature definition for the CRF training and test data. Each line indicates the feature's name and type, as well as the type of binning scheme (simple or cumulative), and the cut points to use when binning.

a cumulative binning scheme. Again, the user specifies cut points which divide the range of the variable into *i* bins. In this case, however, binary features are set to true for bin whose end point is greater than or equal to the actual value of the variable. To specify all feature definitions and their types, the user defines each feature variable, its location in the training and test data files, the binning scheme, and the cutoff points for binning. Table 7.1 shows the feature definitions used for the results reported later in this chapter. This scheme allows for rapid testing of different feature combinations and binning schemes, depending on what data is available.

7.5.6 A feature example

Figure 7.3 shows an example deletion from the NA18507 gold standard data set, along with tracks showing each of the features described above, as well as the original Cloudbreak call identifying the deletion and a call made by running inference on the trained CRF model. One can observe an elevated likelihood ratio in the first track covering the deletion. At or near the breakpoints of the deletion there are higher scores for the insert size change point, soft clip count, and singleton alignment features. The deletion span is also associated with lower coverage values.

7.5.7 Interaction and neighbor features

Several of these features may be most informative when considered in conjunction with other features. For example, low coverage depth may mean something different when it occurs in a repetitive element, since that region of the genome might be harder to align reads to. Similarly, the features of neighboring windows could also inform the choice of label; since the CRF makes no demands of independence between neighboring feature functions, it is possible to include these neighboring features in the model. To that end, in addition to adding the features defined above to each bin, we also add as separately labeled features:

- Interaction terms for all of the features for the current window.
- The features of the windows immediately before and after the current window.
- The features of the windows within a certain radius of the current window (currently 7 neighboring bins).

Although this creates a large feature space, we hope that by training with regularization the optimizer will be able to select the most important features and avoid overfitting.

7.6 Training the CRF

As noted in Chapter 6, there are limited complete SV annotations of real data sets available. Therefore, we decided to train on simulated data. It should be noted that training on simulated data only exposes the model to one form of noise: incorrect or incomplete mappings. In real data sets, there are additional sources of noise in short read data sets, including chimeric fragments; complex structural variations, and errors in the genome reference. Therefore, any gains that can be made by training on simulated data and testing on real data are less than the possible gains that could be achieved with a more realistic training set.

To increase the number of training examples, we took the original simulation of Chromosome 2 described in Chapter 6 and expanded it to a whole genome simulation. This



Figure 7.3: An example deletion with features from the NA18507 data set. The true deletion is shown in track 8. Features shown are in each track are: 1) Likelihood ratio of distribution of insert sizes. 2) Insert size change points. 3) Number of soft clipping events in each window. 4) Number of singleton mappings in each window. 5) Read coverage in each window. 6) Coverage change points. 7) Coverage drops. 11) Simple repeats. 12) Repeats. The original Cloudbreak call is shown in Track 13, and a call made by running inference in the CRF model is shown in Track 14.

also has the effect of increasing the possibility for mapping ambiguity, adding noise to the signal. The simulation includes all of the insertions and deletions annotated for J. Craig Venter's genome. Considering variants with a length over 40bp, there are 5,610 deletions and 6068 insertions. We simulated 30X coverage 100bp paired-end reads with an insert size of 300bp and aligned them to the reference genome using BWA, as described in Section 6.1.2.

We created a set of labels based on a bitmask representation of the following four states that a window could be in with respect to deletions and insertions: "Deletion",



Figure 7.4: An example of the labeling scheme used for CRF model training and testing. Labels are a bitmask composed of variant and flanking information. First bit: Deletion. Second bit: Deletion Flank. Third bit: Insertion. Fourth bit: Insertion Flank.

"Deletion Flank", "Insertion", and "Insertion Flank". We decided to include separate labels for flanking regions because several of the features we selected would be most likely to occur in the immediate flanks of deletion or insertion variants, such as singleton mappings, or the change point detection features. The use of a bitmasked label also means that we capture the windows that actually contain the breakpoints for each deletion or insertion under a separate label, since those windows contain both flanking sequence and variant sequence. See Figure 7.4 for an example of how labels are assigned to windows. This labeling scheme translates to a set of six commonly used labels: "Outside Variant", "Insertion Flank", "Insertion Breakpoint", "Deletion Flank", "Deletion Breakpoint", and "Inside Deletion". In our view, this set of labels represents the different categories of windows that are likely to have informative features. However, it is possible that other labeling schemes might prove to represent the feature space better; this would be a useful area for a more thorough future study.

We also had to choose what portions of the data to train the model on, since it would be infeasible to train on the entire reference genome, and would likely bias the model against predicting any deletions because the vast majority of the reference does not participate in a deletion variant. First, we took all of the simulated insertion and deletion variants, and added regions of 250bp with the label "Deletion Flank" to either side. In addition, we selected 10,774 Cloudbreak calls. Some of these overlapped with the true variants, but the others added regions to the training set which had produced false positives. We then expanded all of the 9,497 regions selected for training by first adding 400bp to either end, and then adding the length of the resulting region to both sides. This ensured that all true variants and false positives were flanked by many windows that were properly labeled "Outside Variant".

To train our model, we coded it in Factorie and used their implementation of the LGBFS algorithm with L2 regularization. We also experimented with other optimizers including the online AdaGrad regularized dual averaging algorithm [49] with L1 regularization but did not see large differences in preliminary testing. A limited difference in results on the test data combined with faster training times guided our choice of LGBFS; however, exploring different optimization methods with different feature sets could still be a useful area for research.

7.7 Improving Cloudbreak Calls with CRF Predictions

For our initial implementation, we adopted the approach of rapidly identifying candidate regions using a fast tool (Cloudbreak in this instance), adding flanking regions to them, and then running the CRF model on those candidate regions to try to label the true variants. Of course, this bounds the recall of the CRF approach to the recall of the candidate regions selected by the preliminary screen, but allows for a more tractable computation than trying to label the entire genome.

For the test set, we used the same data set for individual NA18507 described in Section 6.3, with 37X coverage by high-quality 100bp paired end reads with an insert size of 300bp. To choose regions to test on, we used Cloudbreak predictions for the same data set, at a low threshold to increase sensitivity. We then created test windows to run inference on in the CRF model by again adding 200bp of flank to each side of each deletion prediction interval, and then adding the length of the interval in each direction. To conduct inference we use the Viterbi max-product belief propagation algorithm implemented in Factorie.

We then took all windows that the CRF model labeled deletions and merged contiguous blocks of windows to form intervals. For each such interval, we assign a confidence score as follows: first, for each window in the interval we calculate the sum of the likelihoods assigned to labels which indicate a deletion (i.e. have the "Deletion" bit set to one). We then compute the average of this window score across the entire deletion interval, giving us an overall CRF confidence in the deletion interval as a whole.

Finally, we refined the initial set of Cloudbreak deletion calls using the CRF calls according to the following rule: if a Cloudbreak deletion call and a CRF deletion call share a reciprocal overlap of at least 60%, we consider that Cloudbreak call to be confirmed by the CRF model. By reciprocal overlap, we mean that 60% of the length of the Cloudbreak call overlaps the CRF call, and 60% of the CRF call overlaps the Cloudbreak call. For confirmed calls, we take the boundaries of the CRF deletion interval to be the deletion variant boundaries, and multiply the Cloudbreak deletion likelihood ratio by the CRF confidence score to get an adjusted confidence score for the confirmed call.

7.8 Results

Figure 7.5 shows a ROC curve that compares the confirmed CRF calls with the other methods including Cloudbreak. The sensitivity of the merged CRF predictions is limited compared to the other methods. However, the updated score, which combines the Cloudbreak likelihood ratio with the CRF confidence, does provide a small improvement in discriminative power, as the true positives occur at higher thresholds than when using the unadjusted Cloudbreak score.

Although the CRF method provides only a limited benefit to discriminative power over the raw Cloudbreak calls, it proves to be excellent at improving the breakpoint resolution of the calls. Figure 7.6 shows the breakpoint resolution of each of the methods, as reported previously in Chapter 6, with the addition of the set of Cloudbreak calls confirmed by the CRF model. The confirmed calls have dramatically better resolution than the original Cloudbreak call set. With CRF confirmation, Cloudbreak's median difference between the true deletion length and the predicted length, 17bp, is now similar to that of DELLY-RP, which at 16bp is the best performing read-pair based method according to this metric (methods that take advantage of split-read mappings, such as DELLY-SR and Pindel, of course do better in this category). When one considers that Cloudbreak's predictions and the CRF features and labels are all based on the use of 25bp windows, it is clear that most of the time the CRF predictions are accurately identifying the window which contains the



Figure 7.5: ROC curve showing the accuracy of Cloudbreak calls that have been verified and refined with conditional random field predictions (Cloudbreak-CRF)

true breakpoint.

7.9 Features Selected by the CRF

It can often be informative to examine the features that are given high scores by a machine learning technique, as they can sometimes provide insights into non-obvious features associated with certain labels. It should be noted, however, that a simple list of the highest-scoring features does not necessarily give a complete picture of the features that were important for learning performance. For example, if several features are correlated or appear with each other only in certain types of interactions, individual features within that correlated group may not be given high scores even though the group as a whole may have been quite important in learning the training data. This problem is compounded by the use of regularizers in training; in the case of L2 regularization, as used in these results,



Figure 7.6: Breakpoint resolution of each tool's correct deletion predictions on the NA18507 data set, including Cloudbreak calls which were verified and refined with CRF predictions (Cloudbreak-CRF)

correlated features will all be reduced in weight similarly by the regularization term, potentially leaving none of them with high weights even though the group of features was highly significant in training.

Despite these caveats, we can still identify several interesting features with high scores in the trained CRF model. Table 7.9 shows the top 25 features learned by the CRF for the window labels that correspond to "Deletion Breakpoint" (the windows which contain the endpoints of deleted regions) and "Insertion". The CRF has learned a strong association between the deletion breakpoint label and the presence of simple repeats, showing how much correspondence there is between those features and the variants in the training set. As we might expect given the large improvement that the model achieved in localizing breakpoint locations, soft clipping-related features are given high scores, particularly when they occur in conjunction with coverage drops. The mu2 feature is used heavily for both deletions and insertions. Further examination of the feature scores may allow for improvements in feature engineering to create more parsimonious and effective models.
| Deletion | | Insertion | | |
|---|-------|---|-------|--|
| Feature Name | Score | Feature Name | Score | |
| simpleRepeat | 0.6 | m softClip > 1.0 | 0.47 | |
| 260 < mu2 < 340 and simpleRepeat | 0.34 | simpleRepeat | 0.41 | |
| simpleRepeat neighbor | 0.34 | m softClip > 3.0 | 0.39 | |
| rdepth < .1 and simpleRepeat | 0.32 | $260 < \mathrm{mu2} < 340 	ext{ and softClip} > 1.0$ | 0.38 | |
| 260 < mu2 < 340 and simpleRepeat neighbor | 0.29 | ${ m softClip} > 1.0 { m neighbor}$ | 0.31 | |
| m softClip > 1.0 ~and ~covDrop20 > 5.0 | 0.23 | m softClip>2.0 | 0.29 | |
| m softClip > 3.0 ~and ~covDrop20 > 5.0 | 0.2 | ${ m mu2} < 260 { m ~in~region}$ | 0.26 | |
| ${ m mu2}<260~{ m and}~{ m lr}=0~{ m in}~{ m region}$ | 0.19 | $260 < \mathrm{mu2} < 340 	ext{ and softClip} > 3.0$ | 0.25 | |
| m softClip > 1.0 | 0.18 | rdepth < .1 and simpleRepeat | 0.24 | |
| m softClip > 3.0 ~and ~covDrop20 > 10.0 | 0.17 | rdepth > 1 and $simpleRepeat neighbor$ | 0.23 | |
| m softClip > 2.0 ~and ~covDrop20 > 5.0 | 0.17 | ${ m softClip}>2.0~{ m neighbor}$ | 0.22 | |
| mu2 > 340 and simpleRepeat | 0.17 | ${ m softClip} > 1.0 { m in region}$ | 0.22 | |
| covDrop20 > 5.0 in region | 0.17 | mu2 < 260 and simpleRepeat | 0.21 | |
| changePoint > 5.0 in region | 0.16 | simpleRepeat and repeat | 0.21 | |
| m softClip>2.0 ~and~ covDrop20>10.0 | 0.15 | singletons > 1.0 and covDrop20 > 5.0 | 0.21 | |
| covChangePoint > 10.0 in region | 0.15 | m softClip > 2.0 ~in ~region | | |
| $ m softClip > 1.0 \ neighbor$ | 0.14 | ${ m softClip}>3.0~{ m neighbor}$ | 0.2 | |
| mu2 > 340 in region | 0.14 | $260 < \mathrm{mu2} < 340 	ext{ and softClip} > 2.0$ | 0.2 | |
| $\mathrm{mu2}>340 \mathrm{~and~lr}>0.75$ | 0.14 | rdepth > 1 and $simpleRepeat$ | 0.2 | |
| changePoint > 15.0 and rdepth $< .5$ in region | 0.14 | ${ m softClip}>1.0 { m and rdepth}>1$ | 0.16 | |
| ${ m mu2}>340 { m ~and~ softClip}>1.0$ | 0.13 | mu2 < 260 neighbor | 0.16 | |
| m softClip > 1.0 ~and ~covDrop20 > 10.0 | 0.13 | lr > 0.75 and simpleRepeat | | |
| 260 < mu2 < 340 and $covDrop20 > 10.0$ neighbor | 0.13 | $260 < \mathrm{mu2} < 340 	ext{ and } \mathrm{lr} = 0 	ext{ in region}$ | 0.15 | |
| rdepth < .1 and covDrop20 > 5.0 neighbor | 0.13 | softClip > 1.0 and repeat | 0.15 | |
| $\mathrm{rdepth} < .25$ | 0.13 | changePoint > 5.0 and softClip > 3.0 | 0.14 | |

Table 7.2: Most important features learned by the CRF model for deletion and insertion breakpoints. "Neighbor": feature occurs in neighboring window. "In region": feature occurs within 300bp of current window. Feature names are the same as those in Table 7.1

7.10 Discussion

With the CRF model presented in this chapter, we have shown that the use of local features is a useful abstraction that can enable innovative algorithmic approaches to the SV detection problem. We redefined SV detection as a sequence labeling problem, and created a framework for generating and managing local features across the genome that can integrate read pair, split read, and read depth related features, as well as features that incorporate prior knowledge such as genome annotations. Despite only training with simulated data that likely does not incorporate all of the aspects of the SV detection problem that make it difficult, we were able to generate useful predictions from our model.

Although the improvements in accuracy we achieved with the CRF model are marginal at best, it did prove to be very effective in improving Cloudbreak's breakpoint resolution, which was one of the weak points of the initial implementation of Cloudbreak as described in the previous chapters. The fact that for the majority of correctly predicted variants the CRF model was able to correctly identify the windows in which the breakpoint occurred indicates that the resolution could potentially be further improved with reductions in window size. In addition, those windows identified could be excellent targets for exhaustive split read analysis or local assembly.

Opening the door to arbitrary features potentially allows the inclusion of a variety of types of information useful for SV detection in a single unified framework. For example, other other types of genomic annotations could be very useful for detecting where the SV signals might be distorted due to sequence content. In this work we used binary features indicating the presence of repeats and segmental duplications. There are more fine-grained measures of repetitiveness and its effect on short read alignment, such as the genome mappability score [100] that could help to distinguish difficult regions. In addition, GC content in genomic sequence is known to have an effect on sequencing depth and copy number estimation [19] and could be added as an additional feature to help correct for these biases. Another type of feature that similarly represents prior knowledge about the genome and could be incorporated into this framework is sets of predictions from different SV detection tools. The SV detection methods that have achieved the greatest overall accuracy for non-cancer domains are those that pool multiple samples from across the same population, such as Genome STRiP [65]. The pooled signals are used to identify candidate variant regions, which are then genotyped in each sample separately in a second pass over the data. These candidate variant regions could also be used as features within an integrated machine learning technique.

Finally, the characterization of the problem as a generic sequence labeling task over arbitrary features could potentially enable the use of a variety of different statistical and machine learning techniques for the SV detection problem in addition to the CRFs explored in this chapter. For example, deep learning techniques such as sequential deep belief networks [7] have recently been used very successfully in sequence processing tasks in speech recognition and other domains. Since it is difficult to find fully annotated training data for the SV detection problem, the fact that deep learning networks are amenable to semi-supervised training with unlabeled data [192] makes them a very attractive area for future research.

Chapter 8

Analysis of Evolutionary Breakpoint Features

This chapter describes a data analysis project unrelated to the algorithmic development of Cloudbreak and its extensions, although it is related at the biological level because it deals with genome breakpoints and rearrangements. In addition to being able to locate genomic breakpoints, we also seek to understand the genomic contexts in which they are located to gain insights into their origins. As we discussed in Chapter 2, the sequence features of the genome that neighbor structural variations can leave clues about the mechanisms that caused the rearrangements, as in the cases of large homologies that point to the activity of non-allelic homologous recombination (NAHR) or the microhomologies that indicate non-homologous end joining (NEHJ). In this chapter we examine a set of structural variations that have reshaped a genome at the evolutionary timescale: the chromosomal rearrangements of the gibbon genome. We analyze the locations in which the gibbon genome has been rearranged relative to humans and the other apes, first using a limited set of breakpoint data, and then using the entire gibbon genome reference sequence created by the International Consortium for Sequencing and Annotation of the Gibbon Genome. The associations we discover could help to provide insights into the mechanisms by which the gibbon genome was rearranged. The main contributions of this chapter are 1) the statistical analysis of the enrichment or depletion of certain genomic features in regions of the gibbon genome that have participated in genomic breakpoints, 2) development of an open-source software tool that can harness compute clusters to conduct permutation analysis for statistical enrichment of features in the genome, and 3) a novel, cross-species

analysis of the locations of binding sites of the transcription factor CTCF with respect to genome breakpoints in the gibbon family.

8.1 Background

In Chapter 2, we described how SVs can occur in the genomes of many different populations, ranging from rearrangements that occur in populations of cells within a tumor in a single individual, to SVs that are polymorphic within the population of a species, to the structural differences between the genomes of different species. This last type of rearrangement represents SVs that occurred within a population and then became fixed in that species and were preserved as it evolved and differentiated.

Rather than identifying them by comparing short-read sequencing data to a reference genome, as has been the focus of the previous chapters, we can detect these *evolutionary* breakpoints by directly comparing the genomes of species to one another. This can be done experimentally, by large fragments of DNA from one species onto the chromosomes of another; in Section 8.2 we will describe the analysis of data from one such experiment that used a technique based on bacterial artificial chromosomes (BACs). With the advent of whole genome sequences they can be found by conducting sequence alignments (or multiple sequence alignments) of the the reference genomes for the species under study. When species share regions that contain the same genes and other genomic features, in the same order, it is referred to as shared synteny. By examining the endpoints of syntenic blocks of the genomes of two species, or synteny breakpoints, it is possible to find the breakpoints of the structural variations that rearranged the structure of the two genomes.

Identifying evolutionary breakpoints can give insights into the mechanisms of their formation. It was long thought that chromosomal breakpoints are largely random, such that they are equally likely to occur at any location in the genome. This theory is known as the *random breakage* model [144, 134]. Higher-resolution analysis, however, revealed that certain regions of the genome, or "hot-spots", are much more likely to give rise to evolutionary breakpoints [148], and this model, called the *fragile breakage* model, was reconfirmed once whole-genome reference sequences for many species became available [132].

Tantalizingly, Murphy et al. also showed that these reused evolutionary breakpoint regions overlapped significantly with the breakpoints of SVs found in cancer genomes [132].

This raises the question of whether these regions of the genome have special properties that make them more likely to participate in evolutionary breakpoints. In Chapter 2, we described the fact that certain mechanisms for SV formation, in particular NAHR, are a result of having regions with high sequence similarity at different locations in the genome, as the cell's machinery attempts to repair double-stranded breaks in DNA by joining homologous regions together. As one might expect if evolutionary breakpoints were the result of NAHR, a large number of chromosomal rearrangements in mammals, approaching 40%, are associated with segmental duplications [15, 16]. Closer examination of breakpoints from a variety of species, including human, chimp, macaque, rat, mouse, pig, cattle, dog, opossum, and chicken, confirmed that they contain more sequence that is not unique relative to the rest of the genome than would be expected by chance, including segmental duplications (SDs) and other low-copy number repeats [97]. However, analysis of evolutionary sequence divergence in these repeats indicates that some of the duplications may have occurred after the rearrangements, casting doubt on the theory that an NAHR-like process is primarily responsible [15], and indeed the precise mechanisms behind evolutionary breakpoints remain unknown.

Gibbons represent a unique opportunity to study the process of evolutionary genomic rearrangement because of their highly rearranged genomes. In most branches of the evolutionary tree, large genomic rearrangements are rare events, occurring at a rate of approximately two every ten million years [193]. However, some species appear to have faster or slower rates of rearrangements. For example, the orang-utan genome has very few medium and large scale rearrangements relative to the expected number given its time of evolutionary divergence [114]. At the other end of the spectrum, gibbon genomes have 10 to 20 times more rearrangements than would be expected given the rate in other mammals, and in fact the four genera of gibbons have numbers of chromosomes ranging from 28 to 52, suggesting that gibbon genome has been shuffled rapidly since it diverged from the other apes 17 to 18 million years ago [130].

Given their highly rearranged karyotypes, gibbons represent an excellent model in

which to study mechanisms of chromosomal rearrangements. Until recently, however, the whole genome sequence of the gibbon was not available. Therefore, several studies analyzed gibbon genome breakpoints using techniques involving BACs [62, 30, 29]. BACs are medium sized (150-350kb) fragments of DNA from a sample that are isolated, turned into plasmid bacterial chromosomes, and then amplified in a culture. By probing gibbon BACs with arrays or testing their hybridization to human chromosomes using fluorescent in situ hybridization (FISH) experiments, it is possible to isolate BACs which span human-gibbon evolutionary breakpoints. This allows one to locate the breakpoint regions to within several hundred kilobases, and the small size of the BACs relative to the entire genome makes them much easier to sequence, allowing the analysis of sequence features near the breakpoints. Initial studies examined selected BACs from the northern white-cheeked gibbon species Nomascus leucogenys leucogenys (NLE) and found a high level of segmental duplications and repetitive sequence at the breakpoints [30, 159]. Similar results were found in BACs from the white-handed gibbon species Hylobates lar (HLA) [130]. A later study increased the scope by examining 24 sequenced NLE BACs that spanned breakpoints, and analyzed gibbon-specific insertions of repeats and segmental duplications at the breakpoints to postulate several possible rearrangement mechanisms [62]. More recently, Carbone et al. [29] analyzed 57 NLE BACs and, in addition to segmental duplications, found enrichment for Alu repetitive elements. The latter study also found differences in CpG methylation marks on Alu elements near the breakpoints, suggesting that there may be an epigenetic process involved in gibbon genomic rearrangements.

8.2 Evolutionary Breakpoints in the Gibbon Genome Identified by BACs

To determine whether or not the characteristics observed in the breakpoints identified by BACs from NLE and HLA gibbons were generalizable to the entire gibbon family, we conducted an expanded analysis of gibbon breakpoints based on the identification of additional BACs covering breakpoints in the remaining two gibbon genera using samples from the species *Symphalangus syndactylus* (SSY) and *Hoolock leuconedys* (HLE). Since each of the gibbon genera has a different karyotype, identifying breakpoints across all four allows for greater insight into the mechanisms and timing of the evolution of the gibbon family. I contributed to this research by analyzing the overlap between breakpoints from all four gibbon genera with genomic features. This work was published in Capozzi et al. [28].

8.2.1 Methods

To determine whether there exists an enrichment or depletion for genomic features associated with gibbon chromosomal breakpoints we computed the significance of their overlap using Monte Carlo permutation tests. The purpose of the permutation analysis is to discover the null distribution for the number of overlaps a set of intervals has with a particular feature in the genome. In essence, the test asks: if my intervals were placed in random locations on the genome, what is the probability of seeing the number of overlaps we observed with that features in the actual data? In this case, the intervals whose locations we permute are the gibbon breakpoint regions.

For this test, we identified the human regions that are syntenic to the locations of the breakpoints in gibbons. We then permuted their start coordinates 10,000 times using BEDTools version 2.16.2 [155], while maintaining the chromosomal assignment and length of breakpoint regions. Genomic regions annotated as centromeres and telomeres in the "Gaps" track of the hg19 build were excluded from possible random placements of the regions. Locations of the features were held constant. We then compared the actual number of features that overlapped a breakpoint region to the distribution of overlap counts among the randomly permuted regions, and used the quantile of the real observed value in that distribution as an estimate of the p-value of observing a value equal to or greater than the real observation. For events that are rare, it is necessary to consider a large number of permutations or Monte Carlo samples in order to accurately estimate the p-value [160]. In order to facilitate conducting many permutations, we created a pipeline for distributing the necessary computation across a compute cluster using the grid management system HTCondor [181].

The analysis was performed on the human hg19 assembly. The features examined were genes, human segmental duplications, and some repeat families (Alu, LINE, ERV, and

SVA). We also investigated the associations between breakpoint regions and chromatin structure by testing the overlap with open chromatin regions in human embryonic stem cells reported by the ENCODE consortium [52].

8.2.2 Results

We found a significant enrichment for genes (Bonferroni adjusted P-value = 0.0287), human segmental duplications (P = 0.0366), Alu (P < 0.0001), and SVA (P = 0.0008) Figure 8.1 displays the histograms of overlap counts in the random distributions for segmental duplications, Alu elements, and SVA elements. We did not find significant enrichment for LINE and ERV repeats, nor for the ENCODE open chromatin regions. SVA elements are not found in gibbons; therefore, their status as significantly enriched in the human syntenic breakpoint regions is surprising. However, SVAs are known to correlate with Alu elements due to their preference for G+C rich regions of the genome (Wang et al. 2005). It seems likely that the association between human SVA locations and gibbon breakpoint regions is therefore an indirect one, dependent on the presence of additional genomic features present in both humans and gibbons.

Systematically shifting the location of breakpoint regions by increments of 10 kb upand downstream of their actual location, up to a maximum of 1 MB, shows that the locations of the breakpoint regions gives the greatest or close to the greatest number of overlaps with the four significantly overlapping features (genes, segmental duplications, *Alu*, and SVA) in the local genomic neighborhood, also shown in Figure 8.1.

8.3 Analysis of Breakpoints from the Gibbon Genome Reference Sequence

While analysis using breakpoints identified with BAC clones has provided useful results, the International Consortium for Sequencing and Annotation of the Gibbon Genome has recently created a draft assembly of the whole gibbon genome for an NLE individual. Comparison of the assembly sequence with the human reference genome has enabled the identification of a complete set of synteny breakpoints between the human and gibbon



Figure 8.1: Enrichment of genomic features in regions of the human genome syntenic to gibbon breakpoint regions identified using BACs. Permutation tests were used to assess the overlap between the gibbon breakpoints and genomic features. (A) Segmental duplications; (B) *Alu* elements; and (C) SVA elements. The black vertical line indicates the observed value for the breakpoints identified in the study. In all three cases it is evident that the genomic features have a higher overlap with the breakpoints than one could expect by chance. Figure reproduced from Capozzi et al. [28].

genomes at single nucleotide resolution. This allows for a much more comprehensive picture of the sequence features surrounding gibbon breakpoints than was previously possible. As a member of the project, I was responsible for the breakpoint analyses described below.

8.3.1 Overlap with Genomic Features: Repeats and Genes

To analyze the enrichment of genomic features in the regions flanking evolutionary breakpoints, we used a similar permutation based approach. We compared the number of overlaps between breakpoint flanks and each feature of interest in the assembled genome (the observed overlap count) compared to a background distribution estimated by randomly permuting the locations of breakpoint regions 100,000 times. The Nleu1.1 version of the assembly was used for these analyses. To create breakpoint regions, for those breakpoints for which we had single nucleotide resolution we added flanking regions to either side of the breakpoint; for breakpoints that fell within a gibbon-specific repeat element, we chose the flanking regions of the repeat. In each permutation, the location of each breakpoint region was randomly changed, while keeping its length and scaffold assignment the same. We then counted the number of overlaps between the randomized breakpoint regions and the feature of interest (the permuted overlap count). Enrichment p-values were computed as the proportion of permuted overlap counts that were more extreme than the observed overlap count. We also visualized the spatial relationship of breakpoint regions to each type of feature by simultaneously shifting the locations of the breakpoint regions up to 1MB in each direction, in increments of 25kb, and counting the proportion of shifted breakpoint regions that overlapped a feature of interest, after discarding regions that were shifted beyond the beginning or end of a scaffold. Permutation testing and shift testing were carried out using custom Python scripts and the BEDtools [155], pybedtools [43], and BEDOPS [136] libraries. For this project we enhanced our testing pipeline, adding command-line options to support the distribution of permutations across Condor clusters, and made the code publicly available at https://github.com/cwhelan/permuting-feature-enrichment-test.

We tested for enrichment in the breakpoint regions of genes, segmental duplications, and several classes of repetitive elements: Alu, L1, LAVA, and LTR. In addition to testing the entire Alu family, we also tested the subfamilies AluS, AluJ, and AluY individually.



Figure 8.2: Enrichment or depletion of gibbon genome breakpoints for several features. The number of overlaps between the 1kb regions flanking gibbon-human synteny breakpoints and segmental duplications, *Alu* elements, and genes in the gibbon genome is shown by the arrow labeled "breakpoints". Histograms represent the random distribution for these overlap counts obtained by permuting the locations of the breakpoints 100,000 times.

Gene locations were taken from Ensembl build 70. For the segmental duplication analysis, we used the segmental duplications identified by our collaborators in the Eichler laboratory using the WSSD method [17]. *Alu*, L1, and LTR locations were identified using RepeatMasker output. In order to determine the distance from the breakpoints at which enrichments are strongest, we varied the size of the breakpoint flanking regions by adding differently sized intervals; we tested flanking regions of size 100bp, 250bp, 500bp, and 1000bp. We corrected for multiple testing using the FDR under dependency method of Bejamini and Yekutieli [20].

Breakpoint regions are depleted for genes, but enriched for Alu elements and segmental duplications (Figure 8.2, Table 8.1). The enrichment for Alu becomes particularly strong at distances of 750bp or 1000bp from the breakpoint (Figure 8.3). The enrichment for Alu is primarily due to a strong enrichment of the AluS subfamily.

We also conducted a shift analysis similar to the one we described in Section 8.2. Breakpoint regions were simultaneously shifted in increments of 25kb, up to a maximum of 1MB in each direction, and the proportion of breakpoint regions that overlap a feature of interest is reported. In Figure 8.4, the shifts show that breakpoints are centered on regions that are depleted in genes but close to regions that contain genes, while the opposite is true for segmental duplications. Alu elements are more evenly spread across the shift regions.

| Feature | Breakpoints that Overlap | Quantile | Adjusted -1 * $\log(P-value)$ |
|----------------------------|--------------------------|----------|-------------------------------|
| Alu | 109 | 1.0000 | 8.8696 |
| A l u J | 35 | 0.9952 | 3.2191 |
| AluS | 81 | 1.0000 | 8.8696 |
| AluY | 26 | 0.9918 | 2.7599 |
| CTCF Peak | 12 | 0.9990 | 4.5930 |
| Gene | 22 | 0.0000 | -8.8696 |
| L1 | 70 | 0.9516 | 1.2222 |
| LAVA | 1 | 0.9755 | 1.7284 |
| LTR | 48 | 0.9697 | 1.6054 |
| WSSD Segmental Duplication | 35 | 1.0000 | 8.8696 |

Table 8.1: Enrichment counts and scores of features in breakpoint flanking regions. For each feature type, we display the number of 1kb regions flanking breakpoints that overlap with a feature of that type, the quantile of that count in the empirical distribution obtained by permuting breakpoint flank locations 100,000 times, and the negative log FDR of that quantile treated as a p-value. Negative values indicate a depletion rather than an enrichment. Prior to FDR correction quantiles of zero were adjusted to p-values of 0.00001.

In conjunction with our collaborator Larry Wilhelm, in addition to testing the count of overlaps between breakpoint flanking regions and repeats, we also conducted a complementary test that examined the distance of each breakpoint to the nearest repeat of a given class. For this test we used only the 44 breakpoints for which we had single nucleotide breakpoint resolution, with no Gibbon-specific repeats directly intersecting the breakpoint. We compared these breakpoint locations to 10,000 randomly selected regions in the nomLeu2 genome. For each random location, we determined the distance to the nearest repeat. We compared the distribution of distances to a repeat from the breakpoints to the distribution of distances to a repeat from the randomly selected positions using a Kolmogorov-Smirnov test. We examined the distance to any repeat, as well as those for Alu, LINE, and LTR elements, and finally the AluJ, AluS, and AluY subfamilies. After FDR correction for multiple hypothesis testing, the distance test showed similar results to the overlap test described above, with significant results for the Alu family as a whole, and for the AluJ and AluS subfamilies, indicating that the breakpoints tend to be closer to those repeats than random locations in the genome.



Figure 8.3: Enrichment or depletion of regions of the gibbon genome for various features, as a function of distance from the breakpoint. Positive y values are the negative log q-values of enrichment for each feature in the regions flanking the breakpoint of the size indicated on the x-axis. Negative y-values represent the q-values of depletion. Q values were calculated using the FDR correction under dependency procedure. The dotted line indicates an FDR threshold of 0.05. Breakpoints are enriched for Alu elements, and particularly the AluSsubfamily, especially at distances of 750bp and 1000bp from the breakpoints. Breakpoints are also enriched for segmental duplications, depleted for genes, and enriched for CTCF binding sites (see Section 8.3.2.)

8.3.2 Overlap with CTCF Binding Sites

Using the gibbon genome reference, we conducted another set of analyses to determine the relationship of gibbon genome breakpoints with binding sites of the evolutionarily conserved binding factor CTCF. CTCF is a transcription factor with many functions,



Figure 8.4: The proportion of breakpoint flanking intervals (defined as the 1000bp intervals flanking exact breakpoint locations or gibbon-specific repeats located at the breakpoints) which overlap with a feature of interest, when the flanking intervals are shifted left or right from their actual locations. Breakpoints are located in locations enriched in Alu (especially AluS) and segmental duplications relative to their genomic neighborhoods, but which are depleted for genes relative to their genomic neighborhoods.

including the ability to separate regulatory domains in the genome from one another [149]. CTCF has also recently been shown to be associated with the three-dimensional structure of DNA and chromatin in the cell [48], and therefore may have important associations with genomic DNA breakpoints and structural variations. We examined CTCF in the gibbon through the analysis of chromatin immunoprecipitation followed by sequencing (ChIP-seq) data [147]. Briefly, in ChIP-seq DNA from a sample is cross-linked *in vivo* and then fragmented, so that proteins (in particular transcription factors) remain bound to the fragments of DNA that contain their binding sites. After selecting only those fragments with a bound protein of interest using immunoprecipitation, the DNA is then sequenced. Mapping the reads back to the genome reference sequence creates "peaks" of coverage along the genome that indicate the location of transcription factor binding sites. In this analysis we identified a set of CTCF peaks from other primate species to separate CTCF peaks shared with a common ancestor from those that are unique to gibbons.

ChIP-seq for CTCF

CTCF ChIP-seq assays were performed in the Carbone laboratory by Elizabeth Terhune and Michelle Ward (visiting from the laboratory of Duncan Odom), according to the protocols described in Schmidt et al. [166] on eight EBV-transformed gibbon lymphoblastoid cell lines. In brief, CTCF-bound DNA was immunoprecipitated using an Anti-CTCF rabbit polyclonal antibody (07-729, Millipore). End-repair was performed on immunoprecipitated and input DNA prior to A-tailing and ligation to single-end Illumina sequencing adapters. DNA was amplified using Illumina primers 1.1 and 2.1 in an 18-cycle PCR reaction. Gel electrophoresis was used to select 200-300 bp DNA fragments. DNA libraries were sequenced using 36 bp reads on an Illumina Genome Analyser II according to the manufacturer's instructions.

Peak Calling from CTCF ChIP-seq Data

We aligned reads to the nomLeu2 reference using BWA (version 0.62) [106] with default parameters, and removed non-uniquely mapping reads. We then called peaks using CCAT [196], with parameters fragmentSize 100, slidingWindowSize 150, movingStep 10, isStrandSensitiveMode 1, minCount 10, minScore 4.0, and bootstrapPass 50. We then combined the peaks called across the different individuals and chose the following set for further analysis: any peak called in an individual individual by CCAT with an FDR of less than 0.05, as well as any peak that was called in more than one individual with an FDR of less than 0.1.

Determination of Gibbon-specific and Shared CTCF Binding Sites

We classified Gibbon peaks as shared or gibbon-specific by comparing them to a set of CTCF peaks called on human, macaque, and orangutan individuals [168]. This work was carried out in conjunction with our collaborators Javier Herrero of the European Bioinformatics Institute and Larry Wilhelm. First, orthologous locations of gibbon CTCF peaks in other primate species were determined using a local installation of the EnsemblCompara multi-species alignment database [53, 186]. This database contains alignments of the references for human (GRCh37), chimp (CHIMP2.1.4), gorilla (gorGor3.1), orangutan (PPYG2), rhesus macaque (MMUL_1), and gibbon (Nleu1.1). Gibbon CTCF peaks for which no multi-species alignment were present, or where the nucleotide alignment identity to human and macaque was less than 70%, were excluded from analysis. We then created a non-redundant list of non-gibbon CTCF peaks by converting the human, orangutan and macaque peaks to gibbon coordinates using the Compara database and removing redundant entries. Shared and gibbon-specific CTCF peaks were then identified as those gibbon CTCF peaks that did or did not intersect a peak in the non-redundant list of non-gibbon CTCF peaks.

Analysis of Enrichment for CTCF Binding Sites in Gibbon Genome Breakpoints

We identified 52,685 gibbon CTCF binding sites from the eight individuals. We found that 12 of the 1kb regions flanking breakpoints overlap with CTCF binding sites (see Figure 8.5 for an example). Using the same permutation analysis described above, this overlap has an enrichment p-value of 0.0009. This effect was even stronger when we expanded the breakpoint flanking regions to 25kb, as 95 expanded flanking regions overlap CTCF peaks for an enrichment p-value < 0.0001.

We then tested whether the CTCF peaks causing the enrichment in breakpoint regions are specific to gibbons. Using the CTCF ChIP-seq data from other species described above, we classified CTCF binding sites as unique to gibbons (11,449 sites) or shared with a primate ancestor (41,236 sites). We found that the gibbon breakpoint regions are heavily enriched for CTCF binding sites shared with a primate ancestor (enrichment pvalue = 0.0006) but are not significantly enriched for gibbon-specific CTCF binding sites. Again, the enrichment is stronger in the 25kb expanded breakpoint regions (enrichment p-value <0.0001) for shared binding sites, but not for gibbon-specific binding sites. This suggests that the formation or selection of gibbon genome rearrangements was associated with ancestral CTCF binding sites.

8.4 Discussion

In our work examining breakpoints that included data from HLE and SSY gibbons, we extended previous analyses that had only considered data from NLE. This confirmed that the associations between breakpoints and segmental duplications, genes, and repetitive elements of the *Alu* family previously observed in NLE extended to breakpoints in other genera of the gibbon family. Extending this analysis to the other genera should assist the study of gibbon chromosome evolution, especially when used in conjunction with the release of the gibbon genome reference, which is based on data from an NLE individual.

In Capozzi et al.[28], we were surprised to find an association between breakpoints and genes, given that most genome breaks will have deleterious effects if they occur in coding or regulatory sequence. However, we noted that those results were based on analyzing the features of the human genome reference that are orthologous to the gibbon breakpoint regions, and that the result might change when the full gibbon genome sequence was available to be tested. Indeed, that proved to be the case, as the depletion for genes in the breakpoints from the gibbon reference data shows.

Our discovery that gibbon genome breakpoints are enriched for CTCF binding sites



coverage depth from the eight gibbon CTCF ChIP-seq samples, clearly showing a peak representing a CTCF binding site. The Figure 8.5: An example of CTCF binding sites in a breakpoint region. The track labeled "CTCF ChIP-seq" shows merged read "Breakpoints" track shows the location of a gibbon breakpoint region. This breakpoint also contains a protein-coding gene, GH2-202.

indicates that there is a connection between regions with similar regulatory environments and the genome rearrangements preserved in the gibbon family, and the fact that breakpoints are only enriched for CTCF binding sites shared with gibbon's common ancestors suggests that the heavy rearrangement of the gibbon genome may have had to respect existing regulatory domains; breakages in other locations may have been too deleterious to survival to be preserved through evolution. In addition, the interaction between genomic structural variation and the three-dimensional structure of DNA in the cell is just beginning to be explored [48], but it is interesting to hypothesize that locations where the three-dimensional conformation of chromatin in the cell change, as represented by CTCF binding sites, might be vulnerable to genomic breakage under certain circumstances that may have been present in the gibbon's evolutionary past.

Chapter 9

Future Work

In this dissertation we have demonstrated that distributed computing, in the form of Hadoop and MapReduce, can be applied to the SV detection problem, enabling highly accurate algorithms with very fast runtimes. In addition, we showed that the use of discriminative sequence labeling techniques, in the form of conditional random fields, can be used to integrate many different signals of SVs and by doing so can improve Cloudbreak's results. We hope that the work presented in this dissertation is not only useful in itself but is extensible enough to contribute to additional innovations in the detection of structural variations.

9.1 Cloudbreak

There are many possible extensions that could be made to enhance Cloudbreak's effectiveness as a general SV analysis tool for high-throughput sequencing data. These include:

• Support the detection of additional SV types such as longer deletions, inversions, and translocations. These would be useful and necessary additions to a complete variant detection pipeline. We hope that the abstract formulation of our MapReduce framework in Chapter 4 will create a base from which to extend Cloudbreak's capabilities and implement many additional SV detection algorithms in the future. For example, detection of small to medium-sized inversions could be realized by adding the orientation of mapped read pairs as an additional component of the mixtures used in Cloudbreak's feature generation function. One difficulty in this effort will be the relative lack of validated SVs that are not deletions and inversion to develop and test

against, but as large scale projects such as the 1000 Genomes continue to generate data this concern should be mitigated.

- Extend Cloudbreak's GMM-based feature generation function to relax the assumption of a diploid genome sample. Currently Cloudbreak forces a GMM with two components. This assumption would not apply for some sequencing projects, especially in cancer-related research. In a heterogeneous tumor sample, for example, there may be multiple tumor sub-clones, as well as normal DNA, in the sequenced sample, and each set of input DNA might have different genotypes. With additional modeling of these mixtures, it might be possible for Cloudbreak not only to detect different variants in a sample but also to estimate their relative abundances, something that would be useful for tracking tumor clone evolution.
- Add a local assembly step to increase breakpoint resolution and validate SV candidates. Cloudbreak's breakpoint resolution is lower than that of many other RP algorithms. Although we improved that result in Chapter 7, we could not obtain single nucleotide resolution, because our features were defined in terms of fixedsize genomic windows. As we mentioned in Chapter 3, one approach to improving breakpoint resolution, and providing additional validation of predicted results, is to attempt to conduct a *de novo* assembly of the reads that mapped near the breakpoints as well as their paired sequences. If effectively implemented, this would be a very useful addition to any SV detection algorithm. Theoretically, each breakpoint could be assembled simultaneously in reduce tasks in an additional MapReduce job. This would greatly increase the confidence and utility of Cloudbreak's predictions.
- Support the simultaneous analysis of multiple samples and libraries. Approaches such as Genome STRiP [65] have demonstrated that they can achieve higher accuracy than single-sample SV detection algorithms by simultaneously considering many samples at once. Even if a variant is only supported by a very low number of reads in each individual sample, by pooling data from multiple samples the presence of a variant can become clear. Additional modeling in Cloudbreak of multiple samples, each with its own insert size distribution, might provide a similar increase in power.

Alternatively, data could be pooled in Cloudbreak and the resulting predictions could be used as features in our discriminative machine learning models.

• Adopt emerging standardized Hadoop libraries and file formats. As more effort is put into developing Hadoop-based sequencing analysis applications, it will become increasingly important that these applications use consistent APIs and data formats so that they can be composed into large pipelines. Once they contain the requisite features, refactoring Cloudbreak to use libraries such as Hadoop-BAM [138] would be a step in that direction.

9.2 SV Detection with Discriminative Machine Learning

We believe that the application of CRFs to SV detection we described in Chapter 7 is very promising and could be extended in many directions:

- Construction of improved training data sets. The most obvious limiting factor for this project was the training and test data used. Training on simulated data ignores many potential sources of noise as well as possible signals that could be used to detect SVs. Training on real data sets, such as those that are being generated for the 1000 Genomes project, would allow much greater potential for learning.
- Using additional or different sets of labels. In some sequence labeling tasks, it can improve performance to model the start and end of variants with separate labels, or to use different types of labels for flanking sequence. It would be useful to test different labeling schemes to determine the optimal one for this task. In addition, the authors of forestSV [126] demonstrated some utility in directly labeling regions known to cause false positives for other algorithms in training their classifier; this might be useful in our model as well.
- Add additional features to the model. There are many other possible features that could be added to our model. One example would be possible SNVs near the breakpoint locations, indicated by mismatches between the reads and the reference. There are also many other possible genome annotations or sources of prior knowledge that

could be used as features, including the locations of previously identified variants in other samples.

- Learning for additional variant types. As mentioned above, it would be useful to train our model to detect additional variant types by expanding the label set and set of features used
- Alternative learning methods. Recently, many advances have been made in machine learning for sequencing labeling tasks through the use of deep learning using artificial neural networks. With additional data to train on, it may be possible to apply these techniques to the SV detection problem.

9.3 Gibbon Genome Breakpoint Analysis

The identification of enrichment of genomic features at the location of gibbon-human synteny breakpoints can be used as background for further research into the evolutionary history of gibbons, with an ultimate goal of understanding the processes that caused them:

- Analysis of epigenetic marks associated with breakpoints. The Carbone lab has been investigating epigenetic signatures near genomic breakpoints, and has found distinct patterns of DNA methylation in breakpoint-associated transposable elements [99]. DNA methylation may have a role in the formation or conservation of rearranged genomes. Further analysis may reveal new mechanisms of gibbon genome rearrangement.
- Exploration of association of large DNA structures and domains with breakpoints. The association of CTCF binding sites with gibbon breakpoints suggests a possible relationship between rearrangement sites and the three dimensional structure of DNA. Alternatively, CTCF's role as separator of regulatory domains may have rendered genome rearrangements that did not respect those boundaries unviable, causing only those present in today's gibbons to be preserved. Additional analysis of other types of data may allow us to gain insight into these questions.

Bibliography

- 1000 GENOMES PROJECT CONSORTIUM, ABECASIS, G. R., AUTON, A., BROOKS, L. D., DEPRISTO, M. A., DURBIN, R. M., HANDSAKER, R. E., KANG, H. M., MARTH, G. T., AND MCVEAN, G. A. An integrated map of genetic variation from 1,092 human genomes. *Nature 491*, 7422 (Nov. 2012), 56–65.
- [2] ABYZOV, A., URBAN, A. E., SNYDER, M., AND GERSTEIN, M. CNVnator: an approach to discover, genotype, and characterize typical and atypical CNVs from family and population genome sequencing. *Genome Research* 21, 6 (June 2011), 974–984.
- [3] AFGAN, E., BAKER, D., CORAOR, N., CHAPMAN, B., NEKRUTENKO, A., AND TAYLOR, J. Galaxy CloudMan: delivering cloud compute clusters. *BMC Bioinformatics* 11, Suppl 12 (2010), S4.
- [4] ALKAN, C., COE, B. P., AND EICHLER, E. E. Genome structural variation discovery and genotyping. *Nature Reviews Genetics* 12, 5 (May 2011), 363–376.
- [5] ALKAN, C., KIDD, J. M., MARQUES-BONET, T., AKSAY, G., ANTONACCI, F., HORMOZDIARI, F., KITZMAN, J. O., BAKER, C., MALIG, M., MUTLU, O., SAHI-NALP, S. C., GIBBS, R. A., AND EICHLER, E. E. Personalized copy number and segmental duplication maps using next-generation sequencing. *Nature Genetics* 41, 10 (Oct. 2009), 1061–1067.
- [6] ALKAN, C., SAJJADIAN, S., AND EICHLER, E. E. Limitations of next-generation genome sequence assembly. *Nature Methods* 8, 1 (Jan. 2011), 61–65.
- [7] ANDREW, G., AND BILMES, J. Sequential deep belief networks. In Acoustics, Speech and Signal Processing (ICASSP), 2012 IEEE International Conference on (2012), pp. 4265–4268.
- [8] ANDREWS, S. FastQC: A Quality Control tool for High Throughput Sequence Data. http://www.bioinformatics.babraham.ac.uk/projects/fastqc/.

- [9] ANGIUOLI, S. V., MATALKA, M., GUSSMAN, A., GALENS, K., VANGALA, M., RILEY, D. R., ARZE, C., WHITE, J. R., WHITE, O., AND FRICKE, W. F. CloVR: A virtual machine for automated and portable sequence analysis from the desktop using cloud computing. *BMC Bioinformatics* 12, 1 (2011), 356.
- [10] ANNALA, M. J., PARKER, B. C., ZHANG, W., AND NYKTER, M. Fusion genes and their discovery using high throughput sequencing. *Cancer Letters 340*, 2 (Nov. 2013), 192–200.
- [11] APACHE SOFTWARE FOUNDATION. Hbase. http://hbase.apache.org.
- [12] APACHE SOFTWARE FOUNDATION. Pig. http://pig.apache.org.
- [13] APACHE SOFTWARE FOUNDATION. Whirr. http://whirr.apache.org.
- [14] ATEN, E., WHITE, S. J., KALF, M. E., VOSSEN, R. H. A. M., THYGESEN, H. H., RUIVENKAMP, C. A., KRIEK, M., BREUNING, M. H. B., AND DEN DUNNEN, J. T. Methods to detect CNVs in the human genome. *Cytogenetic and Genome Research* 123, 1-4 (2008), 313–321.
- [15] BAILEY, J. A., BAERTSCH, R., KENT, W. J., HAUSSLER, D., AND EICHLER, E. E. Hotspots of mammalian chromosomal evolution. *Genome Biology* 5, 4 (2004), R23.
- [16] BAILEY, J. A., AND EICHLER, E. E. Primate segmental duplications: crucibles of evolution, diversity and disease. *Nature Reviews Genetics* 7, 7 (July 2006), 552–564.
- [17] BAILEY, J. A., GU, Z., CLARK, R. A., REINERT, K., SAMONTE, R. V., SCHWARTZ, S., ADAMS, M. D., MYERS, E. W., LI, P. W., AND EICHLER, E. E. Recent segmental duplications in the human genome. *Science 297*, 5583 (Aug. 2002), 1003–1007.
- [18] BARNETT, D. W., GARRISON, E. K., QUINLAN, A. R., STROMBERG, M. P., AND MARTH, G. T. BamTools: a C++ API and toolkit for analyzing and managing BAM files. *Bioinformatics* 27, 12 (June 2011), 1691–1692.
- [19] BENJAMINI, Y., AND SPEED, T. P. Summarizing and correcting the GC content bias in high-throughput sequencing. *Nucleic Acids Research* 40, 10 (May 2012), e72.
- [20] BENJAMINI, Y., AND YEKUTIELI, D. The control of the false discovery rate in multiple testing under dependency. Annals of Statistics (2001), 1165–1188.

- [21] BENKO, S., FANTES, J. A., AMIEL, J., KLEINJAN, D.-J., THOMAS, S., RAM-SAY, J., JAMSHIDI, N., ESSAFI, A., HEANEY, S., GORDON, C. T., MCBRIDE, D., GOLZIO, C., FISHER, M., PERRY, P., ABADIE, V., AYUSO, C., HOLDER-ESPINASSE, M., KILPATRICK, N., LEES, M. M., PICARD, A., TEMPLE, I. K., THOMAS, P., VAZQUEZ, M.-P., VEKEMANS, M., ROEST CROLLIUS, H., HASTIE, N. D., MUNNICH, A., ETCHEVERS, H. C., PELET, A., FARLIE, P. G., FITZ-PATRICK, D. R., AND LYONNET, S. Highly conserved non-coding elements on either side of SOX9 associated with Pierre Robin sequence. *Nature Genetics* 41, 3 (Mar. 2009), 359–364.
- [22] BENSON, G. Tandem repeats finder: a program to analyze dna sequences. Nucleic Acids Research 27, 2 (1999), 573–580.
- [23] BREIMAN, L. Random forests. Machine Learning 45, 1 (2001), 5–32.
- [24] BROAD INSTITUTE. Picard tools. http://picard.sourceforge.net.
- [25] BURROWS, M., AND WHEELER, D. J. A block-sorting lossless data compression algorithm. Tech. rep., Digital SRC Research Report, 1994.
- [26] CAHAN, P., LI, Y., IZUMI, M., AND GRAUBERT, T. A. The impact of copy number variation on local gene expression in mouse hematopoietic stem and progenitor cells. *Nature Genetics* 41, 4 (Apr. 2009), 430–437.
- [27] CAMPBELL, P. J., STEPHENS, P. J., PLEASANCE, E. D., O'MEARA, S., LI, H., SANTARIUS, T., STEBBINGS, L. A., LEROY, C., EDKINS, S., HARDY, C., TEAGUE, J. W., MENZIES, A., GOODHEAD, I., TURNER, D. J., CLEE, C. M., QUAIL, M. A., COX, A., BROWN, C., DURBIN, R., HURLES, M. E., EDWARDS, P. A. W., BIGNELL, G. R., STRATTON, M. R., AND FUTREAL, P. A. Identification of somatically acquired rearrangements in cancer using genome-wide massively parallel paired-end sequencing. *Nature Genetics* 40, 6 (June 2008), 722–729.
- [28] CAPOZZI, O., CARBONE, L., STANYON, R. R., MARRA, A., YANG, F., WHELAN, C. W., DE JONG, P. J., ROCCHI, M., AND ARCHIDIACONO, N. A comprehensive molecular cytogenetic analysis of chromosome rearrangements in gibbons. *Genome Research* (Aug. 2012).
- [29] CARBONE, L., HARRIS, R. A., VESSERE, G. M., MOOTNICK, A. R., HUMPHRAY, S., ROGERS, J., KIM, S. K., WALL, J. D., MARTIN, D., JURKA, J., MILOSAVL-JEVIC, A., AND DE JONG, P. J. Evolutionary Breakpoints in the Gibbon Suggest

Association between Cytosine Methylation and Karyotype Evolution. *PLoS Genetics* 5, 6 (June 2009), e1000538.

- [30] CARBONE, L., VESSERE, G. M., TEN HALLERS, B. F. H., ZHU, B., OSOEGAWA, K., MOOTNICK, A., KOFLER, A., WIENBERG, J., ROGERS, J., HUMPHRAY, S., SCOTT, C., HARRIS, R. A., MILOSAVLJEVIC, A., AND DE JONG, P. J. A highresolution map of synteny disruptions in gibbon and human genomes. *PLoS Genetics* 2, 12 (Dec. 2006), e223.
- [31] CHAN, Y. F., MARKS, M. E., JONES, F. C., VILLARREAL, G., SHAPIRO, M. D., BRADY, S. D., SOUTHWICK, A. M., ABSHER, D. M., GRIMWOOD, J., SCHMUTZ, J., MYERS, R. M., PETROV, D., JONSSON, B., SCHLUTER, D., BELL, M. A., AND KINGSLEY, D. M. Adaptive Evolution of Pelvic Reduction in Sticklebacks by Recurrent Deletion of a Pitx1 Enhancer. *Science 327*, 5963 (Jan. 2010), 302–305.
- [32] CHANG, F., DEAN, J., GHEMAWAT, S., HSIEH, W. C., WALLACH, D. A., BUR-ROWS, M., CHANDRA, T., FIKES, A., AND GRUBER, R. E. Bigtable: A distributed storage system for structured data. ACM Trans. Comput. Syst. 26, 2 (June 2008), 4:1–4:26.
- [33] CHANG, Y.-J., CHEN, C.-C., CHEN, C.-L., AND HO, J.-M. A de novo next generation genomic sequence assembler based on string graph and MapReduce cloud computing framework. *BMC Genomics* 13 Suppl 7 (2012), S28.
- [34] CHANG, Y.-J., CHEN, C.-C., HO, J.-M., AND CHEN, C.-L. De novo assembly of high-throughput sequencing data with cloud computing and new operations on string graphs. 2012 IEEE Fifth International Conference on Cloud Computing (2012), 155– 161.
- [35] CHEN, J., LU, Y., XU, J., HUANG, Y., CHENG, H., HU, G., LUO, C., LOU, M., CAO, G., XIE, Y., AND YING, K. Identification and characterization of NBEAL1, a novel human neurobeachin-like 1 protein gene from fetal brain, which is up regulated in glioma. *Molecular Brain Research 125*, 1-2 (June 2004), 147–155.
- [36] CHEN, K., CHEN, L., FAN, X., WALLIS, J., DING, L., AND WEINSTOCK, G. TIGRA: A Targeted Iterative Graph Routing Assembler for breakpoint assembly. *Genome Research* (Dec. 2013).
- [37] CHEN, K., WALLIS, J. W., MCLELLAN, M. D., LARSON, D. E., KALICKI, J. M., POHL, C. S., MCGRATH, S. D., WENDL, M. C., ZHANG, Q., LOCKE, D. P., SHI, X., FULTON, R. S., LEY, T. J., WILSON, R. K., DING, L., AND MARDIS,

E. R. BreakDancer: an algorithm for high-resolution mapping of genomic structural variation. *Nature Methods* 6, 9 (Aug. 2009), 677–681.

- [38] CHIANG, D. Y., GETZ, G., JAFFE, D. B., O'KELLY, M. J. T., ZHAO, X., CARTER, S. L., RUSS, C., NUSBAUM, C., MEYERSON, M., AND LANDER, E. S. High-resolution mapping of copy-number alterations with massively parallel sequencing. *Nature Methods* 6, 1 (Jan. 2009), 99–103.
- [39] CHIARA, M., PESOLE, G., AND HORNER, D. S. SVM²: an improved pairedend-based tool for the detection of small genomic structural variations using highthroughput single-genome resequencing data. *Nucleic Acids Research 40*, 18 (Oct. 2012), e145.
- [40] COCK, P. J. A., FIELDS, C. J., GOTO, N., HEUER, M. L., AND RICE, P. M. The Sanger FASTQ file format for sequences with quality scores, and the Solexa/Illumina FASTQ variants. *Nucleic Acids Research* 38, 6 (Apr. 2010), 1767–1771.
- [41] CONRAD, D. F., BIRD, C., BLACKBURNE, B., LINDSAY, S., MAMANOVA, L., LEE, C., TURNER, D. J., AND HURLES, M. E. Mutation spectrum revealed by breakpoint sequencing of human germline CNVs. *Nature Genetics* 42, 5 (May 2010), 385–391.
- [42] CONRAD, D. F., PINTO, D., REDON, R., FEUK, L., GOKCUMEN, O., ZHANG, Y., AERTS, J., ANDREWS, T. D., BARNES, C., CAMPBELL, P., FITZGERALD, T., HU, M., IHM, C. H., KRISTIANSSON, K., MACARTHUR, D. G., MACDONALD, J. R., ONYIAH, I., PANG, A. W. C., ROBSON, S., STIRRUPS, K., VALSESIA, A., WALTER, K., WEI, J., WELLCOME TRUST CASE CONTROL CONSORTIUM, TYLER-SMITH, C., CARTER, N. P., LEE, C., SCHERER, S. W., AND HURLES, M. E. Origins and functional impact of copy number variation in the human genome. Nature 464, 7289 (Apr. 2010), 704–712.
- [43] DALE, R. K., PEDERSEN, B. S., AND QUINLAN, A. R. Pybedtools: a flexible Python library for manipulating genomic datasets and annotations. *Bioinformatics* 27, 24 (Dec. 2011), 3423–3424.
- [44] DANECEK, P., AUTON, A., ABECASIS, G., ALBERS, C. A., BANKS, E., DE-PRISTO, M. A., HANDSAKER, R. E., LUNTER, G., MARTH, G. T., SHERRY, S. T., MCVEAN, G., DURBIN, R., AND 1000 GENOMES PROJECT ANALYSIS GROUP. The variant call format and VCFtools. *Bioinformatics* 27, 15 (Aug. 2011), 2156–2158.
- [45] DEAN, J., AND GHEMAWAT, S. MapReduce: Simplified data processing on large clusters. *Communications of the ACM 51*, 1 (2008), 107–113.

- [46] DEPRISTO, M. A., BANKS, E., POPLIN, R., GARIMELLA, K. V., MAGUIRE, J. R., HARTL, C., PHILIPPAKIS, A. A., DEL ANGEL, G., RIVAS, M. A., HANNA, M., MCKENNA, A., FENNELL, T. J., KERNYTSKY, A. M., SIVACHENKO, A. Y., CIBULSKIS, K., GABRIEL, S. B., ALTSHULER, D., AND DALY, M. J. A framework for variation discovery and genotyping using next-generation DNA sequencing data. *Nature Genetics* 43, 5 (Apr. 2011), 491–498.
- [47] DING, M., ZHENG, L., LU, Y., LI, L., GUO, S., AND GUO, M. More convenient more overhead: The performance evaluation of hadoop streaming. In *Proceedings* of the 2011 ACM Symposium on Research in Applied Computation (New York, NY, USA, 2011), RACS '11, ACM, pp. 307–313.
- [48] DIXON, J. R., SELVARAJ, S., YUE, F., KIM, A., LI, Y., SHEN, Y., HU, M., LIU, J. S., AND REN, B. Topological domains in mammalian genomes identified by analysis of chromatin interactions. *NATURE* 485, 7398 (Apr. 2012), 376–380.
- [49] DUCHI, J., HAZAN, E., AND SINGER, Y. Adaptive subgradient methods for online learning and stochastic optimization. *Journal of Machine Learning Research 12* (July 2011), 2121–2159.
- [50] EILERS, M., AND EISENMAN, R. N. Myc's broad reach. Genes and Development 22, 20 (Oct. 2008), 2755–2766.
- [51] EMDE, A.-K., SCHULZ, M. H., WEESE, D., SUN, R., VINGRON, M., KALSCHEUER, V. M., HAAS, S. A., AND REINERT, K. Detecting genomic indel variants with exact breakpoints in single- and paired-end sequencing data using SplazerS. *Bioinformatics* 28, 5 (Mar. 2012), 619–627.
- [52] ENCODE PROJECT CONSORTIUM. A user's guide to the encyclopedia of DNA elements (ENCODE). *PLoS Biology* 9, 4 (Apr. 2011), e1001046.
- [53] ENSEMBL. EnsemblCompara. http://uswest.ensembl.org/info/docs/api/ compara/index.html.
- [54] ESCARAMIS, G., TORNADOR, C., BASSAGANYAS, L., RABIONET, R., TUBÍO, J. M. C., MARTÍNEZ-FUNDICHELY, A., CÁCERES, M., GUT, M., OSSOWSKI, S., AND ESTIVILL, X. PeSV-Fisher: Identification of Somatic and Non-Somatic Structural Variants Using Next Generation Sequencing Data. *PLoS ONE 8*, 5 (May 2013), e63377.
- [55] ESCOT, C., THEILLET, C., LIDEREAU, R., SPYRATOS, F., CHAMPEME, M. H., GEST, J., AND CALLAHAN, R. Genetic alteration of the c-myc protooncogene

(MYC) in human primary breast carcinomas. Proceedings of the National Academy of Sciences of the United States of America 83, 13 (July 1986), 4834–4838.

- [56] EVANI, U. S., CHALLIS, D., YU, J., JACKSON, A. R., PAITHANKAR, S., BAIN-BRIDGE, M. N., JAKKAMSETTI, A., PHAM, P., COARFA, C., MILOSAVLJEVIC, A., AND YU, F. Atlas2 Cloud: a framework for personal genome analysis in the cloud. BMC Genomics 13 Suppl 6 (2012), S19.
- [57] FENG, X., GROSSMAN, R., AND STEIN, L. PeakRanger: A cloud-enabled peak caller for ChIP-seq data. BMC Bioinformatics 12, 1 (2011), 139.
- [58] FERRAGINA, P., AND MANZINI, G. Opportunistic data structures with applications. In Foundations of Computer Science, 2000. Proceedings. 41st Annual Symposium on (2000), pp. 390–398.
- [59] FISCHER, M., SNAJDER, R., PABINGER, S., DANDER, A., SCHOSSIG, A., ZSCHOCKE, J., TRAJANOSKI, Z., AND STOCKER, G. SIMPLEX: cloud-enabled pipeline for the comprehensive analysis of exome sequencing data. *PLoS ONE* 7, 8 (2012), e41948.
- [60] GHEMAWAT, S., GOBIOFF, H., AND LEUNG, S.-T. The google file system. In Proceedings of the Nineteenth ACM Symposium on Operating Systems Principles (New York, NY, USA, 2003), SOSP '03, ACM, pp. 29–43.
- [61] GIARDINE, B., RIEMER, C., HARDISON, R. C., BURHANS, R., ELNITSKI, L., SHAH, P., ZHANG, Y., BLANKENBERG, D., ALBERT, I., TAYLOR, J., MILLER, W., KENT, W. J., AND NEKRUTENKO, A. Galaxy: a platform for interactive large-scale genome analysis. *Genome Research 15*, 10 (Oct. 2005), 1451–1455.
- [62] GIRIRAJAN, S., CHEN, L., GRAVES, T., MARQUES-BONET, T., VENTURA, M., FRONICK, C., FULTON, L., ROCCHI, M., FULTON, R. S., WILSON, R. K., MARDIS, E. R., AND EICHLER, E. E. Sequencing human-gibbon breakpoints of synteny reveals mosaic new insertions at rearrangement sites. *Genome Research 19*, 2 (Feb. 2009), 178–190.
- [63] GOOGLE, INC. Snappy. https://code.google.com/p/snappy/.
- [64] HAMPTON, O. A., DEN HOLLANDER, P., MILLER, C. A., DELGADO, D. A., LI, J., COARFA, C., HARRIS, R. A., RICHARDS, S., SCHERER, S. E., AND MUZNY, D. M. A sequence-level map of chromosomal breakpoints in the MCF-7 breast cancer cell line yields insights into the evolution of a cancer genome. *Genome Research 19*, 2 (2009), 167–177.

- [65] HANDSAKER, R. E., KORN, J. M., NEMESH, J., AND MCCARROLL, S. A. Discovery and genotyping of genome structural polymorphism by sequencing on a population scale. *Nature Genetics* 43, 3 (Mar. 2011), 269–276.
- [66] HART, S. N., SARANGI, V., MOORE, R., BAHETI, S., BHAVSAR, J. D., COUCH, F. J., AND KOCHER, J.-P. A. SoftSearch: Integration of Multiple Sequence Features to Identify Breakpoints of Structural Variations. *PLoS ONE* 8, 12 (2013), e83356.
- [67] HASTINGS, P. J., LUPSKI, J. R., ROSENBERG, S. M., AND IRA, G. Mechanisms of change in gene copy number. *Nature Reviews Genetics* 10, 8 (Aug. 2009), 551–564.
- [68] HAYES, M., AND LI, J. Bellerophon: a hybrid method for detecting interchromosomal rearrangements at base pair resolution using next-generation sequencing data. *BMC Bioinformatics 14 Suppl 5* (2013), S6.
- [69] HAYES, M., PYON, Y. S., AND LI, J. A model-based clustering method for genomic structural variant prediction and genotyping using paired-end sequencing data. *PLoS ONE* 7, 12 (Dec. 2012), e52881.
- [70] HOMER, N. DNAA. http://sourceforge.net/apps/mediawiki/dnaa/.
- [71] HONG, D., RHIE, A., PARK, S.-S., LEE, J., JU, Y. S., KIM, S., YU, S.-B., BLEAZARD, T., PARK, H.-S., AND RHEE, H. FX: an RNA-Seq analysis tool on the cloud. *Bioinformatics* 28, 5 (2012), 721–723.
- [72] HORMOZDIARI, F., ALKAN, C., EICHLER, E. E., AND SAHINALP, S. C. Combinatorial algorithms for structural variation detection in high-throughput sequenced genomes. *Genome Research* 19, 7 (July 2009), 1270–1278.
- [73] HOSOKAWA, Y., SUZUKI, R., JOH, T., MAEDA, Y., NAKAMURA, S., KODERA, Y., ARNOLD, A., AND SETO, M. A small deletion in the 3'-untranslated region of the cyclin D1/PRAD1/bcl-1 oncogene in a patient with chronic lymphocytic leukemia. *International journal of cancer 76*, 6 (June 1998), 791–796.
- [74] HUANG, H., TATA, S., AND PRILL, R. J. BlueSNP: R package for highly scalable genome-wide association studies using Hadoop clusters. *Bioinformatics 29*, 1 (Nov. 2012), 135–136.
- [75] IQBAL, Z., CACCAMO, M., TURNER, I., FLICEK, P., AND MCVEAN, G. De novo assembly and genotyping of variants using colored de Bruijn graphs. *Nature Genetics* (Jan. 2012).

- [76] ISSA, S. A., KIENZLER, R., EL-KALIOBY, M., TONELLATO, P. J., WALL, D., BRUGGMANN, R., AND ABOUELHODA, M. Streaming support for data intensive cloud-based sequence analysis. *BioMed Research International 2013* (2013), 791051.
- [77] JIANG, Y., WANG, Y., AND BRUDNO, M. PRISM: pair-read informed split-read mapping for base-pair level detection of insertion, deletion and structural variants. *Bioinformatics* 28, 20 (Oct. 2012), 2576–2583.
- [78] JOURDREN, L., BERNARD, M., DILLIES, M.-A., AND LE CROM, S. Eoulsan: A Cloud Computing-Based Framework Facilitating High Throughput Sequencing Analyses. *Bioinformatics* (Apr. 2012).
- [79] KELLEY, D. R., SCHATZ, M. C., AND SALZBERG, S. L. Quake: quality-aware detection and correction of sequencing errors. *Genome Biology* 11, 11 (2010), R116.
- [80] KENT, W. J. BLAT: The BLAST-Like Alignment Tool. Genome Research 12, 4 (2002), 656–664.
- [81] KEUN KIM, D., HEE YOON, J., HWA KONG, J., KYUN HONG, S., AND JOO LEE, U. Cloud-scale snp detection from rna-seq data. In *Data Mining and Intelligent Information Technology Applications (ICMiA), 2011 3rd International Conference* on (2011), pp. 321–323.
- [82] KIDD, J. M., COOPER, G. M., DONAHUE, W. F., HAYDEN, H. S., SAMPAS, N., GRAVES, T., HANSEN, N., TEAGUE, B., ALKAN, C., ANTONACCI, F., HAUGEN, E., ZERR, T., YAMADA, N. A., TSANG, P., NEWMAN, T. L., TÜZÜN, E., CHENG, Z., EBLING, H. M., TUSNEEM, N., DAVID, R., GILLETT, W., PHELPS, K. A., WEAVER, M., SARANGA, D., BRAND, A., TAO, W., GUSTAFSON, E., MCKERNAN, K., CHEN, L., MALIG, M., SMITH, J. D., KORN, J. M., MCCARROLL, S. A., ALTSHULER, D. A., PEIFFER, D. A., DORSCHNER, M., STAMATOYANNOPOULOS, J., SCHWARTZ, D., NICKERSON, D. A., MULLIKIN, J. C., WILSON, R. K., BRUHN, L., OLSON, M. V., KAUL, R., SMITH, D. R., AND EICHLER, E. E. Mapping and sequencing of structural variation from eight human genomes. *Nature 453*, 7191 (May 2008), 56–64.
- [83] KOBOLDT, D. C., LARSON, D. E., CHEN, K., DING, L., AND WILSON, R. K. Massively parallel sequencing approaches for characterization of structural variation. *Methods in Molecular Biology 838* (2012), 369–384.
- [84] KORBEL, J. O., ABYZOV, A., MU, X. J., CARRIERO, N., CAYTING, P., ZHANG, Z., SNYDER, M., AND GERSTEIN, M. B. PEMer: a computational framework with

simulation-based error models for inferring genomic structural variants from massive paired-end sequencing data. *Genome Biology* 10, 2 (2009), R23.

- [85] KORBEL, J. O., URBAN, A. E., AFFOURTIT, J. P., GODWIN, B., GRUBERT, F., SIMONS, J. F., KIM, P. M., PALEJEV, D., CARRIERO, N. J., DU, L., TAILLON, B. E., CHEN, Z., TANZER, A., SAUNDERS, A. C. E., CHI, J., YANG, F., CARTER, N. P., HURLES, M. E., WEISSMAN, S. M., HARKINS, T. T., GERSTEIN, M. B., EGHOLM, M., AND SNYDER, M. Paired-end mapping reveals extensive structural variation in the human genome. *Science 318*, 5849 (Oct. 2007), 420–426.
- [86] KRAMPIS, K., BOOTH, T., CHAPMAN, B., TIWARI, B., BICAK, M., FIELD, D., AND NELSON, K. E. Cloud BioLinux: pre-configured and on-demand bioinformatics computing for the genomics community. *BMC Bioinformatics* 13, 1 (2012), 42.
- [87] KURZROCK, R., KANTARJIAN, H. M., DRUKER, B. J., AND TALPAZ, M. Philadelphia chromosome-positive leukemias: from basic mechanisms to molecular therapeutics. Annals of Internal Medicine 138, 10 (2003), 819–830.
- [88] LAFFERTY, J. D., MCCALLUM, A., AND PEREIRA, F. C. N. Conditional random fields: Probabilistic models for segmenting and labeling sequence data. In *Proceedings* of the Eighteenth International Conference on Machine Learning (San Francisco, CA, USA, 2001), ICML '01, Morgan Kaufmann Publishers Inc., pp. 282–289.
- [89] LAM, H. Y. K., MU, X. J., STÜTZ, A. M., TANZER, A., CAYTING, P. D., SNYDER, M., KIM, P. M., KORBEL, J. O., AND GERSTEIN, M. B. Nucleotideresolution analysis of structural variants using BreakSeq and a breakpoint library. *Nature Biotechnology* 28, 1 (2010), 47–55.
- [90] LAM, H. Y. K., PAN, C., CLARK, M. J., LACROUTE, P., CHEN, R., HARAKSINGH, R., O'HUALLACHAIN, M., GERSTEIN, M. B., KIDD, J. M., BUSTAMANTE, C. D., AND SNYDER, M. Detecting and annotating genetic variations using the HugeSeq pipeline. *Nature Biotechnology 30*, 3 (Mar. 2012), 226–229.
- [91] LÄMMEL, R. Google's MapReduce programming model Revisited. Science of Computer Programming 70, 1 (2008), 1 – 30.
- [92] LANDRY, J. J. M., PYL, P. T., RAUSCH, T., ZICHNER, T., TEKKEDIL, M. M., STÜTZ, A. M., JAUCH, A., AIYAR, R. S., PAU, G., DELHOMME, N., GAGNEUR, J., KORBEL, J. O., HUBER, W., AND STEINMETZ, L. M. The genomic and transcriptomic landscape of a HeLa cell line. *G3 (Bethesda, Md.) 3*, 8 (Aug. 2013), 1213–1224.

- [93] LANGMEAD, B., HANSEN, K. D., AND LEEK, J. T. Cloud-scale RNA-sequencing differential expression analysis with Myrna. *Genome Biology* 11, 8 (2010), R83.
- [94] LANGMEAD, B., AND SALZBERG, S. L. Fast gapped-read alignment with Bowtie 2. Nature Methods 9, 4 (Apr. 2012), 357–359.
- [95] LANGMEAD, B., SCHATZ, M. C., LIN, J., POP, M., AND SALZBERG, S. L. Searching for SNPs with cloud computing. *Genome Biology* 10, 11 (2009), R134.
- [96] LANGMEAD, B., TRAPNELL, C., POP, M., AND SALZBERG, S. L. Ultrafast and memory-efficient alignment of short DNA sequences to the human genome. *Genome Biology* 10, 3 (2009), R25.
- [97] LARKIN, D. M., PAPE, G., DONTHU, R., AUVIL, L., WELGE, M., AND LEWIN, H. A. Breakpoint regions and homologous syntemy blocks in chromosomes have different evolutionary histories. *Genome Research 19*, 5 (May 2009), 770–777.
- [98] LAYER, R. M., HALL, I. M., AND QUINLAN, A. R. LUMPY: A probabilistic framework for structural variant discovery. *ArXiv e-prints* (Oct. 2012).
- [99] LAZAR, N. H., WILHELM, L. J., TERHUNE, E., MEYER, T. J., WHELAN, C. W., SÖNMEZ, K., AND CARBONE, L. Gibbon-human syntemy breakpoints carry a distinctive epigenetic signature. *Genome Research* (under review).
- [100] LEE, H., AND SCHATZ, M. C. Genomic dark matter: the reliability of short read mapping illustrated by the genome mappability score. *Bioinformatics 28*, 16 (Aug. 2012), 2097–2105.
- [101] LEE, J. A., CARVALHO, C. M. B., AND LUPSKI, J. R. A DNA replication mechanism for generating nonrecurrent rearrangements associated with genomic disorders. *Cell* 131, 7 (Dec. 2007), 1235–1247.
- [102] LEE, S., HORMOZDIARI, F., ALKAN, C., AND BRUDNO, M. MoDIL: detecting small indels from clone-end sequencing with mixtures of distributions. *Nature Methods* 6, 7 (July 2009), 473–474.
- [103] LEE, W.-P., STROMBERG, M., WARD, A., STEWART, C., GARRISON, E., AND MARTH, G. T. MOSAIK: A hash-based algorithm for accurate next-generation sequencing read mapping. *ArXiv e-prints* (Sept. 2013).

- [104] LEHRMAN, M. A., SCHNEIDER, W. J., SÜDHOF, T. C., BROWN, M. S., GOLD-STEIN, J. L., AND RUSSELL, D. W. Mutation in LDL receptor: Alu-Alu recombination deletes exons encoding transmembrane and cytoplasmic domains. *Science* 227, 4683 (Jan. 1985), 140–146.
- [105] LEVY, S., SUTTON, G., NG, P. C., FEUK, L., HALPERN, A. L., WALENZ, B. P., AXELROD, N., HUANG, J., KIRKNESS, E. F., DENISOV, G., LIN, Y., MACDON-ALD, J. R., PANG, A. W. C., SHAGO, M., STOCKWELL, T. B., TSIAMOURI, A., BAFNA, V., BANSAL, V., KRAVITZ, S. A., BUSAM, D. A., BEESON, K. Y., MCINTOSH, T. C., REMINGTON, K. A., ABRIL, J. F., GILL, J., BORMAN, J., ROGERS, Y.-H., FRAZIER, M. E., SCHERER, S. W., STRAUSBERG, R. L., AND VENTER, J. C. The diploid genome sequence of an individual human. *PLoS Biology* 5, 10 (Sept. 2007), e254.
- [106] LI, H., AND DURBIN, R. Fast and accurate short read alignment with Burrows-Wheeler transform. *Bioinformatics* 25, 14 (July 2009), 1754–1760.
- [107] LI, H., HANDSAKER, B., WYSOKER, A., FENNELL, T., RUAN, J., HOMER, N., MARTH, G., ABECASIS, G., AND DURBIN, R. The sequence alignment/map format and SAMtools. *Bioinformatics* 25, 16 (2009), 2078–2079.
- [108] LI, H., AND HOMER, N. A survey of sequence alignment algorithms for nextgeneration sequencing. *Briefings in Bioinformatics* 11, 5 (Sept. 2010), 473–483.
- [109] LI, R., LI, Y., FANG, X., YANG, H., WANG, J., KRISTIANSEN, K., AND WANG, J. SNP detection for massively parallel whole-genome resequencing. *Genome Research* 19, 6 (June 2009), 1124–1132.
- [110] LI, Y., ZHENG, H., LUO, R., WU, H., ZHU, H., LI, R., CAO, H., WU, B., HUANG, S., SHAO, H., MA, H., ZHANG, F., FENG, S., ZHANG, W., DU, H., TIAN, G., LI, J., ZHANG, X., LI, S., BOLUND, L., KRISTIANSEN, K., DE SMITH, A. J., BLAKEMORE, A. I. F., COIN, L. J. M., YANG, H., WANG, J., AND WANG, J. Structural variation in two human genomes mapped at single-nucleotide resolution by whole genome de novo assembly. *Nature Biotechnology 29*, 8 (July 2011), 723–730.
- [111] LISKAY, R. M., LETSOU, A., AND STACHELEK, J. L. Homology requirement for efficient gene conversion between duplicated chromosomal sequences in mammalian cells. *Genetics* 115, 1 (Jan. 1987), 161–167.
- [112] LIU, P., CARVALHO, C. M. B., HASTINGS, P. J., AND LUPSKI, J. R. Mechanisms for recurrent and complex human genomic rearrangements. *Current Opinion in Genetics & Development 22*, 3 (June 2012), 211–220.
- [113] LIU, P., EREZ, A., NAGAMANI, S. C. S., DHAR, S. U., KOŁODZIEJSKA, K. E., DHARMADHIKARI, A. V., COOPER, M. L., WISZNIEWSKA, J., ZHANG, F., WITH-ERS, M. A., BACINO, C. A., CAMPOS-ACEVEDO, L. D., DELGADO, M. R., FREEDENBERG, D., GARNICA, A., GREBE, T. A., HERNÁNDEZ-ALMAGUER, D., IMMKEN, L., LALANI, S. R., MCLEAN, S. D., NORTHRUP, H., SCAGLIA, F., STRATHEARN, L., TRAPANE, P., KANG, S.-H. L., PATEL, A., CHEUNG, S. W., HASTINGS, P. J., STANKIEWICZ, P., LUPSKI, J. R., AND BI, W. Chromosome catastrophes involve replication mechanisms generating complex genomic rearrangements. *Cell* 146, 6 (Sept. 2011), 889–903.
- [114] LOCKE, D. P., HILLIER, L. W., WARREN, W. C., WORLEY, K. C., NAZARETH, L. V., MUZNY, D. M., YANG, S.-P., WANG, Z., CHINWALLA, A. T., MINX, P., MITREVA, M., COOK, L., DELEHAUNTY, K. D., FRONICK, C., SCHMIDT, H., FULTON, L. A., FULTON, R. S., NELSON, J. O., MAGRINI, V., POHL, C., GRAVES, T. A., MARKOVIC, C., CREE, A., DINH, H. H., HUME, J., KOVAR, C. L., FOWLER, G. R., LUNTER, G., MEADER, S., HEGER, A., PONTING, C. P., MARQUES-BONET, T., ALKAN, C., CHEN, L., CHENG, Z., KIDD, J. M., EICHLER, E. E., WHITE, S., SEARLE, S., VILELLA, A. J., CHEN, Y., FLICEK, P., MA, J., RANEY, B., SUH, B., BURHANS, R., HERRERO, J., HAUSSLER, D., FARIA, R., FERNANDO, O., DARRÉ, F., FARRÉ, D., GAZAVE, E., OLIVA, M., NAVARRO, A., Roberto, R., Capozzi, O., Archidiacono, N., Della Valle, G., Purgato, S., Rocchi, M., Konkel, M. K., Walker, J. A., Ullmer, B., Batzer, M. A., SMIT, A. F. A., HUBLEY, R., CASOLA, C., SCHRIDER, D. R., HAHN, M. W., QUESADA, V., PUENTE, X. S., ORDÓÑEZ, G. R., LÓPEZ-OTÍN, C., VINAR, T., BREJOVA, B., RATAN, A., HARRIS, R. S., MILLER, W., KOSIOL, C., LAWSON, H. A., TALIWAL, V., MARTINS, A. L., SIEPEL, A., ROYCHOUDHURY, A., MA, X., Degenhardt, J., Bustamante, C. D., Gutenkunst, R. N., Mailund, T., DUTHEIL, J. Y., HOBOLTH, A., SCHIERUP, M. H., RYDER, O. A., YOSHINAGA, Y., DE JONG, P. J., WEINSTOCK, G. M., ROGERS, J., MARDIS, E. R., GIBBS, R. A., AND WILSON, R. K. Comparative and demographic analysis of orang-utan genomes. NATURE 469, 7331 (Jan. 2011), 529–533.
- [115] LUPSKI, J. R. Genomic disorders: structural features of the genome can lead to DNA rearrangements and human disease traits. *Trends in Genetics* 14, 10 (Oct. 1998), 417–422.
- [116] MALHOTRA, A., LINDBERG, M., FAUST, G. G., LEIBOWITZ, M. L., CLARK, R. A., LAYER, R. M., QUINLAN, A. R., AND HALL, I. M. Breakpoint profiling of

64 cancer genomes reveals numerous complex rearrangements spawned by homologyindependent mechanisms. *Genome Research 23*, 5 (May 2013), 762–776.

- [117] MARCO SOLA, S., SAMMETH, M., GUIGÓ, R., AND RIBECA, P. The GEM mapper: fast, accurate and versatile alignment by filtration. *Nature Methods 9*, 12 (Dec. 2012), 1185–1188.
- [118] MARSCHALL, T., COSTA, I. G., CANZAR, S., BAUER, M., KLAU, G. W., SCHLIEP, A., AND SCHÖNHUTH, A. CLEVER: clique-enumerating variant finder. *Bioinformatics* 28, 22 (Nov. 2012), 2875–2882.
- [119] MARTIN, M. Cutadapt removes adapter sequences from high-throughput sequencing reads. *EMBnet.journal* 17, 1 (2011), 10–12.
- [120] MARX, V. Biology: The big challenges of big data. NATURE 498, 7453 (June 2013), 255–260.
- [121] MCCALLUM, A., SCHULTZ, K., AND SINGH, S. FACTORIE: Probabilistic programming via imperatively defined factor graphs. In *Neural Information Processing Systems (NIPS)* (2009).
- [122] MCCARROLL, S. A., HUETT, A., KUBALLA, P., CHILEWSKI, S. D., LANDRY, A., GOYETTE, P., ZODY, M. C., HALL, J. L., BRANT, S. R., CHO, J. H., DUERR, R. H., SILVERBERG, M. S., TAYLOR, K. D., RIOUX, J. D., ALTSHULER, D., DALY, M. J., AND XAVIER, R. J. Deletion polymorphism upstream of IRGM associated with altered IRGM expression and Crohn's disease. *Nature Genetics* 40, 9 (Sept. 2008), 1107–1112.
- [123] MCCARROLL, S. A., KURUVILLA, F. G., KORN, J. M., CAWLEY, S., NEMESH, J., WYSOKER, A., SHAPERO, M. H., DE BAKKER, P. I. W., MALLER, J. B., KIRBY, A., ELLIOTT, A. L., PARKIN, M., HUBBELL, E., WEBSTER, T., MEI, R., VEITCH, J., COLLINS, P. J., HANDSAKER, R., LINCOLN, S., NIZZARI, M., BLUME, J., JONES, K. W., RAVA, R., DALY, M. J., GABRIEL, S. B., AND ALTSHULER, D. Integrated detection and population-genetic analysis of SNPs and copy number variation. *Nature Genetics* 40, 10 (Oct. 2008), 1166–1174.
- [124] MCKENNA, A., HANNA, M., BANKS, E., SIVACHENKO, A., CIBULSKIS, K., KERNYTSKY, A., GARIMELLA, K., ALTSHULER, D., GABRIEL, S., DALY, M., AND DEPRISTO, M. A. The Genome Analysis Toolkit: A MapReduce framework for analyzing next-generation DNA sequencing data. *Genome Research 20*, 9 (Sept. 2010), 1297–1303.

- [125] MEDVEDEV, P., FIUME, M., DZAMBA, M., SMITH, T., AND BRUDNO, M. Detecting copy number variation with mated short reads. *Genome Research 20*, 11 (Nov. 2010), 1613–1622.
- [126] MICHAELSON, J. J., AND SEBAT, J. forestSV: structural variant discovery through statistical learning. *Nature Methods 9*, 8 (Aug. 2012), 819–821.
- [127] MILLS, R. E., PITTARD, W. S., MULLANEY, J. M., FAROOQ, U., CREASY, T. H., MAHURKAR, A. A., KEMEZA, D. M., STRASSLER, D. S., PONTING, C. P., WEB-BER, C., AND DEVINE, S. E. Natural genetic variation caused by small insertions and deletions in the human genome. *Genome Research 21*, 6 (June 2011), 830–839.
- [128] MILLS, R. E., WALTER, K., STEWART, C., HANDSAKER, R. E., CHEN, K., ALKAN, C., ABYZOV, A., YOON, S. C., YE, K., CHEETHAM, R. K., CHINWALLA, A., CONRAD, D. F., FU, Y., GRUBERT, F., HAJIRASOULIHA, I., HORMOZDIARI, F., IAKOUCHEVA, L. M., IQBAL, Z., KANG, S., KIDD, J. M., KONKEL, M. K., KORN, J., KHURANA, E., KURAL, D., LAM, H. Y. K., LENG, J., LI, R., LI, Y., LIN, C.-Y., LUO, R., MU, X. J., NEMESH, J., PECKHAM, H. E., RAUSCH, T., SCALLY, A., SHI, X., STROMBERG, M. P., STÜTZ, A. M., URBAN, A. E., WALKER, J. A., WU, J., ZHANG, Y., ZHANG, Z. D., BATZER, M. A., DING, L., MARTH, G. T., MCVEAN, G., SEBAT, J., SNYDER, M., WANG, J., YE, K., EICH-LER, E. E., GERSTEIN, M. B., HURLES, M. E., LEE, C., MCCARROLL, S. A., AND KORBEL, J. O. Mapping copy number variation by population-scale genome sequencing. *Nature 470*, 7332 (Feb. 2011), 59–65.
- [129] MIMORI, T., NARIAI, N., KOJIMA, K., TAKAHASHI, M., ONO, A., SATO, Y., YAMAGUCHI-KABATA, Y., AND NAGASAKI, M. iSVP: an integrated structural variant calling pipeline from high-throughput sequencing data. *BMC Systems Biology* 7 (2013), S8.
- [130] MISCEO, D., CAPOZZI, O., ROBERTO, R., DELL'OGLIO, M. P., ROCCHI, M., STANYON, R., AND ARCHIDIACONO, N. Tracking the complex flow of chromosome rearrangements from the Hominoidea Ancestor to extant Hylobates and Nomascus Gibbons by high-resolution synteny mapping. *Genome Research 18*, 9 (Sept. 2008), 1530–1537.
- [131] MOHAMED, N. M., LIN, H., AND FENG, W.-C. Accelerating Data-Intensive Genome Analysis in the Cloud. In 5th International Conference on Bioinformatics and Computational Biology (BICoB) (Honolulu, Hawaii, USA, March 2013).

- [132] MURPHY, W. J., LARKIN, D. M., EVERTS-VAN DER WIND, A., BOURQUE, G., TESLER, G., AUVIL, L., BEEVER, J. E., CHOWDHARY, B. P., GALIBERT, F., GATZKE, L., HITTE, C., MEYERS, S. N., MILAN, D., OSTRANDER, E. A., PAPE, G., PARKER, H. G., RAUDSEPP, T., ROGATCHEVA, M. B., SCHOOK, L. B., SKOW, L. C., WELGE, M., WOMACK, J. E., O'BRIEN, S. J., PEVZNER, P. A., AND LEWIN, H. A. Dynamics of mammalian chromosome evolution inferred from multispecies comparative maps. *Science 309*, 5734 (July 2005), 613–617.
- [133] MYERS, E. W. The fragment assembly string graph. Bioinformatics 21 Suppl 2 (Sept. 2005), ii79–85.
- [134] NADEAU, J. H., AND TAYLOR, B. A. Lengths of chromosomal segments conserved since divergence of man and mouse. *Proceedings of the National Academy of Sciences* of the United States of America 81, 3 (Feb. 1984), 814–818.
- [135] NAGARAJAN, N., AND POP, M. Sequence assembly demystified. Nature Reviews Genetics 14, 3 (Jan. 2013), 157–167.
- [136] NEPH, S., KUEHN, M. S., REYNOLDS, A. P., HAUGEN, E., THURMAN, R. E., JOHNSON, A. K., RYNES, E., MAURANO, M. T., VIERSTRA, J., THOMAS, S., SANDSTROM, R., HUMBERT, R., AND STAMATOYANNOPOULOS, J. A. BEDOPS: high-performance genomic feature operations. *Bioinformatics* 28, 14 (July 2012), 1919–1920.
- [137] NGUYEN, T., SHI, W., AND RUDEN, D. CloudAligner: A fast and full-featured MapReduce based tool for sequence mapping. BMC Research Notes 4 (2011), 171.
- [138] NIEMENMAA, M., KALLIO, A., SCHUMACHER, A., KLEMELÄ, P., KORPELAINEN, E., AND HELJANKO, K. Hadoop-BAM: directly manipulating next generation sequencing data in the cloud. *Bioinformatics* 28, 6 (Mar. 2012), 876–877.
- [139] NORD, A. S., LEE, M., KING, M.-C., AND WALSH, T. Accurate and exact CNV identification from targeted high-throughput sequence data. *BMC Genomics* 12 (2011), 184.
- [140] NORDBERG, H., BHATIA, K., WANG, K., AND WANG, Z. BioPig: a Hadoop-based analytic toolkit for large-scale sequence data. *Bioinformatics* 29, 23 (Dec. 2013), 3014–3019.
- [141] NORTON, N., LI, D., RIEDER, M. J., SIEGFRIED, J. D., RAMPERSAUD, E., ZÜCH-NER, S., MANGOS, S., GONZALEZ-QUINTANA, J., WANG, L., MCGEE, S., REISER,

J., MARTIN, E., NICKERSON, D. A., AND HERSHBERGER, R. E. Genome-wide studies of copy number variation and exome sequencing identify rare variants in BAG3 as a cause of dilated cardiomyopathy. *American Journal of Human Genetics* 88, 3 (Mar. 2011), 273–282.

- [142] NOVOCRAFT, INC. Novoalign. http://www.novocraft.com.
- [143] O'CONNOR, B. D., MERRIMAN, B., AND NELSON, S. F. SeqWare Query Engine: storing and searching sequence data in the cloud. *BMC Bioinformatics 11*, Suppl 12 (2010), S2.
- [144] OHNO, S. Ancient linkage groups and frozen accidents. NATURE 244, 5414 (Aug. 1973), 259–262.
- [145] OPEN CLOUD CONSORTIUM. Bionimbus. http://bionimbus. opensciencedatacloud.org/.
- [146] OSHLACK, A., ROBINSON, M. D., AND YOUNG, M. D. From RNA-seq reads to differential expression results. *Genome Biology* 11, 12 (2010), 220.
- [147] PARK, P. J. ChIP-seq: advantages and challenges of a maturing technology. Nature Reviews Genetics 10, 10 (Oct. 2009), 669–680.
- [148] PEVZNER, P., AND TESLER, G. Human and mouse genomic sequences reveal extensive breakpoint reuse in mammalian evolution. *Proceedings of the National Academy* of Sciences of the United States of America 100, 13 (June 2003), 7672–7677.
- [149] PHILLIPS, J. E., AND CORCES, V. G. CTCF: master weaver of the genome. Cell 137, 7 (June 2009), 1194–1211.
- [150] PIREDDU, L., LEO, S., AND ZANETTI, G. Mapreducing a genomic sequencing workflow. In *Proceedings of the Second International Workshop on MapReduce and Its Applications* (New York, NY, USA, 2011), MapReduce '11, ACM, pp. 67–74.
- [151] PIREDDU, L., LEO, S., AND ZANETTI, G. SEAL: a distributed short read mapping and duplicate removal tool. *Bioinformatics* 27, 15 (July 2011), 2159–2160.
- [152] PLEASANCE, E. D., CHEETHAM, R. K., STEPHENS, P. J., MCBRIDE, D. J., HUMPHRAY, S. J., GREENMAN, C. D., VARELA, I., LIN, M.-L., ORDÓÑEZ, G. R., BIGNELL, G. R., YE, K., ALIPAZ, J., BAUER, M. J., BEARE, D., BUTLER, A., CARTER, R. J., CHEN, L., COX, A. J., EDKINS, S., KOKKO-GONZALES, P. I., GORMLEY, N. A., GROCOCK, R. J., HAUDENSCHILD, C. D., HIMS, M. M.,

JAMES, T., JIA, M., KINGSBURY, Z., LEROY, C., MARSHALL, J., MENZIES, A., MUDIE, L. J., NING, Z., ROYCE, T., SCHULZ-TRIEGLAFF, O. B., SPIRIDOU, A., STEBBINGS, L. A., SZAJKOWSKI, L., TEAGUE, J., WILLIAMSON, D., CHIN, L., ROSS, M. T., CAMPBELL, P. J., BENTLEY, D. R., FUTREAL, P. A., AND STRATTON, M. R. A comprehensive catalogue of somatic mutations from a human cancer genome. *NATURE 463*, 7278 (Jan. 2010), 191–196.

- [153] QI, J., AND ZHAO, F. inGAP-sv: a novel scheme to identify and visualize structural variation from paired end mapping data. *Nucleic Acids Research 39*, Web Server issue (July 2011), W567–75.
- [154] QUINLAN, A. R., CLARK, R. A., SOKOLOVA, S., LEIBOWITZ, M. L., ZHANG, Y., HURLES, M. E., MELL, J. C., AND HALL, I. M. Genome-wide mapping and assembly of structural variant breakpoints in the mouse genome. *Genome Research* 20, 5 (May 2010), 623–635.
- [155] QUINLAN, A. R., AND HALL, I. M. BEDTools: a flexible suite of utilities for comparing genomic features. *Bioinformatics* 26, 6 (Mar. 2010), 841–842.
- [156] RAPHAEL, B. J., VOLIK, S., COLLINS, C., AND PEVZNER, P. A. Reconstructing tumor genome architectures. *Bioinformatics 19 Suppl 2* (Oct. 2003), ii162–71.
- [157] RAUSCH, T., ZICHNER, T., SCHLATTL, A., STÜTZ, A. M., BENES, V., AND KORBEL, J. O. DELLY: structural variant discovery by integrated paired-end and split-read analysis. *Bioinformatics* 28, 18 (Sept. 2012), i333–i339.
- [158] ROBBERECHT, C., VOET, T., ZAMANI ESTEKI, M., NOWAKOWSKA, B. A., AND VERMEESCH, J. R. Nonallelic homologous recombination between retrotransposable elements is a driver of de novo unbalanced translocations. *Genome Research 23*, 3 (Mar. 2013), 411–418.
- [159] ROBERTO, R., CAPOZZI, O., WILSON, R. K., MARDIS, E. R., LOMIENTO, M., TÜZÜN, E., CHENG, Z., MOOTNICK, A. R., ARCHIDIACONO, N., ROCCHI, M., AND EICHLER, E. E. Molecular refinement of gibbon genome rearrangements. *Genome Research 17*, 2 (Feb. 2007), 249–257.
- [160] RUBINSTEIN, R. Y., AND KROESE, D. P. Simulation and the Monte Carlo Method (Wiley Series in Probability and Statistics), 2 ed. Wiley, 2007.
- [161] RUFFALO, M., LAFRAMBOISE, T., AND KOYUTÜRK, M. Comparative analysis of algorithms for next-generation sequencing read alignment. *Bioinformatics* (Aug. 2011).

- [162] SCHADT, E. E., LINDERMAN, M. D., SORENSON, J., LEE, L., AND NOLAN, G. P. Computational solutions to large-scale data management and analysis. *Nature Re*views Genetics 11, 9 (Sept. 2010), 647–657.
- [163] SCHATZ, M. C. CloudBurst: highly sensitive read mapping with MapReduce. Bioinformatics 25, 11 (June 2009), 1363–1369.
- [164] SCHATZ, M. C., LANGMEAD, B., AND SALZBERG, S. L. Cloud computing and the DNA data race. *Nature Biotechnology* 28, 7 (July 2010), 691–693.
- [165] SCHATZ, M. C., SOMMER, D., KELLEY, D., AND POP, M. De novo assembly of large genomes using cloud computing. In *The Biology of Genomes* (2010).
- [166] SCHMIDT, D., SCHWALIE, P. C., WILSON, M. D., BALLESTER, B., GONÇALVES, Â., KUTTER, C., BROWN, G. D., MARSHALL, A., FLICEK, P., AND ODOM, D. T. Waves of Retrotransposon Expansion Remodel Genome Organization and CTCF Binding in Multiple Mammalian Lineages. *Cell* (Jan. 2012).
- [167] SCHUMACHER, A., PIREDDU, L., NIEMENMAA, M., KALLIO, A., KORPELAINEN, E., ZANETTI, G., AND HELJANKO, K. SeqPig: simple and scalable scripting for large sequencing data sets in Hadoop. *Bioinformatics* (Nov. 2013).
- [168] SCHWALIE, P. C., WARD, M., CAIN, C., FAURE, A., GILAD, Y., ODOM, D., AND FLICEK, P. Co-binding by YY1 identifies the transcriptionally active, highly conserved set of CTCF-bound regions in primates. *Genome Biology* (under review).
- [169] SEBAT, J., LAKSHMI, B., MALHOTRA, D., TROGE, J., LESE-MARTIN, C., WALSH, T., YAMROM, B., YOON, S., KRASNITZ, A., KENDALL, J., LEOTTA, A., PAI, D., ZHANG, R., LEE, Y.-H., HICKS, J., SPENCE, S. J., LEE, A. T., PUURA, K., LEHTIMÄKI, T., LEDBETTER, D., GREGERSEN, P. K., BREGMAN, J., SUTCLIFFE, J. S., JOBANPUTRA, V., CHUNG, W., WARBURTON, D., KING, M.-C., SKUSE, D., GESCHWIND, D. H., GILLIAM, T. C., YE, K., AND WIGLER, M. Strong association of de novo copy number mutations with autism. *Science 316*, 5823 (Apr. 2007), 445–449.
- [170] SHARP, A. J., HANSEN, S., SELZER, R. R., CHENG, Z., REGAN, R., HURST, J. A., STEWART, H., PRICE, S. M., BLAIR, E., HENNEKAM, R. C., FITZPATRICK, C. A., SEGRAVES, R., RICHMOND, T. A., GUIVER, C., ALBERTSON, D. G., PINKEL, D., EIS, P. S., SCHWARTZ, S., KNIGHT, S. J. L., AND EICHLER, E. E. Discovery of previously unidentified genomic disorders from the duplication architecture of the human genome. *Nature Genetics 38*, 9 (Sept. 2006), 1038–1042.

- [171] SHEN, Y., GU, Y., AND PE'ER, I. A hidden Markov model for copy number variant prediction from whole genome resequencing data. BMC Bioinformatics 12 Suppl 6 (2011), S4.
- [172] SHENDURE, J., AND JI, H. Next-generation DNA sequencing. Nature Biotechnology 26, 10 (Oct. 2008), 1135–1145.
- [173] SINDI, S., HELMAN, E., BASHIR, A., AND RAPHAEL, B. J. A geometric approach for classification and comparison of structural variants. *Bioinformatics* 25, 12 (June 2009), i222–30.
- [174] SINDI, S. S., ONAL, S., PENG, L., WU, H.-T., AND RAPHAEL, B. J. An integrative probabilistic model for identification of structural variation in sequencing data. *Genome Biology* 13, 3 (Mar. 2012), R22.
- [175] SMIT, A., HUBLEY, R., AND GREEN, P. Repeatmasker open-3.0. http://www. repeatmasker.org, 1996-2010.
- [176] SMITH, T. F., AND WATERMAN, M. S. Identification of common molecular subsequences. Journal of Molecular Biology 147, 1 (Mar. 1981), 195–197.
- [177] STEIN, L. D. Towards a cyberinfrastructure for the biological sciences: progress, visions and challenges. *Nature Reviews Genetics* 9, 9 (Sept. 2008), 678–688.
- [178] STEIN, L. D. The case for cloud computing in genome informatics. *Genome Biology* 11, 5 (2010), 207.
- [179] STEPHENS, P. J., GREENMAN, C. D., FU, B., YANG, F., BIGNELL, G. R., MUDIE, L. J., PLEASANCE, E. D., LAU, K. W., BEARE, D., STEBBINGS, L. A., MCLAREN, S., LIN, M.-L., MCBRIDE, D. J., VARELA, I., NIK-ZAINAL, S., LEROY, C., JIA, M., MENZIES, A., BUTLER, A. P., TEAGUE, J. W., QUAIL, M. A., BURTON, J., SWERDLOW, H., CARTER, N. P., MORSBERGER, L. A., IACOBUZIO-DONAHUE, C., FOLLOWS, G. A., GREEN, A. R., FLANAGAN, A. M., STRATTON, M. R., FUTREAL, P. A., AND CAMPBELL, P. J. Massive genomic rearrangement acquired in a single catastrophic event during cancer development. *Cell 144*, 1 (Jan. 2011), 27–40.
- [180] TALWALKAR, A., LIPTRAP, J., NEWCOMB, J., HARTL, C., TERHORST, J., CUR-TIS, K., BRESLER, M., SONG, Y. S., JORDAN, M. I., AND PATTERSON, D. SMASH: A Benchmarking Toolkit for Variant Calling. ArXiv e-prints (Oct. 2013).

- [181] THAIN, D., TANNENBAUM, T., AND LIVNY, M. Distributed computing in practice: the condor experience. *Concurrency - Practice and Experience* 17, 2-4 (2005), 323– 356.
- [182] TREANGEN, T. J., AND SALZBERG, S. L. Repetitive DNA and next-generation sequencing: computational challenges and solutions. *Nature Reviews Genetics* 13, 1 (Jan. 2012), 36–46.
- [183] TRINH, Q. M., JEN, F.-Y. A., ZHOU, Z., CHU, K. M., PERRY, M. D., KEPHART, E. T., CONTRINO, S., RUZANOV, P., AND STEIN, L. D. Cloud-based uniform ChIP-Seq processing tools for modENCODE and ENCODE. *BMC Genomics* 14 (2013), 494.
- [184] TÜZÜN, E., SHARP, A. J., BAILEY, J. A., KAUL, R., MORRISON, V. A., PERTZ, L. M., HAUGEN, E., HAYDEN, H., ALBERTSON, D., PINKEL, D., OLSON, M. V., AND EICHLER, E. E. Fine-scale structural variation of the human genome. *Nature Genetics* 37, 7 (July 2005), 727–732.
- [185] UNTERGASSER, A., CUTCUTACHE, I., KORESSAAR, T., YE, J., FAIRCLOTH, B. C., REMM, M., AND ROZEN, S. G. Primer3 - new capabilities and interfaces. *Nucleic Acids Research* 40, 15 (2012), e115.
- [186] VILELLA, A. J., SEVERIN, J., URETA-VIDAL, A., HENG, L., DURBIN, R., AND BIRNEY, E. EnsemblCompara GeneTrees: Complete, duplication-aware phylogenetic trees in vertebrates. *Genome Research 19*, 2 (Feb. 2009), 327–335.
- [187] VOLIK, S., ZHAO, S., CHIN, K., BREBNER, J. H., HERNDON, D. R., TAO, Q., KOWBEL, D., HUANG, G., LAPUK, A., KUO, W.-L., MAGRANE, G., DE JONG, P., GRAY, J. W., AND COLLINS, C. End-sequence profiling: sequence-based analysis of aberrant genomes. *Proceedings of the National Academy of Sciences of the United States of America 100*, 13 (June 2003), 7696–7701.
- [188] WALSH, T., MCCLELLAN, J. M., MCCARTHY, S. E., ADDINGTON, A. M., PIERCE, S. B., COOPER, G. M., NORD, A. S., KUSENDA, M., MALHOTRA, D., BHANDARI, A., STRAY, S. M., RIPPEY, C. F., ROCCANOVA, P., MAKAROV, V., LAKSHMI, B., FINDLING, R. L., SIKICH, L., STROMBERG, T., MERRIMAN, B., GOGTAY, N., BUTLER, P., ECKSTRAND, K., NOORY, L., GOCHMAN, P., LONG, R., CHEN, Z., DAVIS, S., BAKER, C., EICHLER, E. E., MELTZER, P. S., NELSON, S. F., SINGLETON, A. B., LEE, M. K., RAPOPORT, J. L., KING, M.-C., AND SEBAT, J. Rare structural variants disrupt multiple genes in neurodevelopmental pathways in schizophrenia. *Science 320*, 5875 (Apr. 2008), 539–543.

- [189] WANG, J., MULLIGHAN, C. G., EASTON, J., ROBERTS, S., HEATLEY, S. L., MA, J., RUSCH, M. C., CHEN, K., HARRIS, C. C., DING, L., HOLMFELDT, L., PAYNE-TURNER, D., FAN, X., WEI, L., ZHAO, D., OBENAUER, J. C., NAEVE, C., MARDIS, E. R., WILSON, R. K., DOWNING, J. R., AND ZHANG, J. CREST maps somatic structural variation in cancer genomes with base-pair resolution. *Nature Methods 8*, 8 (Aug. 2011), 652–654.
- [190] WEESE, D., HOLTGREWE, M., AND REINERT, K. RazerS 3: Faster, fully sensitive read mapping. *Bioinformatics* 28, 20 (Oct. 2012), 2592–2599.
- [191] WEISCHENFELDT, J., SYMMONS, O., SPITZ, F., AND KORBEL, J. O. Phenotypic impact of genomic structural variation: insights from and for human disease. *Nature Reviews Genetics* 14, 2 (Feb. 2013), 125–138.
- [192] WESTON, J., RATLE, F., MOBAHI, H., AND COLLOBERT, R. Deep learning via semi-supervised embedding. In *Neural Networks: Tricks of the Trade*, G. Montavon, G. Orr, and K.-R. Müller, Eds., vol. 7700 of *Lecture Notes in Computer Science*. Springer Berlin Heidelberg, 2012, pp. 639–655.
- [193] WIENBERG, J. The evolution of eutherian chromosomes. Current Opinion in Genetics & Development 14, 6 (Dec. 2004), 657–666.
- [194] WONG, K., KEANE, T. M., STALKER, J., AND ADAMS, D. J. Enhanced structural variant and breakpoint detection using SVMerge by integration of multiple detection methods and local assembly. *Genome Biology* 11, 12 (2010), R128.
- [195] WU, T. D., AND NACU, S. Fast and SNP-tolerant detection of complex variants and splicing in short reads. *Bioinformatics* 26, 7 (Apr. 2010), 873–881.
- [196] XU, H., HANDOKO, L., WEI, X., YE, C., SHENG, J., WEI, C.-L., LIN, F., AND SUNG, W.-K. A signal-noise model for significance analysis of ChIP-seq with negative control. *Bioinformatics 26*, 9 (May 2010), 1199–1204.
- [197] YALCIN, B., WONG, K., AGAM, A., GOODSON, M., KEANE, T. M., GAN, X., NELLÅKER, C., GOODSTADT, L., NICOD, J., AND BHOMRA, A. Sequence-based characterization of structural variation in the mouse genome. *NATURE* 477, 7364 (2011), 326–329.
- [198] YANG, L., LUQUETTE, L. J., GEHLENBORG, N., XI, R., HASELEY, P. S., HSIEH, C.-H., ZHANG, C., REN, X., PROTOPOPOV, A., CHIN, L., KUCHERLAPATI, R., LEE, C., AND PARK, P. J. Diverse mechanisms of somatic structural variations in human cancer genomes. *Cell 153*, 4 (May 2013), 919–929.

- [199] YASUDA, T., SUZUKI, S., NAGASAKI, M., AND MIYANO, S. ChopSticks: Highresolution analysis of homozygous deletions by exploiting concordant read pairs. *BMC Bioinformatics 13*, 1 (Oct. 2012), 279.
- [200] YE, K., SCHULZ, M. H., LONG, Q., APWEILER, R., AND NING, Z. Pindel: a pattern growth approach to detect break points of large deletions and medium sized insertions from paired-end short reads. *Bioinformatics* 25, 21 (Nov. 2009), 2865– 2871.
- [201] YOON, S., XUAN, Z., MAKAROV, V., YE, K., AND SEBAT, J. Sensitive and accurate detection of copy number variants using read depth of coverage. *Genome Research 19*, 9 (Sept. 2009), 1586–1592.
- [202] ZEITOUNI, B., BOEVA, V., JANOUEIX-LEROSEY, I., LOEILLET, S., LEGOIX-NÉ, P., NICOLAS, A., DELATTRE, O., AND BARILLOT, E. SVDetect: a tool to identify genomic structural variations from paired-end and mate-pair sequencing data. *Bioinformatics 26*, 15 (Aug. 2010), 1895–1896.
- [203] ZHANG, J., AND WU, Y. SVseq: an approach for detecting exact breakpoints of deletions with low-coverage sequence data. *Bioinformatics* 27, 23 (Nov. 2011), 3228– 3234.
- [204] ZHANG, L., GU, S., WANG, B., LIU, Y., AND AZUAJE, F. Gene set analysis in the cloud. *Bioinformatics* (Nov. 2011).
- [205] ZHAO, S., PRENGER, K., SMITH, L., MESSINA, T., FAN, H., JAEGER, E., AND STEPHENS, S. Rainbow: a tool for large-scale whole-genome sequencing data analysis using cloud computing. *BMC Genomics* 14, 1 (2013), 425.
- [206] ZHENG, Y.-S., ZHANG, H., ZHANG, X.-J., FENG, D.-D., LUO, X.-Q., ZENG, C.-W., LIN, K.-Y., ZHOU, H., QU, L.-H., ZHANG, P., AND CHEN, Y.-Q. MiR-100 regulates cell differentiation and survival by targeting RBSP3, a phosphatase-like tumor suppressor in acute myeloid leukemia. *Oncogene 31*, 1 (Jan. 2012), 80–92.